

Innovation for Sustainable Business

Edited by Iwona Zdonek and Adam R. Szromek Printed Edition of the Special Issue Published in *Sustainability*



www.mdpi.com/journal/sustainability

Innovation for Sustainable Business

Innovation for Sustainable Business

Editors

Iwona Zdonek Adam R. Szromek

MDPI • Basel • Beijing • Wuhan • Barcelona • Belgrade • Manchester • Tokyo • Cluj • Tianjin



Editors Iwona Zdonek The Silesian University of Technology Poland

Adam R. Szromek The Silesian University of Technology Poland

Editorial Office MDPI St. Alban-Anlage 66 4052 Basel, Switzerland

This is a reprint of articles from the Special Issue published online in the open access journal *Sustainability* (ISSN 2071-1050) (available at: https://www.mdpi.com/journal/sustainability/ special_issues/Innovation_for_Sustainable_Business).

For citation purposes, cite each article independently as indicated on the article page online and as indicated below:

LastName, A.A.; LastName, B.B.; LastName, C.C. Article Title. *Journal Name* Year, *Volume Number*, Page Range.

ISBN 978-3-0365-6481-4 (Hbk) ISBN 978-3-0365-6482-1 (PDF)

© 2023 by the authors. Articles in this book are Open Access and distributed under the Creative Commons Attribution (CC BY) license, which allows users to download, copy and build upon published articles, as long as the author and publisher are properly credited, which ensures maximum dissemination and a wider impact of our publications.

The book as a whole is distributed by MDPI under the terms and conditions of the Creative Commons license CC BY-NC-ND.

Contents

About the Editors
Preface to "Innovation for Sustainable Business"
Magdalena Pichlak
Reprinted from: <i>Sustainability</i> 2021 , <i>13</i> , 5354, doi:10.3390/su13105354
Iwona Zdonek, Anna Mularczyk and Grzegorz PolokThe Idea of Corporate Social Responsibility in the Opinion of Future Managers—ComparativeResearch between Poland and GeorgiaReprinted from: Sustainability 2021, 13, 7045, doi:10.3390/su1313704519
Fernanda Q. Sperotto and Iván G. P. Tartaruga The Green Side of Industry: The Drivers and the Impacts of ECO-Innovations in Brazil Reprinted from: Sustainability 2021, 13, 8065, doi:10.3390/su13148065
Francis Ezieshi Monyei, Wilfred I. Ukpere, Emmanuel Kalu Agbaeze, Solomon Omonona,Lovlyn Ekeowa Kelvin-Iloafu and Happiness Ozioma Obi-AnikeThe Impact of Succession Management on Small and Medium Enterprises' Sustainability inLagos State, NigeriaReprinted from: Sustainability 2021, 13, 13489, doi:10.3390/su13231348963
Zhenfang Zhang, Nan Zhang and Jiguang WangThe Influencing Factors on Impulse Buying Behavior of Consumers under the Mode of HungerMarketing in Live CommerceReprinted from: Sustainability 2022, 14, 2122, doi:10.3390/su14042122
Marzena PodgórskaChallenges and Perspectives in Innovative Projects Focused on Sustainable Industry 4.0—ACase Study on Polish Project TeamsReprinted from: Sustainability 2022, 14, 5334, doi:10.3390/su14095334Study on Polish Project Teams
Yu Liu and Hao Zhang Driving Sustainable Innovation in New Ventures: A Study Based on the fsQCA Approach Reprinted from: <i>Sustainability</i> 2022, 14, 5738, doi:10.3390/su14095738

About the Editors

Iwona Zdonek

Dr Iwona Zdonek is an academic researcher from Silesian University of Technology in Poland. She works in the Faculty of Organization and Management in the Department of Economics and IT.

Her research interests focus on sustainable development. Especially they concern problems related to the transformation towards circular economy and Society 5.0. Most of her articles in these areas has been published in journals such as Energies and Sustainability.

Adam R. Szromek

Adam R. Szromek is a professor of Social Science. He is a lecturer of management science at the Silesian University of Technology in Poland and a member of the Commission on Management Sciences at the Polish Academy of Sciences.

His research interests focus on heritage tourism and health tourism with special attention to spa tourism management and business models in spa enterprises. Professor Szromek is also a lecturer in statistics and computer science at engineering studies. In his spare time he writes adventure novels.

Prof. Szromek's research has appeared in the Tourism Management, American Journal of Tourism Management, Health Policy, Health Services Research, Sustainability, Studies in Physical Culture and Tourism and Human Resources for Health and in many other scientific journals.

Preface to "Innovation for Sustainable Business"

Contemporary problems related to climate change and limited resources draw our attention towards changes in the field of sustainable activities. Moreover, the need for sustainable development creates opportunities for businesses by implementing innovations that are designed to provide a competitive advantage to those who apply more sustainable practices and who offer more sustainable products.

To promote the effectiveness of innovations for sustainable business, there is an urgent need to understand the conditions that influence the success or failure of their implementation, and to comprehend the complexities of their social, economic, and environmental dimensions.

We are pleased to invite you to read our book about innovation for sustainable business. It contains research, the conception, and case studies of ecological innovation and the green industry. A large part of the discussed issues was devoted to leadership.

We would like to thank all the authors of the following chapters of this book. We hope that their effort put into the exploration of the presented issues will contribute to a better understanding of the mechanisms inherent in innovation for sustainable business.

Iwona Zdonek and Adam R. Szromek Editors





Article The Drivers of Technological Eco-Innovation—Dynamic Capabilities and Leadership

Magdalena Pichlak

Faculty of Organization and Management, Institute of Economy and Informatics, Silesian University of Technology, 44-100 Gliwice, Poland; magdalena.pichlak@polsl.pl

Abstract: In the paper, a theoretical framework that combines the multidimensional conceptualization of dynamic capabilities (sensing, seizing, and reconfiguring) with two leadership styles (transactional and transformational) and two types of eco-innovation (incremental and radical) was developed and empirically tested. The purpose of this study is to answer the key question: how different leadership styles influence the potential of dynamic capabilities to generate ecological changes. The research examining the theoretical framework was quantitative and was based on a deliberately selection of the sample, which included 54 of the most evo-innovative Polish companies. The results indicate that transformational leadership moderates the positive relationship between seizing capability and both incremental and radical technological eco-innovation, whereas transactional leadership moderates the positive relationship between seizing three theoretical concepts, showing the importance of a given leadership style as a factor enhancing the potential of dynamic capabilities for the development of technological eco-innovation. Moreover, the study may be a contribution to a broader discussion on the specifics of eco-innovative activity and its behavioral conditions.

Keywords: dynamic capabilities; eco-innovation; leadership

Citation: Pichlak, M. The Drivers of Technological Eco-Innovation— Dynamic Capabilities and Leadership. *Sustainability* **2021**, *13*, 5354. https:// doi.org/10.3390/su13105354

Academic Editors: Iwona Zdonek and Adam R. Szromek

Received: 19 April 2021 Accepted: 10 May 2021 Published: 11 May 2021

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2021 by the author. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/).

1. Introduction

Modern companies have to deal with increasingly frequent changes taking place in their environment. These changes result from globalization [1], liberalization of trade, the growing intensity of competition in international markets [2], the accelerating pace of technological progress, and the associated shortened life cycle of many technologies [3]. These changes are accompanied by environmental problems related to the increase in population and industrial production and the depletion of nonrenewable resources [4]. The growing range of environmental restrictions, including high and unstable energy prices [5], restrictive environmental regulations [6,7], the limited possibility of using natural resources [8], as well as increasing environmental awareness of consumers [9,10] imply the necessity to include ecological issues in the activities undertaken by companies. After years of ignoring the ecological consequences of their activities, most entrepreneurs are aware of the need to create innovative technological solutions leading to balancing economic and environmental benefits. This is happening in many economies and is testified by the Polish Ministry of Climate's program entitled 'GreenEvo—Green Technology Accelerator' that is prepared for Polish entrepreneurs. The aim of the GreenEvo Program, financed by the Polish National Fund for Environmental Protection and Water Management, is to support the best innovative environmental technologies developed by Polish companies and assist in the promotion and international transfer of these solutions.

Considering eco-innovation in the development strategies implemented by companies requires undertaking academic research leading to the search for an answer to the question: how to conduct eco-innovation activities effectively? The paper refers to previous studies suggesting that efficient management of eco-innovation—embedded in the organizational

1

culture [11], organizational structure [12], strategic vision [13], and strategy development processes [14]—requires developing, integrating, and modifying the resources and competences held by companies. It, therefore, requires building dynamic capabilities. The logic of the dynamic capability concept [2,15] may explain why many companies that succeeded in the past (by generating environmentally friendly innovations) have faced deepening problems resulting from their inability to adapt to changes in their environment flexibly. Moreover, since the concept of dynamic capabilities is embedded in the processes of organizational learning and knowledge management (i.e., critical mechanisms of organizational change) [16,17], an attempt was made to extend it to the role and significance of leadership strategies implemented in companies.

The paper integrates various theoretical concepts, including the innovation theory, dynamic capabilities concept, and the transactional and transformational leadership theory. Then, the impact of different leadership styles on the potential of dynamic capabilities to generate technological eco-innovations in Polish companies was described and empirically tested. The research was carried out among 54 of the most innovative Polish companies that were winners of the GreenEvo Program (1st round of research) and companies selected by the Specialist Observatory in Technologies for Environmental Protection (2nd round of research).

Despite the growing number of scientific papers dealing with the importance of dynamic capabilities, many researchers [17-20] emphasize that this concept still requires empirical confirmation. Moreover, the current scientific achievements in this area are relatively limited. They consist of primarily theoretical research and, if supported by empirical research, mostly case studies (except for [3,11,12,21]). Third, conceptual and empirical works [22,23] explore the role of dynamic capabilities mainly for generating conventional innovations. What is more, the reference of the concept of dynamic capabilities to ecoinnovative activities is a topic explored to a relatively limited extent. Some of the few examples are qualitative studies by Mousavi et al. [24] and studies described by Chen and Chang [25] and Huang and Li [26] who, however, capture dynamic capabilities as a unidimensional construct. Fourth, the literature review indicates a shortage of research that would include the issue of dynamic capabilities in the context of managing technological eco-innovation. Therefore, the results presented in the paper extend the knowledge on the mechanisms of developing such innovation among eco-leaders. Moreover, the scope of the research responds to the need, emphasized by researchers [27,28], to identify organization capabilities that are antecedents of generating and implementing eco-innovations. Finally, the inclusion of leadership styles in a multidimensional conceptualization of dynamic capabilities has—so far—not been included in any model, despite its importance emphasized by researchers [25,29] for eco-innovation.

This paper aims to fill described gap, both in terms of the theoretical and empirical layers. In summary, although categories such as 'dynamic capabilities', 'eco-innovation', 'transactional leadership' and 'transformational leadership' have been theoretically analyzed in various approaches and contexts, the study is the first to examine—to the author's best knowledge—the moderating effect of leadership styles on the relationship between multidimensional view of dynamic capabilities and incremental and radical technological eco-innovations.

2. Literature Review and Hypothesis Development

The changes that shape the business are critical determinants of the development of sustainable innovations (eco-innovations, green innovations). Eco-innovation is defined in this paper as a new solution that leads to the avoidance or reduction of adverse environmental impacts [13,29–38]. Eco-innovation can be developed by companies or non-profit organizations [39], it can occur in different sectors [37], it can be marketed or not, and it can be technological or non-technological nature [30]. Eco-innovation may concern the development of new products—including eco-design and recycling of products as a particular form of their recovery [40], new technological processes—end-of-pipe technologies

or cleaner technologies [36], new organizational methods—environmental management systems [41] and new marketing solutions—i.e., eco-labels [4].

The paper analyses technological environmental innovations (TEIs) developed by Polish companies. Due to the specificity of the GreenEvo Program, which is part of the implementation of the National Strategy of Energy Security and Environment, the simple division of technologies into integrated and end-of-pipe technologies was abandoned. However, because interviewed companies were sometimes multiple winners of the GreenEvo Program, the TEIs were divided into incremental and radical (for a given company). Incremental (evolutionary) TEIs concern progressive modifications of technological solutions known to the company [42]. Radical (revolutionary) TEIs are rarer than incremental ones, more original and complex, more challenging, and more resistant to change [43,44]. Referring to the definition of Chen et al. [13] (p. 7789), incremental TEIs concern 'environmental technology that reinforces, modifies, or extends current environmental knowledge'. Radical TEIs refer to 'environmental technology that departs from current environmental knowledge'. As both incremental and radical TEIs require the development of resources (especially knowledge resources) and competences, they were included in the research model.

2.1. Determinants of Technological Eco-Innovation

The results of numerous studies indicate that generating and implementing ecoinnovation brings significant financial benefits. It also contributes to the increase in nonfinancial efficiency, strengthening the company's reputation, and building a green image. Research confirming this conclusion was carried out on various research samples, in different national contexts, and among companies of contrasting sizes, including Polish and Hungarian publicly traded companies [45], Slovenian companies [46], Italian manufacturing firms [40], green-oriented SMEs in New Zealand [47], and Taiwanese manufacturing companies [48] as well as ICT firms [26] and Chinese companies [38]. For the eco-innovative activity conducted by companies to have an efficiency dimension, it is necessary to have a specific resource and competence base. According to innovation theory, the resources required for conducting such activities include financial, human (knowledge, skills, and experience of organizational members), and physical resources (machines and equipment for laboratories and other research and development units) [49,50]. According to the resource-based view of the firm (RBV), resources should be valuable, rare, imperfectly imitable, and non-substitutable (Barney's VRIN framework), as well as durable and not easily traded [51,52].

The importance of resources for TEIs has well-established theoretical and empirical foundations. Concerning financial resources, Segarra-Oña et al. [53] state that total expenditure on the acquisition of new technologies determines the eco-innovative orientation of firms. According to human resources, Horbach [33] argues that the improvement of knowledge capital (measured by the number of highly qualified employees) stimulates TEIs. Similarly, Triguero et al. [36] indicate that qualified managers and available technical knowledge obtained from external sources increase the implementation of these changes. Moreover, research conducted among Italian [54] and German [10] entrepreneurs indicates the critical importance of conducting R&D activity for generating TEIs.

The competence-based theory is an extension of RBV, according to which gaining a competitive advantage (thanks to eco-innovation) not only depends on the resources held by firms, but also on the developed competences [55]. Some researchers use the terms 'competences' and 'capabilities' interchangeably, defining them as bundles of skills necessary to organize resources [7], the ability to achieve something by using and coordinating a set of tangible and intangible resources [40], capabilities that result from multiple repeated actions [56] or the ability to arrange resources thanks to the use of organizational processes [5]. These processes are specific to a given company and can be developed due to complex interactions between resources [52].

Both resource-based view of the firm and competence-based theory have been criticized due to their relatively static nature. Even though the resources and competences may be a source of a firm's competitive advantage, Collis [57] states that, although they are valuable, they are not always the source of a permanent advantage and are certainly not its ultimate source. Moreover, other researchers [20,58] argue that such a perspective is inadequate to explain companies' competitive advantage in a dynamically changing environment. Following this argument, it can be assumed that resources and competences alone will not be enough to develop TEIs; it will not be sufficient to maintain a relatively permanent competitive advantage. One also (or primarily) needs other capabilities (higher-order capabilities) to dynamically develop, integrate, and modify the resource and competence base. In other words, dynamic capabilities are needed.

2.2. Dynamic Capabilities and Technological Eco-Innovation

In a landmark paper, Teece et al. [15] (p. 516) defined dynamic capabilities as 'the firm's ability to integrate, build, and reconfigure internal and external competences to address rapidly changing environments'. Following this logic, some researchers consider them as organizational skills or capabilities [59,60]. Such an approach emphasizes that dynamic capabilities not only change the resource base but must also be embedded in it and in nature-repeatable. In contrast, Eisenhardt and Martin [61] define dynamic capabilities in terms of processes (as concrete and identifiable strategic and organizational processes), while Zollo and Winter [62] consider them in the context of organizational routines, directly referring to the evolutionary perspective of change described by Nelson and Winter [63]. Finally, new definitions of dynamic capabilities characterize them much more broadly, as the orientation [64] or potential [20] of an organization, i.e., an aggregated multidimensional construct consisting of interrelated components.

According to Teece et al. [15] and Teece [2], dynamic capabilities can be disaggregated into the firm's capacity: to sense and shape opportunities and threats (sensing capability); to seize opportunities (seizing capability), and to enhance, combine, protect, and reconfigure its resource base (reconfiguring capability).

Sensing capability requires constant searching and exploration—Teece [2] analyses it in terms of creative activity—technological, market, and regulatory changes. Such a strategy not only results in undertaking R&D activities and, consequently, in developing technological competences, but also in understanding hidden demand, structural evolution of markets, and future (foreseen) environmental regulations. The more information and knowledge firms accumulate, the greater their chances of being involved in creating and developing economically viable TEIs and the lower the risk of carrying out such activities (both in terms of incremental and radical changes).

Seizing capability involves maintaining and strengthening technological competences and complementary resources and investing in selected TEIs when the opportunity arises. The most important in this context is the time (overtaking competitors), place (specific consumer groups or market segments), and the level of risk (a derivative of the amount of necessary financial outlays) of strategic activities [2].

Finally, reconfiguring capability is crucial for a flexible and anticipatory response to changes occurring both in the companies and their environment [2]. One of the traps limiting TEIs is the excessive attachment to the resource base, which is manifested by the limited scope of actions related to searching and exploration of changes. Another trap is the targeted development of technological competences conditioned by the held resources and problem-solving strategies. Such a strategy may result in companies not making the most potential opportunities, even if they recognize them early. Therefore, according to Teece's [2] logic, managers must overcome at least two limitations, i.e., cognitive limitations and limitations resulting from the excessive concentration on resources and competences, which—using the Leonard-Barton [65] terminology—may paradoxically lead to core rigidities inhibiting TEIs. The described conceptualization of dynamic capabilities (sensing, seizing, and reconfiguring) has well-established analytical foundations because researchers used it both in theoretical analyses [2,17] and in empirical research [11,12,58]. Therefore, it can be a foundation for understanding the mechanisms that determine incremental and radical TEIs. What is more, such an approach to dynamic capabilities is characterized by internal consistency and solves the ambiguity of this construct emphasized by some researchers [58,66] and facilitates its operationalization.

2.3. Leadership Style As a Driver of Technological Eco-Innovation

The critical importance of the leadership style for eco-innovation is emphasized in the literature [13,25,67,68]. Moreover, concerning the development of TEIs, researchers indicate that the success of such activities—especially given their high level of complexity—requires the support (commitment, motivation, and inspiration) of the CEO and, therefore, requires strong leadership [29,56]. Vaccaro et al. [69] argue that new solutions do not necessarily have to be developed by the CEO. Still, their role is crucial to creating an intra-organizational context conducive to experimenting and introducing new processes, innovative practices, or flexible organizational structures. In the most synthetic approach, leaders can stimulate TEIs by formulating a shared and ecological vision [29], implementing an effective incentive system [25], and building an innovative organizational culture [13].

However, considering the level of novelty and the specificity of TEIs, it may turn out that incremental change requires a different leadership style than radical innovation. Building on the work of Burns [70] and Bass [71], the study examines two leadership styles (transactional and transformational leadership) to consider how specific leadership behaviors enhance the potential of dynamic capabilities to generate TEIs. Moreover, because—concerning innovative activity—the transactional and transformational leadership theory has been empirically confirmed, it was included in the research model and associated with TEIs (Figure 1).



Figure 1. A Theoretical framework.

According to Bass's [71] model, the critical difference between transactional and transformational leader is the degree of commitment to conducting eco-innovation activities and the nature of their relationship with subordinates. Transactional leadership is based on building contractual relationships and creates a bond between the leader and subordinates based on the mutual benefits of such a relationship [69–72]. A transactional leader strengthens subordinates' involvement in the development of TEIs by awarding conditional rewards for the effective and timely implementation of tasks [69]. Besides, due to the complexity of TEIs, subordinates may be more likely to become involved in eco-innovation processes if it is associated with the possibility of obtaining tangible benefits (e.g., bonuses) offered by a transactional leader [72]. Moreover, the generation of TEIs often requires maintaining effective relationships with customers, suppliers, research and scientific institutions, participation in professional associations, and adherence to best practices [21,68].

In this context, transactional leadership can effectively establish transparent and straightforward goals and policies and the expected outcomes of collaboration [69,72]. Moreover, in changing environments, companies need stability to identify opportunities or eliminate threats [16]. Due to the technological uncertainty accompanying the processes of generating TEIs, transactional leaders (who analyze the internal and external risks associated with undertaking such activities and identify discrepancies regarding agreed contracts) may favor its maximal reduction. Finally, transactional CEOs (rewarding proactive attitudes of subordinates and punishing them for reactive actions) can support the processes of strengthening the competences necessary to generate TEIs and reduce organizational inertia [73]. It can be assumed that task- and outcome-oriented transactional leaders will strengthen the potential of dynamic capabilities for the generation of TEIs. Based on this assumption, the following research hypotheses were developed:

Hypothesis 1 (H1). *Transactional leadership amplifies the positive relationship of sensing capability with (a) incremental TEIs and (b) radical TEIs.*

Hypothesis 2 (H2). *Transactional leadership amplifies the positive relationship of seizing capability with (a) incremental TEIs and (b) radical TEIs.*

Hypothesis 3 (H3). *Transactional leadership amplifies the positive relationship of reconfiguring capability with (a) incremental TEIs and (b) radical TEIs.*

Transformational leadership is not so much based on external motivation (exchange relationships) but internal one [69–71,74–76]. It is based on the assumption that the CEO considers the interests and the psychological needs of subordinates [76]. Thus, the emphasis is not on the consent or submission of employees but on their commitment, which strengthens their feeling that they go beyond self-interest to achieve the team's goals and/or the company [74]. The impact of transformational leadership on TEIs is manifested in four dimensions: inspirational motivation, individualized consideration, idealized influence, and intellectual stimulation. Thanks to their inspirational motivation, CEOs are a catalyst for TEIs, because they inspire and motivate subordinates expressing high expectations and giving meaning to their work [75]. Through individualized consideration, leaders treat each subordinate exceptionally, considering their individual needs and development paths [76]. Thanks to idealized influence, respected and trusted leaders are an authority for subordinates, making employees want to emulate them [64]. Finally, intellectual stimulation means that leaders encourage their subordinates to think creatively and innovatively [25,74] and mobilize them to seek environmentally safe—and not just economically effective—technological solutions. The climate of trust built by transformational CEOs [11] makes it possible to notice and understand the changing market, technical conditions, as well as potential threats essential to eco-innovative activity [26]. Likewise, supportive leaders facilitate the intensification of learning processes and the sharing of knowledge and experience [69]. Concerning the generation of TEIs, it is-therefore-necessary to adopt a broader perspective, taking into account not only the resources and competences, but also the learning processes, as well as the processes of coordination, integration, and reconfiguration of internal and external knowledge [15]. As argued by Wolter and Veloso [77], in this perspective, knowledge plays an equally important role as technology. In other words, transformational leaders-by supporting the orchestration processes of resources and competences [2], as well as (or primarily) strengthening the incentives for creative, productive, and innovative thinking [25,74] and mobilizing subordinates to question the status quo [76]—will enhance the potential of dynamic capabilities to generate TEIs. Based on this assumption, the following research hypotheses were developed:

Hypothesis 4 (H4). Transformational leadership amplifies the positive relationship of sensing capability with (a) incremental TEIs and (b) radical TEIs.

Hypothesis 5 (H5). *Transformational leadership amplifies the positive relationship of seizing capability with (a) incremental TEIs and (b) radical TEIs.*

Hypothesis 6 (H6). *Transformational leadership amplifies the positive relationship of reconfiguring capability with (a) incremental TEIs and (b) radical TEIs.*

3. Method—Sample, Data Collection, and Measures

To test the research hypotheses, a survey was conducted among the most innovative Polish companies generating TEIs. Non-random sample selection was used due to the requirement to meet restrictive conditions (companies participating in the study should be eco-leaders on the Polish market). The consequence of the adopted research methodology (and, at the same time, a significant limitation of the research) is the limited representativeness of the sample resulting from its arbitrary selection and its relatively small size. Both limitations significantly hinder statistical inference.

The survey was carried out from January to February 2019 and covered all GreenEvo winners (n = 66). As a result, the research sample includes companies involved in the TEIs development, which is confirmed by winning the competition and not only declarative assurances of the respondents. The survey was carried out using a Computer Assisted Telephone Interview. Since only 38 completed questionnaires were obtained (the response rate was 57.6%), it was decided to conduct the second round of survey among the most eco-innovative Silesian companies generating TEIs (n = 32). The companies were selected by the Specialist Observatory in Technologies for Environmental Protection (it was established in 2013 and was part of the 'Network of Regional Specialist Observatories in the Entrepreneurial Discovery Process' Project, implemented in 2017–2019 and co-financed by the European Regional Development Fund). The result of both rounds of surveys was obtaining 54 completely filled-in questionnaires. The descriptive characteristics of the research sample are presented in Table 1. The average organization size is 54 employees, and the average existence of an organization in the market is 19 years.

Green Technologies% of Sample 1Air protection5.5Renewable energy sources37Solutions supporting energy saving28Waste management9

Table 1. Descriptive characteristics of the sample (n = 54)

Water and sewage management

Biodiversity protection

¹ Percentages do not add up because respondents could select more than one answer.

The first stage of the research methodology was the development of a structured questionnaire using a 7-point Likert scale (from 1—'I strongly disagree' to 7—'I strongly agree' with a given statement). The original questionnaire was prepared in English and then translated into Polish. For this reason, the adequacy of the questions was assessed by analyzing them with five experts dealing with innovation and environmental changes. The received expert opinions helped improve several statements to ensure their clear and unambiguous understanding.

20 9

Several standard procedures were used in the study to increase response quality and to avoid the problem of common method bias. As Zhou et al. [3] (p. 738) notice 'using a single survey respondent as the source for both the independent and dependent data may cause the possibility of common method bias'. Therefore, two independent respondents from each surveyed company were examined. Members of the management team answered questions concerning dynamic capabilities and the TEIs, while lower-level managers' answers were the basis for measuring moderating variables. Both groups of respondents were guaranteed full confidentiality and anonymity of their answers. Finally, confirmatory factor analysis was performed, the internal consistency of the research tool was assessed, and the reliability and credibility of the measurement scales were checked. As shown by Kiefer et al. [78] such a procedure is associated with certain limitations, resulting from different respondents' perceptions, which is not unusual in the research on eco-innovation.

The survey questionnaire was divided into four modules. The first module included the measurement of multi-dimensional dynamic capabilities and referred to: (1) sensing capability, (2) seizing capability, and (3) reconfiguring capability. For the first two dimensions, the scales of Wilden et al. [12] were used. An essential premise for this choice was that the developed measuring scales were re-tested by Fainshmidt and Frazier [11]. The original scale developed by Wilden and Gudergan [21] was adapted for the measurement of reconfiguring capability. This scale is more detailed than the 2013 scale and allows capturing a broader spectrum of issues. The second module concerned the measure of technological eco-innovations developed by Polish companies. For both incremental and radical TEIs, the modification of Subramaniam and Youndt's [43] measurement scales was used [13]. The questionnaire included six statements (three for each dimension) regarding the development of TEIs, with varying degrees of novelty. In module three, respondents were asked to characterize their company's leadership style. Due to the necessity to limit the number of questions, the measurement of leadership with the original research tool (Multifactor Leadership Questionnaire) was abandoned. Instead, the questionnaire included six transformational leadership statements developed by Chen and Chang [25] and five transactional leadership statements, a modification of the measurement scale developed by Podsakoff et al. [79]. An essential premise of this choice was that Bass [71] originally distinguished only two aspects of transformational leadership, i.e., intellectual and emotional aspects, considering the dimension of charisma and inspirational motivation. Moreover, in subsequent studies, Bass showed the empirical lack of independence of both dimensions, emphasized by Waldman et al. [74]. Moreover, the meta-analysis by Lowe et al. [80] showed a lack of independence between individualized consideration and other transformational behaviors.

Finally, the statistical part of the questionnaire covered control variables, particularly the age and size of the company (measured by the number of employees) and the technological domain of their activity.

4. Analyses and Results

The Confirmatory Factor Analysis (CFA) was performed to assess the validity and reliability of measures. The obtained results showed that (concerning all variables) the degree of matching of the model to empirical data is satisfactory, which is confirmed by the theoretically proposed conceptual dimensions. Then, to assess convergent validity, the following parameters were calculated: Cronbach's Alpha (CA), composite reliability (CR), average variance extracted (AVE), and maximum shared variance (MSV). For all constructs, reliability parameters (CA and CR) reached values above the required thresholds of 0.7 [21]. The AVE surpassed the threshold of 0.5 for all scales [12]. Moreover, the condition relating to discriminant validity was met, stating that AVE should be higher than MSV. Taken together, these results show that the developed theoretical model largely overlaps with the empirical model (Table 2).

To test the research hypotheses, an analysis was performed based on hierarchical regression models, for which incremental and radical TEIs were considered as dependent variables. Following Aiken and West [81], the variables were centered to reduce the potential problem of multi-collinearity. Table 3 shows the results of hierarchical multiple regressions.

Construct	Source	Mean	SD	CA	CR	AVE	MSV
Sensing capability	[12]	4.42	1.17	0.725	0.815	0.527	0.069
Seizing capability	[12]	5.69	0.48	0.825	0.854	0.595	0.069
Reconfiguring capability	[21]	3.45	0.97	0.863	0.878	0.510	0.048
Incremental TEIs	[13]	4.17	1.54	0.701	0.835	0.628	0.002
Radical TEIs	[13]	4.91	1.76	0.871	0.917	0.786	0.002
Transactional leadership ¹	[74]	4.81	1.18	0.853	0.843	0.520	0.028
Transformational leadership	[22]	4.96	0.76	0.961	0.950	0.760	0.028

Table 2. Measurement scales and reliability.

¹ Regarding transactional leadership, it turned out that Cronbach's alpha calculated after removing the last item from the scale increased from 0.853 to 0.888. Such results indicate that this item negatively affects the consistency of the entire scale and constitute a premise for its removal from further statistical analyses.

Table 3.	The	results	of	the	Hiera	rchical	Regro	ession	Analysis	;.

Construct		Iı	ncremental TE	Is	Radical TEIs		
	Construct	Model 1A	Model 1B	Model 1C	Model 2A	Model 2B	Model 2C
	Organizational Age ^	-0.001	-0.061	0.020	-0.034	-0.029	-0.003
Control	Organizational Size ^	0.184	0.251	0.172	0.017	0.029	-0.015
Variables	Technological domain 1	-0.090	-0.076	-0.084	-0.037	-0.079	-0.011
	Technological domain 2	-0.021	0.025	0.023	-0.139	-0.116	-0.115
Sensing Capab	pility (SNC)	-0.010	1.426	0.379	0.323 **	-1.016	0.136 *
Transactional 1	Leadership (TRC)		1.023 *			-0.338	
Transformation	nal Leadership (TRF)			0.506			0.087
Interaction	SNC*TRC		-1.807			1.588	
Effects	SNC*TRF			-0.638			0.408 *
R ²		0.050	0.120	0.075	0.136	0.338	0.331
ΔR^2		-0.049	-0.014	-0.066	0.046	0.238	0.229
F		0.501	0.898	0.533	1.509	3.360	3.251
	Organizational Age ^	-0.020	-0.008	-0.016	-0.133	-0.139	-0.127
Control	Organizational Size ^	0.149	0.150	0.150	-0.028	0.019	0.005
Variables	Technological domain 1	-0.078	-0.072	-0.074	-0.017	-0.033	0.032
	Technological domain 2	0.009	0.010	0.014	-0.115	-0.083	-0.065
Seizing Capab	ility (SZC)	0.299 **	1.549 **	1.321 **	0.426 **	1.174	0.834 **
Transactional 1	Leadership (TRC)		2.238 *			1.773	
Transformation	nal Leadership (TRF)			1.119 **			1.608
Interaction	SZC*TRC		-2.691*			-1.748	
Effects	SZC*TRF			-1.103 **			-1.584 **
\mathbb{R}^2		0.135	0.218	0.273	0.209	0.295	0.365
ΔR^2		0.045	0.099	0.162	0.127	0.187	0.269
F		1.501	1.832	2.468	2.543	2.745	3.785
	Organizational Age ^	-0.110	-0.232	-0.214	-0.283	-0.218	-0.224
Control	Organizational Size ^	0.260	0.362 **	0.338 **	0.145	0.125	0.108
Variables	Technological domain 1	-0.068	-0.021	0.012	0.001	-0.046	0.017
	Technological domain 2	-0.016	-0.029	0.011	-0.151	-0.102	-0.119
Reconfiguring Capability (RCC)		0.304 **	2.326 **	2.140 **	0.493 ***	0.591 **	-0.378
Transactional Leadership (TRC)			1.139**			0.370	
Transformational Leadership (TRF)				1.488 **			-0.048
Interaction RCC*TRC			-2.537 **			-0.294 **	
Effects	RCC*TRF			-2.765 *			0.954
R [∠]		0.133	0.290	0.258	0.254	0.374	0.290
ΔR^2		0.042	0.182	0.145	0.177	0.278	0.182
F		1.467	2.679	2.288	3.274	3.919	2.682

^ Natural logarithm. Technological domain: 1—Water and sewage management. Technological domain 2—Biodiversity conservation. The estimation of the parameters for adjusting moderation models to empirical data is based on the use of the least squares' method. * p < 0.1 ** p < 0.05 *** p < 0.001.

An essential part of the analysis was the identification of direct relationships between dynamic capabilities and the generation of both incremental and radical TEIs. Model 2A indicates that sensing capability directly and significantly (p < 0.05) affects radical TEIs ($\beta = 0.323$). In regards to incremental TEIs, no significant relationship was identified (p = 0.946). Seizing capability directly and significantly affects the generation of both incremental and radical TEIs ($\beta = 0.299$; 0.426). Reconfiguring capability directly and significantly affects both types of eco-innovation (the values of the β coefficients are: 0.304 and 0.493, respectively).

To test the research hypotheses, the remaining hierarchical regression models were assessed. An analysis of them shows that, for incremental TEIs, the coefficients of interaction between sensing capability and transactional and transformational leadership are statistically insignificant (p > 0.1). Similarly, concerning radical TEIs, the interaction factor between sensing capability and transactional leadership is insignificant (model 2B). In turn, the estimation of model 2C only allows for the assumption that there is a tendency for the moderating influence of transformational leadership on the analyzed dependence ($\beta = 0.408$, p = 0.084). Therefore, the obtained results do not justify the confirmation of H1 and H4, assuming that (transactional and transformational) leadership style amplifies the positive relationship of sensing capability with (a) incremental TEIs and (b) radical TEIs.

In contrast, the results presented in Table 3 show that the interaction coefficients between seizing capability and transformational leadership are significant in both model 1C and 2C (p = 0.011 and 0.016, respectively). Such results confirm H5, which states that transformational leadership amplifies the positive relationship of seizing capability with (a) incremental TEIs and (b) radical TEIs. In turn, by narrowing the analysis to transactional leadership, the estimation of model 1B only allows assuming the tendency for the existence of a moderating influence of this leadership style on the relation between seizing capability and incremental TEIs ($\beta = -2.691$, p = 0.080), which—as in the case of sensing capability—does not constitute a basis for concluding the nature of this relationship. Finally, the data included in Table 3 (model 2B) shows that the interaction coefficient between seizing capability and transactional leadership is statistically insignificant. Therefore, the obtained results do not justify the confirmation of H2, which states that transactional leadership amplifies the positive relationship of seizing capability with (a) incremental TEIs and (b) radical TEIs.

Regarding reconfiguring capability, the coefficients of interaction between this dimension and transactional leadership are statistically significant, both in model 1B and 2B (p = 0.049 and 0.027, respectively). In other words, the obtained results confirm H3, assuming that transactional leadership amplifies the positive relationship of reconfiguring capability with (a) incremental TEIs and (b) radical TEIs. The analogous analysis carried out for transformational leadership (model 1C) only indicates the tendency to have a moderating influence of this leadership style on the relationship between reconfiguring capability and incremental TEIs ($\beta = -2.765$, p = 0.057). In turn, as shown by the estimation of model 2C, the interaction coefficient between reconfiguring capability and transformational leadership is statistically insignificant. Therefore, the obtained results do not justify the confirmation of H6, assuming that transformational leadership amplifies the positive relationship of reconfiguring capability with (a) incremental TEIs and (b) radical TEIs.

5. Discussion

The objective of the paper was to investigate how different leadership styles affect the potential of dynamic capabilities to generate technological environmental innovations in Polish companies. The results confirm (overwhelmingly) the importance of dynamic capabilities for both incremental and radical TEIs. Only concerning sensing capability and incremental TEIs, no statistically significant relationship was noted. Such results mean that companies with a strong ability to monitor the environment will better understand what the stakeholders expect and meet these needs by generating (according to the research radical rather than incremental) TEIs. This conclusion can also be linked with the concept of organizational learning [82] and with the research stream relating it to innovation [17]. Although both exploration and exploitation play an essential role in undertaking innovative activities, the main effect of exploitation is incremental innovation. The result of exploration is the development of radical solutions [83,84]. It can be assumed that radical TEIs require organizational procedures that support the search and detection of new external knowledge while using internal knowledge [85]. In turn—according to the obtained results—relying on internal knowledge may be enough to develop less original and less complex TEIs successfully. Moreover, the highest statistical significance was noted for the relationship between seizing and reconfiguring capability and radical TEIs. Therefore, the results show that investing in green research and development and increasing interdisciplinary cooperation in developing new environmental technologies are the critical conditions for creating original and complex TEIs.

5.1. Implications for Theory

A significant theoretical contribution of this paper is the confirmation—in the conditions of the Polish economy—of the usefulness of operationalization of individual variables. The reliability of the applied measurement scales indicates that the developed research model can be used in subsequent studies confirming the strength and direction of the analyzed relationships. Moreover, the results of the conducted research indicate the existence of several interesting theoretical implications relating to individual dimensions of dynamic capabilities.

5.1.1. Sensing Capability

The identified lack of statistically significant interactions between sensing capability and the leadership styles suggests that identifying new opportunities and eliminating threats (in conducting effective eco-innovation activities) may require changing the concept of leadership adopted in this paper. In other words, the described dimensions of leadership may not be theoretically appropriate for the context of dynamic search for opportunities to create TEIs. Secondly, even with the failure to build contractual or emotional relations between leaders and subordinates, they may undertake cooperation (in eco-innovation) with customers, suppliers, and R&D partners. It would not necessarily result from the proactive attitudes of employees (being a consequence, rather than a condition for undertaking cooperation), but from the desire to maintain a positive image of the company [41] or meet the expectations of key stakeholders [74]. Thirdly, the dimensions of transactional and transformational leadership can complement each other. This joint style of leadership is known as situational leadership [86] and has much in common with the recently studied enabling leadership [87]. What is more, leadership may be direct (close leadership) or indirect (distant leadership), in which the influence of leaders on subordinates occurs in a cascade through the decisions of middle and lower-level management [75]. In this approach, the transactional (or transformational) leadership style does not necessarily condition the same behavior at lower levels in the management hierarchy. Finally, it may turn out that other leadership characteristics affect the relationship between sensing capability and generating TEIs. Such a conclusion refers to Teece [2] and Fainshmidt and Frazier [11], who argue that the effectiveness of dynamic capabilities is an effect of the skills, knowledge, and experience of senior managers. Therefore, when interpreting the results, it is worth referring to the analysis of the impact of the management team's characteristics on undertaking innovative activities. Numerous studies (e.g., [88]) explain why the same environment is perceived by some managers as a source of opportunities, while for others, it is synonymous with danger. Such divergent managerial attitudes lead to the general conclusion that environment is usually interpreted by the CEO through mental 'filters', resulting from their age and level of education [89], managerial tenure [90], and professional experience [91].

5.1.2. Seizing Capability

The critical difference between a transactional and transformational leader is the nature of their relationship with subordinates. A transformational leader promotes employees' creativity, inspires them to think creatively, and mobilizes them to undertake research and experimental work [25,72]. Such activities (focused on the effective use of opportunities) result—as the research shows—in incremental and radical TEIs. These results partially support the arguments of Ambrosini et al. [18], according to which managerial perceptions of the need to change are crucial for developing dynamic capabilities. Moreover, the results obtained are consistent with the findings of Lopez-Cabrales et al. [92], who empirically proved that transformational CEOs' leadership style is positively related to seizing capabilities.

Thanks to inspirational motivation and individualized consideration, a transformational leader strengthens the personal involvement of subordinates in conducting environmental R&D activity and increases its effectiveness (by stimulating subordinates' sensitivity to specific ecological problems). Moreover, since generating TEIs is risky, a charismatic leader may facilitate the implementation of selected technological projects by formulating a clear vision integrated with the green strategy (common ecological goals, collective values, and desired organizational behavior [75]). Finally, through intellectual stimulation, a transformational leader strengthens the subordinates' sense of the economic importance of the opportunities and the environmental and social outcomes of the projects. This is especially important when undertaking experimental activities with results that are difficult to predict. Thus, the results complement the research conducted by Akkaya [93], indicating the critical role of transformational leadership in building dynamic capabilities by creating an organizational environment in which employees are encouraged, motivated, and inspired, as they are open to change and innovation.

Interpretation of the results is related to the specificity of activities leading to the use of opportunities and their time horizon. The logic of transactional and transformational leadership indicates that the former is based on the pursuit of personal (and assumed short-term) benefits by both the leader and their subordinates. In contrast, transformational leadership enables the organization's long-term goals to be achieved as a whole [74]. Since conducting environmental R&D is assumed to be long-term in nature, it can be expected that transformational (not transactional) leadership will strengthen the effectiveness of developing both incremental and radical TEIs.

5.1.3. Reconfiguring Capability

Despite the suggestion that transactional leadership may be less effective (in the context of enhancing the effectiveness of innovative activities) [72], the results indicate that it is transactional behavior that strengthens TEIs due to reconfiguring resources and competences. Such results do not support the findings of Lopez-Cabrales et al. [92], according to which not transactional but transformational leadership style is positively related to reconfiguration capability.

However, even a small change of processes, procedures and organizational structures is often costly and challenging (as it involves questioning conventional ways of thinking), deviating from proven (and routine) problem-solving methods [2]. Consequently, it can lead to increased anxiety within the organization. Therefore, the critical factor in this context is the scope and manner of influence of the CEO, who, thanks to a precise formulation of goals, directions, pace, and anticipated effects of such changes, can mitigate the negative attitudes of subordinates. The results obtained are consistent with discussion by Schoemaker et al. [94] on how dynamic capabilities, individual leadership skills, and strategic innovation are connected in a volatile, uncertain, complex, and ambiguous environment. Researchers indicate that crucial leader's characteristics that support reconfiguration are alignment and learning. Both refer to rather transactional actions such as ensuring challenging issues are surfaced to pinpoint misalignment, provide a compelling strategic vision, encourage and exemplify transparent and rigorous debriefs, and stay agile course-correct quickly if off track.

Since strong resistance to change also results from psychological barriers, the most effective way to overcome it may be an equivalent exchange of benefits (which is the basis of transactional leadership). As emphasized by Kang et al. [72], in conditions of considerable uncertainty (referring to employees' concerns regarding their further employment), employees of new and innovative firms may be more interested in obtaining tangible benefits offered for the effective and timely implementation of their tasks.

Effective management of uncertainty (as well as overcoming the psychological barriers of subordinates) is much easier in innovative culture, i.e., an organizational culture, in which innovation is recognized as both a desired and a regular pattern of organizational changes. In such a way, the results obtained are consistent with the arguments of Vera and Crossan [95], who indicate that transactional leaders seek to strengthen not only strategy and structure but also the organization's culture.

Creating an innovative culture (including values, beliefs, and organizational symbols) is complex and requires the CEO to stimulate innovative attitudes of subordinates. Such motivating does not necessarily result from building emotional ties between the leader and subordinates but can (just as effectively) be based on the immediate satisfaction of their needs. Moreover, transactional leadership is not limited to rewarding (and punishing) the fulfillment of agreed contracts. It also considers the exchange of values (responsibility, honesty) between the leader and employees. Therefore, both transformational and transactional leadership may imply creating an organizational context supporting the involvement of subordinates in the implementation of a long-term green vision of organizational development.

5.2. Managerial Implications

The research results provide valuable insights to managers who wish to invest in both incremental and radical technological environmental innovations. While there is no universal recipe for success, this paper shows that building dynamic capabilities is positively related to TEIs. Thus, from a practical point of view, the sometimes elusive and abstract concept of dynamic capabilities may strengthen the effectiveness of companies' eco-innovative activities.

Moreover, research results suggest that the CEO's leadership style significantly determines the potential of dynamic capabilities to generate TEIs. In other words, not only the support and empathy of leaders but also establishing formal control and incentive mechanisms can help build an organizational climate that supports the use of dynamic capabilities in the TEIs development process. Building emotional relations between the leader and subordinates may increase eco-innovation activities by motivating and inspiring project members, investing in environmental R&D activities, and implementing individual development paths of employees involved in designing TEIs. On the other hand, building contractual relations between the leader and subordinates will be more critical in a periodic reconfiguration of the organizational structure. It can be done by creating new departments dealing with ecological solutions or appointing interdisciplinary teams dealing with the development of technologies and including specialists in environmental protection. These activities can help managers effectively use emerging opportunities and eliminate threats and configure (and reconfigure) their assets to flexibly respond to technological and competitive challenges of the future.

5.3. Limitations and Future Research

The limitations of the conducted research can be related to two categories concerning the model and research methodology. One of the most critical substantive limitations is popular in the literature view on dynamic capabilities by Teece [2]. In this way, the testing of hypotheses relating to other dimensions of dynamic capabilities was purposely abandoned. Moreover, while the direct and positive influence of these capabilities on TEIs has been identified, developing them also carries costs that should be considered in future research. It would also be interesting to present dynamic capabilities not only through the prism of their multidimensionality but also the importance of each dimension perceived by the respondents. In such an approach, the conceptualization of dynamic capabilities should constitute a non-linear function of individual components. Moreover, their operationalization may require estimating different weights for each dimension [20]. Secondly, the paper is only limited to technological environmental innovations and does not consider other types of changes (e.g., organizational innovations). Thirdly, it views transactional and transformational leadership as two different leadership strategies. The suggested situational character of leadership indicates the limited effectiveness of a single leadership style for sensing capability. Finally, future research may consider other variables moderating the assumed relationships.

Concerning the research methodology, the most important objection is the limited representativeness of the sample and its relatively small size. Hence, the likely extension of the conducted research will be their replication in other contexts and on other (larger) populations. Secondly, because the research used subjective methods of measuring variables, it should complement them with objective indicators that could be verified a priori. Thirdly, the acquired data is cross-sectional. As Dangelico et al. [85] (p. 503) state: 'although this approach is quite common among academic studies, scholars have raised some concerns about its validity that relate to causal inference'. Therefore, longitudinal studies should be a necessary extension of the performed analyses. An interesting direction for future research may also be to conduct qualitative research-based, for example, on in-depth interviews or case studies.

6. Conclusions

The research results indicate the necessity to consider different leadership styles in the analysis of the potential of dynamic capabilities for the generation of technological eco-innovations. The effectiveness of dynamic capabilities depends on specific social interactions, which is one of the most critical challenges for the CEO. On the other hand, the relationships between leaders and subordinates are the basis for implementing long-term development strategies, including the generation of TEIs. An important implication resulting from research is the confirmation of a given leadership style's ambiguous influence on the relationship between dynamic capabilities and TEIs. In other words, the analysis results may contribute to a broader discussion on the specifics of an effective eco-innovative activity in which a given leadership style is characterized by the greatest organizational effectiveness.

Funding: This research was supported by the Faculty of Organization and Management, Silesian University of Technology [statutory research BK-235/ROZ-1/2020].

Conflicts of Interest: The authors declare no conflict of interest. The funders had no role in the design of the study; in the collection, analyses, or interpretation of data; in the writing of the manuscript; or in the decision to publish the results.

References

- Sáez-Martínez, F.J.; Mondéjar-Jiménez, J.; Ferrari, G. Eco-innovation: Trends and approaches for a field of study. *Innov. Manag. Policy Pract.* 2015, 17, 1–5. [CrossRef]
- Teece, D.J. Explicating dynamic capabilities: The nature and microfoundations of (sustainable) enterprise performance. *Strateg. Manag. J.* 2007, 28, 1319–1350. [CrossRef]
- Zhou, S.S.; Zhou, A.J.; Feng, J.; Jiang, S. Dynamic capabilities and organizational performance: The mediating role of innovation. J. Manag. Organ. 2019, 25, 731–747. [CrossRef]
- Albino, V.; Balice, A.; Dangelico, R.M. Environmental Strategies and Green Product Development: An Overview on Sustainability-Driven Companies. Bus. Strategy Environ. 2009, 18, 83–96. [CrossRef]
- Albino, V.; Dangelico, R.M.; Pontrandolfo, P. Do Inter-Organizational Collaborations Enhance a Firm's Environmental Performance? A Study of the Largest U.S. Companies. J. Clean. Prod. 2012, 37, 304–315. [CrossRef]

- Kesidou, E.; Demirel, P. On the drivers of eco-innovations: Empirical evidence from the UK. Res. Policy 2012, 41, 862–870. [CrossRef]
- Doran, J.; Ryan, G. The Importance of the Diverse Drivers and Types of Environmental Innovation for Firm Performance. Bus. Strategy Environ. 2016, 25, 102–119. [CrossRef]
- 8. Hart, S.L.; Dowell, G. A Natural-Resource-Based View of the Firm: Fifteen Years After. J. Manag. 2010, 37, 1464–1479.
- Bossle, M.B.; de Barcellos, M.D.; Vieira, L.M.; Sauvée, L. The drivers for adoption of eco-innovation. J. Clean. Prod. 2016, 113, 861–872. [CrossRef]
- 10. Horbach, J.; Rammer, C.; Rennings, K. Determinants of eco-innovations by type of environmental impact—The role of regulatory push/pull, technology push and market pull. *Ecol. Econ.* **2012**, *78*, 112–122. [CrossRef]
- Fainshmidt, S.; Frazier, M.L. What Facilitates Dynamic Capabilities? The Role of Organizational Climate for Trust. Long Range Plan. 2017, 50, 550–566. [CrossRef]
- 12. Wilden, R.; Gudergan, S.P.; Nielsen, B.B.; Lings, I. Dynamic Capabilities and Performance: Strategy, Structure and Environment. Long Range Plan. 2013, 46, 72–96. [CrossRef]
- Chen, Y.S.; Chang, C.H.; Lin, Y.H. The Determinants of Green Radical and Incremental Innovation Performance: Green Shared Vision, Green Absorptive Capacity, and Green Organizational Ambidexterity. Sustainability 2014, 6, 7787–7806. [CrossRef]
- Wang, C.L.; Senaratne, C.; Rafiq, M. Success Traps, Dynamic Capabilities and Firm Performance. Br. J. Manag. 2016, 26, 26–44. [CrossRef]
- 15. Teece, D.J.; Pisano, G.; Shuen, A. Dynamic capabilities and strategic management. Strateg. Manag. J. 1997, 18, 509–533. [CrossRef]
- Danneels, E. The dynamics of product innovation and firm competences. *Strateg. Manag. J.* 2002, 23, 1095–1121. [CrossRef]
 Fallon-Byrne, L.; Harney, B. Microfoundations of dynamic capabilities for innovation: A review and research agenda. *Ir. J. Manag.*
- 2017, 36, 21–31. [CrossRef]
 18. Ambrosini, V.; Bowman, C.; Collier, N. Dynamic Capabilities: An Exploration of How Firms Renew their Resource Base. *Br. J.*
- Manag. 2009, 20, 9–24. [CrossRef]
 Easterby-Smith, M.; Lyles, M.A.; Peteraf, M.A. Dynamic Capabilities: Current Debates and Future Directions. Br. J. Manag. 2009,
- S1–S8. [CrossRef]
 Barreto, I. Dynamic Capabilities: A Review of Past Research and an Agenda for the Future. J. Manag. 2010, 36, 256–280. [CrossRef]
- Wilden, R.; Gudergan, S.P. The impact of dynamic capabilities on operational marketing and technological capabilities: Investigating the role of environmental turbulence. J. Acad. Mark. Sci. 2015, 43, 181–199. [CrossRef]
- 22. Teece, D.; Leih, S. Uncertainty, Innovation, and Dynamic Capabilities. Calif. Manag. Rev. 2016, 58, 5–12. [CrossRef]
- Farhana, M.; Swietlicki, D. Dynamic Capabilities Impact on Innovation: Niche Market and Startups. J. Technol. Manag. Innov. 2020, 15, 83–96. [CrossRef]
- Mousavi, S.; Bossink, B.; van Vliet, M. Microfoundations of companies' dynamic capabilities for environmentally sustainable innovation: Case study insights from high-tech innovation in science-based companies. *Bus. Strategy Environ.* 2019, 28, 366–387. [CrossRef]
- Chen, Y.S.; Chang, C.H. The Determinants of Green Product Development Performance: Green Dynamic Capabilities, Green Transformational Leadership, and Green Creativity. J. Bus. Ethics 2013, 116, 107–119. [CrossRef]
- Huang, J.W.; Li, Y.H. Green Innovation and Performance: The View of Organizational Capability and Social Reciprocity. J. Bus. Ethics 2017, 145, 309–324. [CrossRef]
- Demirel, P.; Kesidou, E. Sustainability-oriented capabilities for eco-innovation: Meeting the regulatory, technology, and market demands. Bus. Strategy Environ. 2019, 28, 847–857. [CrossRef]
- Arranz, N.; Arroyabe, M.; Li, J.; Fernandez de Arroyabe, J.C. Innovation as a driver of eco-innovation in the firm: An approach from the dynamic capabilities theory. *Bus. Strategy Environ.* 2020, 29, 1494–1503. [CrossRef]
- Chen, Y.S.; Chang, C.H.; Wu, F.S. Origins of green innovations: The differences between proactive and reactive green innovations. Manag. Decis. 2012, 50, 368–398. [CrossRef]
- 30. Hemmelskamp, J. Environmental policy instruments and their effects on innovation. Eur. Plan. Stud. 1991, 5, 177–194. [CrossRef]
- Rennings, K. Redefining innovation—Eco-innovation research and the contribution from ecological economics. Ecol. Econ. 2000, 32, 319–332. [CrossRef]
- Chen, Y.S.; Lai, S.B.; Wen, C.T. The Influence of Green Innovation Performance on Corporate Advantage in Taiwan. J. Bus. Ethics 2006, 67, 331–339. [CrossRef]
- Horbach, J. Determinants of environmental innovation—New evidence from German panel data sources. Res. Policy 2008, 37, 163–173. [CrossRef]
- Oltra, V.; Saint Jean, M. Sectoral systems of environmental innovation: An application to the French automotive industry. *Technol. Forecast. Soc. Chang.* 2009, 76, 567–583. [CrossRef]
- Carrillo-Hermosilla, J.; del Río, P.; Könnölä, T. Diversity of eco-innovations: Reflections from selected case studies. J. Clean. Prod. 2010, 18, 1073–1083. [CrossRef]
- Triguero, A.; Moreno-Mondéjar, L.; Davia, M.A. Drivers of different types of eco-innovation in European SMEs. Ecol. Econ. 2013, 92, 25–33. [CrossRef]
- Melece, L. Eco-innovation and its development in Baltic states. Manag. Theory Stud. Rural Bus. Infrastruct. Dev. 2015, 37, 415–424. [CrossRef]

- Chan, H.K.; Yee, R.W.Y.; Dai, J.; Lim, M.K. The moderating effect of environmental dynamism on green product innovation and performance. Int. J. Prod. Econ. 2016, 181, 384–391. [CrossRef]
- Pacheco, D.A.J.; Caten, C.S.; Jung, C.F.; Ribeiro, J.L.D.; Navas, H.V.G.; Cruz-Machado, V.A. Eco-Innovation determinants in manufacturing SMSs: Systematic review and research directions. J. Clean. Prod. 2017, 142, 2277–2287. [CrossRef]
- Dangelico, R.M.; Pontrandolfo, P.; Pujari, D. Developing Sustainable New Products in the Textile and Upholstered Furniture Industries: Role of External Integrative Capabilities. J. Prod. Innov. Manag. 2013, 30, 642–658. [CrossRef]
- Frondel, M.; Horbach, J.; Rennings, K. What triggers environmental management and innovation? Empirical evidence for Germany. Ecol. Econ. 2008, 66, 153–160. [CrossRef]
- Calza, F.; Parmentola, A.; Tutore, I. Types of Green Innovations: Ways of Implementation in a Non-Green Industry. Sustainability 2017, 9, 1301. [CrossRef]
- Subramaniam, M.; Youndt, M.A. The Influence of Intellectual Capital on the Types of Innovative Capabilities. Acad. Manag. J. 2005, 48, 450–463. [CrossRef]
- Klewitz, J.; Hansen, E.G. Sustainability-oriented innovation of SMSs: A systematic review. J. Clean. Prod. 2014, 65, 57–75. [CrossRef]
- Przychodzen, J.; Przychodzen, W. Relationships between eco-innovation and financial performance—Evidence from publicly traded companies in Poland and Hungary. J. Clean. Prod. 2015, 90, 253–263. [CrossRef]
- Hojnik, J.; Ruzzier, M. The driving forces of process eco-innovation and its impact on performance: Insights from Slovenia. J. Clean. Prod. 2016, 133, 812–825. [CrossRef]
- 47. Zhang, J.A.; Walton, S. Eco-innovation and business performance: The moderating effects of environmental orientation and resource commitment in green-oriented SMEs. *R D Manag.* 2017, 47, E26–E39. [CrossRef]
- Chen, Y.S.; Chang, T.W.; Lin, C.Y.; Lai, P.Y.; Wang, K.H. The Influence of Proactive Green Innovation and Reactive Green Innovation on Green Product Development Performance: The Mediation Role of Green Creativity. *Sustainability* 2016, *8*, 966. [CrossRef]
- 49. Adams, R.; Bessant, J.; Phelps, R. Innovation Management Measurement: A Review. Int. J. Manag. Rev. 2006, 8, 21–47. [CrossRef]
- 50. Pichlak, M. The innovation adoption process: A multidimensional approach. J. Manag. Organ. 2016, 22, 476–494. [CrossRef]
- 51. Barney, J. Firm Resources and Sustained Competitive Advantage. J. Manag. 1991, 17, 99–120. [CrossRef]
- 52. Amit, R.; Schoemaker, P.J.H. Strategic Assets and Organizational Rent. Strateg. Manag. J. 1993, 14, 33–46. [CrossRef]
- Segarra-Oña, M.V.; Peiró-Signes, A.; Albors-Garrigós, J.; Miret-Pastor, M. Impact of Innovative Practices in Environmentally Focused Firms: Moderating Factors. Int. J. Environ. Res. 2011, 5, 425–434.
- Mazzanti, M.; Zoboli, R. Examining the Factors Influencing Environmental Innovations. The Fondazione Eni Enrico Mattei2006. Available online: http://www.feem.it/Feem/Pub/Publications/WPapers/default.htm (accessed on 17 April 2021).
- Hazen, B.T.; Bradley, R.V.; Bell, J.E.; In, J.; Byrd, T.A. Enterprise architecture: A competence-based approach to achieving agility and firm performance. Int. J. Prod. Econ. 2017, 193, 566–577. [CrossRef]
- Del Río, P.; Peñasco, C.; Romero-Jordán, D. What drives eco-innovators? A critical review of the empirical literature based on econometric methods. Int. J. Prod. 2016, 112, 2158–2170. [CrossRef]
- 57. Collis, D.J. Research Note: How Valuable are Organisational Capabilities? Strateg. Manag. J. 1994, 15, 143–152.
- Gajendran, T.; Brewer, G.; Gudergan, S.; Sankaran, S. Deconstructing dynamic capabilities: The role of cognitive and organizational routines in the innovation process. *Constr. Manag. Econ.* 2014, *32*, 246–261. [CrossRef]
- Zahra, S.A.; Sapienza, H.J.; Davidsson, P. Entrepreneurship and Dynamic Capabilities: A Review, Model and Research Agenda. J. Manag. Stud. 2006, 43, 917–955. [CrossRef]
- Helfat, C.E.; Finkelstein, S.; Mitchell, W.; Peteraf, M.A.; Singh, H.; Teece, D.J.; Winter, S.G. Dynamic Capabilities: Understanding Strategic Change in Organizations; Blackwell: London, UK, 2007.
- 61. Eisenhardt, K.M.; Martin, J.A. Dynamic capabilities: What are they? Strateg. Manag. J. 2000, 21, 1105–1121.
- 62. Zollo, M.; Winter, S.G. Deliberate Learning and the Evolution of Dynamic Capabilities. Organ. Sci. 2002, 13, 339–351.
- 63. Nelson, R.R.; Winter, S.G. An Evolutionary Theory of Economic Change; Belknap Press: Cambridge, MA, USA, 1982.
- 64. Wang, C.L.; Ahmed, P.K. Dynamic capabilities: A review and research agenda. Int. J. Manag. Rev. 2007, 9, 31-51. [CrossRef]
- Leonard-Barton, D. Core capabilities and core rigidities: A paradox in managing new product development. *Strateg. Manag. J.* 1992, 13, 111–125. [CrossRef]
- Strønen, F.; Hoholm, T.; Kværner, K.; Støme, L.N. Dynamic capabilities and innovation capabilities: The case of the 'Innovation Clinic'. J. Entrep. Manag. Innov. 2017, 13, 89–116. [CrossRef]
- Del Brío, J.Á.; Junquera, B. A Review of the literature on environmental innovation management in SMEs: Implications for public policies. *Technovation* 2003, 23, 939–948. [CrossRef]
- Dangelico, R.M. Green Product Innovation: Where we are and Where we are Going. Bus. Strategy Environ. 2016, 25, 560–576. [CrossRef]
- Vaccaro, I.G.; Jansen, J.J.P.; Van Den Bosch, F.A.J.; Volberda, H.V. Management Innovation and Leadership: The Moderating Role of Organizational Size. J. Manag. Stud. 2012, 49, 28–51. [CrossRef]
- 70. Burns, J.M.G. Leadership; Harper & Row: New York, USA, 1978.
- 71. Bass, B.M. Leadership and Performance Beyond Expectations; The Free Press: New York, NY, USA, 1985.

- Kang, J.H.; Solomon, G.T.; Choi, D.Y. CEOs' Leadership Styles and Managers' Innovative Behaviour: Investigation of Intervening Effects in an Entrepreneurial Context. J. Manag. Stud. 2015, 52, 531–554. [CrossRef]
- Levinthal, D.A. Organizational Adaptation and Environmental Selection-Interrelated Processes of Change. Organ. Sci. 1991, 2, 140–145. [CrossRef]
- Waldman, D.A.; Siegel, D.S.; Javidan, M. Components of CEO Transformational Leadership and Corporate Social Responsibility. J. Manag. Stud. 2006, 43, 1703–1725.
- Jansen, J.J.P.; George, G.; Van den Bosch, F.A.J.; Volberda, H.V. Senior Team Attributes and Organizational Ambidexterity: The Moderating Role of Transformational Leadership. J. Manag. Stud. 2008, 45, 982–1007. [CrossRef]
- Deconinck, J.; Beth, M. The relationship among transformational leadership, supervisory trust, performance and turnover intentions. J. Bus. Rev. 2013, 2, 205–211.
- Wolter, C.; Veloso, F.M. The Effects of Innovation on Vertical Structure: Perspectives on Transaction Costs and Competences. Acad. Manag. Rev. 2008, 33, 586–605.
- Kiefer, C.P.; Carrillo-Hermosilla, J.; del Río, P.; Barroso, F.J.C. Diversity of eco-innovations: A quantitative approach. J. Clean. Prod. 2017, 166, 1494–1506. [CrossRef]
- Podsakoff, P.M.; MacKenzie, S.B.; Moorman, R.H.; Fetter, R. Transformational leader behaviors and their effects on followers' trust in leader, satisfaction, and organizational citizenship behaviors. *Leadersh. Q.* 1990, 1, 107–142.
- Lowe, K.B.; Kroeck, K.G.; Sivasubramaniam, N. Effectiveness correlates of transformational and transactional leadership: A meta-analytic review of the MLQ literature. *Leadersh. Q.* 1996, 7, 385–425.
- Aiken, L.S.; West, S.G. Multiple Regression: Testing and Interpreting Interactions; Sage Publications: Newbury Park, CA, USA, 1991.
 March, I.G. Exploration and Exploitation in Organizational Learning. Organ. Sci. 1991. 2, 71–87. [CrossRef]
- March, J.G. Exploration and Exploitation in Organizational Learning. *Organ. Sci.* 1991, 2, 71–87. [CrossRef]
 Atrahama Cing K. Baschiler, the Completitive Rividitive Paradau in New Product International L Mark 2005. (O. (1))
- 83. Atuahene-Gima, K. Resolving the Capability-Rigidity Paradox in New Product Innovation. J. Mark. 2005, 69, 61–83.
- 84. Lennerts, S.; Schulze, A.; Tomczak, T. The asymmetric effects of exploitation and exploration on radical and incremental innovation performance: An uneven affair. *Eur. Manag. J.* **2020**, *38*, 121–134. [CrossRef]
- Dangelico, R.M.; Pujari, D.; Pontrandolfo, P. Green Product Innovation in Manufacturing Firms: A Sustainability-Oriented Dynamic Capability Perspective. *Bus. Strategy Environ.* 2017, 26, 490–506.
- 86. Thompson, G.; Vecchio, R.P. Situational leadership theory: A test of three versions. Leadersh. Q. 2009, 20, 837–848. [CrossRef]
- Schulze, J.H.; Pinkow, F. Leadership for Organisational Adaptability: How Enabling Leaders Create Adaptive Space. Adm. Sci. 2020, 10, 37. [CrossRef]
- Kiss, A.N.; Libaers, D.; Barr, P.S.; Wang, T.; Zachary, M.A. CEO cognitive flexibility, information search, and organizational ambidexterity. *Strateg. Manag. J.* 2020, 41, 2200–2233. [CrossRef]
- Bantel, K.A.; Jackson, S.E. Top management and innovations in banking: Does the composition of the top team make a difference? Strateg. Manag. J. 1989, 10, 107–124. [CrossRef]
- Damanpour, F. Organizational Innovation: A Meta-Analysis of Effects Of Determinants and Moderators. Acad. Manag. J. 1991, 34, 555–590.
- Kimberly, J.R.; Evanisko, M.J. Organizational Innovation: The Influence of Individual, Organizational, and Contextual Factors on Hospital Adoption of Technological and Administrative Innovations. *Acad. Manag. J.* 1981, 24, 689–713.
- 92. Lopez-Cabrales, A.; Bornay-Barrachina, M.; Diaz-Fernandez, M. Leadership and dynamic capabilities: The role of HR systems. *Pers. Rev.* 2017, 46, 255–276. [CrossRef]
- Akkaya, B. Review of Leadership Styles in Perspective of Dynamic Capabilities: An Empirical Research on Managers in Manufacturing Firms. J. Adm. Sci. 2020, 18, 389–407.
- Schoemaker, P.J.H.; Heaton, S.; Teece, D. Innovation, Dynamic Capabilities, and Leadership. Calif. Manag. Rev. 2018, 61, 15–42. [CrossRef]
- 95. Vera, D.; Crossan, M. Strategic leadership and organizational learning. Acad. Manag. Rev. 2004, 29, 222–240. [CrossRef]





Article The Idea of Corporate Social Responsibility in the Opinion of Future Managers—Comparative Research between Poland and Georgia

Iwona Zdonek¹, Anna Mularczyk^{1,*} and Grzegorz Polok²

- ¹ Department of Organization and Management, Institute of Economy and Informatics, Silesian University of Technology, 44-100 Gliwice, Poland; iwona.zdonek@polsl.pl
- ² Department of Public Management and Social Sciences, University of Economics in Katowice, 40-287 Katowice, Poland; grzegorz.polok@uekat.pl or grzegorz.polok@ue.katowice.pl
- Correspondence: anna.mularczyk@polsl.pl

Abstract: The article deals with the issue of corporate social responsibility (CSR) in the awareness of economics students as future managers responsible for the implementation of this idea in enterprises. Due to the fact that CSR is a source of many innovations today, the aim of the research was to learn about the awareness and opinions of students about CSR on such issues as: reasons for implementation, main activities, impact on profitability, and the need to explore and assess the usefulness of this knowledge. The research was conducted in 2019 among students from Poland and Georgia (as countries at various stages of advancement in the market economy). The results of the research were statistically analyzed in terms of obtaining answers to the hypotheses and research questions, checking the significance of differences in the answers due to nationality, and additionally, gender (chi-square test and test for two proportions). The primary contribution of this study is an exploration of the views of students on various issues of CSR and sustainability. The results showed that—according to students—ethical values, increased consumer awareness, and the desire to gain a competitive advantage are the most common reasons for the implementation of the CSR concept by companies. The vast majority of respondents indicated the need to enrich the knowledge of business ethics as an important aspect shaping the attitudes of employers and employees. The positive attitude of students regarding the impact of CSR on the company's profitability was confirmed, although the authors expected more unambiguous results. Differences were also distinguished between the responses of students from both countries on some issues, such as: opinions on the developing ethical sensitivity, the need for the credibility of the EU economic policy, ecology and relations with the local community, as well as relations with the closest stakeholders of companies. The results of the conducted research suggest the need for intense activity in the field of public awareness and the requirement of personnel training for the skilful implementation of CSR principles in enterprises.

Keywords: corporate social responsibility; the attitude of future managers towards CSR; ethical aspects of companies' operations

1. Introduction

Innovation in business is part of a self-perpetuating virtuous circle. It leads to higher productivity and efficiency, which, in turn, drives innovation [1]. Although the sources of innovation are undoubtedly new technologies [2] and new social behaviors [3,4], many contemporary researchers argue that corporate social responsibility (CSR) becomes the flywheel of many innovations. CSR covers the methods by which an enterprise can obtain added value by dealing with the social and environmental aspects of its operations. Thus, it is a concept whereby enterprises voluntarily incorporate social and environmental interests in their market action strategies, as well as in their relationships with various stakeholder groups [5]. Conducting social dialogue at the local level is not only conducive

Citation: Zdonek, I.; Mularczyk, A.; Polok, G. The Idea of Corporate Social Responsibility in the Opinion of Future Managers—Comparative Research between Poland and Georgia. *Sustainability* **2021**, *13*, 7045. https://doi.org/10.3390/su13137045

Academic Editor: Wen-Hsien Tsai

Received: 13 May 2021 Accepted: 18 June 2021 Published: 23 June 2021

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2021 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). to increasing the competitiveness of enterprises at the global level but also—importantly shaping the conditions for sustainable social and economic development [6]. According to the research [7], CSR as an innovation process is driven, on the one hand, by social and ecological values and, on the other hand, by economic value. Companies that want to develop and meet the expectations of stakeholders turn to the concept of CSR. They use social innovation to improve their supply chains and reach out to socially and environmentally conscious consumers and often work with social innovators to deliver business solutions to social needs [8]. According to the research [9], social innovation is seen as a key driver of the transition from ad hoc philanthropic programmes to genuine sustainable development strategies.

Since, in the future, economics students will especially create management personnel of various organizations, hence the state of their knowledge, as well as acceptance and willingness to use the principles of corporate social responsibility, are very important, as this will determine the actual transmission of this idea into the social and economic reality. That is why, at universities educating in the fields of economics and management, it is particularly important to explore issues related to corporate social responsibility. Realizing the importance of the problem, an attempt was made to diagnose the actual state of awareness of these students [6]. Until now, most of the research in this area has covered developed or developing countries. Comparative studies between the countries that are between this continuum are in the minority and usually involve closely neighboring countries [10,11]. Therefore, with this state in mind, we diagnosed a research gap that needs to be filled.

Analyzing the approach in the countries of the former Eastern bloc, Poland, as a rapidly developing country, and Georgia, as a country moving slower towards a market economy, were selected as representatives. This is confirmed by the ranking presented in the [12], created on the basis of GDP per capita, adjusted for purchasing power parity (PPP). Poland ranks 46th and Georgia 105th. This allows the assumption that both countries are at a different stage of building civic awareness [13,14], which is confirmed by the ranking [15], also the Global Innovation Index 2020 report [1], and this creates the main premise for the research undertaken in this article.

Bearing in mind the diagnosed research gap, the aim of this article has been divided into four parts. All of them concern the examination of the awareness and opinions of students on the following issues: (1) reasons for the implementation of the CSR idea by companies, (2) main measures characterizing the activities of companies in the field of CSR; (3) the impact of CSR on the profitability of companies; (4) the needs in the field of exploring and assessing the usefulness of knowledge on the ethical aspects of business operations.

To implement the first and second parts, we asked the following research questions:

Research Question 1: What is the awareness of economics students in terms of contemporary reasons for implementing the CSR idea by companies? Do nationality and gender influence attitudes in this regard?

Research Question 2: What, in the opinion of economics students, characterizes the activities of companies in the field of CSR? Are there any differences based on nationality or gender?

We were inspired to pose these questions by looking at the implementation of CSR in various countries from a historical perspective [16–19], showing that the origins of this concept were associated with charity activities of companies, and then with pro-employee and pro-ecological attitudes. Literature studies on the maturity of the implementation of CSR concepts in companies show that the implementation of this concept may have an incidental, tactical or strategic level [20] and largely depends on the state of the general economy and even specific industries. However, the activities of companies related to the concept of CSR always widely echo in the mass media, causing changes in the awareness of society (including consumers) regarding the importance of the CSR idea. Therefore, an

interesting question is how this awareness is shaped in Poland and Georgia among young people who associate their professional future with the business.

To implement the third and fourth parts, we asked the following research questions:

Research Question 3: How, in the opinion of young people, does the implementation of CSR principles affects the profitability of companies? Do nationality and gender influence this opinion?

Research Question 4: What attitudes are represented by young people in terms of the exploration and usefulness of knowledge about the ethical aspects of business? Do these attitudes differ according to nationality and gender?

We were inspired to pose such questions from the literature research, which shows that the idea of CSR is perceived as generating revenues, that not only cover the costs of its implementation but also allow to achieve a competitive advantage [21–27]. On the other hand, there are also studies in which the idea of CSR is perceived as a space for manipulation under the guise of altruistic concern, which makes the idea of CSR having a negative impact on financial results [28]. Therefore, it is puzzling which of these concepts is more convincing to the public.

The research questions posed by us indicate that the nature of our research is exploratory. In our research, we focus on learning about the views of students (from two countries that differ in terms of the development of the market economy) on various issues of CSR and sustainable development. The implementation of the set goals and answers to the research questions determined the organization of this article. The paper is organized as follows. Section 2 highlights hypotheses development in terms of: (1) reasons for implementing the CSR idea, (2) characteristic activities of companies implementing CSR, (3) attitudes regarding the impact of CSR on the profitability of companies, (4) attitudes towards the usefulness of knowledge in the field of CSR. Section 3 describes the methodology of the empirical research. Results are reported in Section 4. Discussion of findings and theoretical and practical implications are described in Section 5. Section 6 presents limitations and Section 7 conclusion.

2. Literature Review and Hypotheses Development

2.1. Reasons for Implementing the CSR Idea

An interesting look at the reasons for implementing the CSR concept in companies is the historical approach to the development of this concept, proposed in the paper [16]. Already in ancient Rome, charity work was visible in the form of creating asylum, houses for the poor and the old, hospitals and orphanages. In the Middle Ages, the activities of the Romans continued in municipal and religious institutions [17]. Some corporations joined charity work in the 16th and 17th centuries. In the 18th and 19th centuries, during the domination of the British Empire, the concept of Victorian philanthropy appeared. It focused on the working class and the creation of social programmes, examples of which could be seen in practice, both in Europe and in the United States of America [18,19]. The end of the 19th century and the beginning of the 20th century saw the popularization of donations to orphan shelters and the creation of model industrial communities, in order to improve the quality of life of employees in the form of building workers' housing estates [18]. In the 1920s and 1930s, business managers began to take responsibility for balancing profit maximization with creating and maintaining a balance with the demands of customers, workforce and the community [18]. Together with the development of the business during World War II and the 1940s, companies began to be perceived as socially responsible institutions, and these responsibilities were discussed more extensively. In the 1950s and 1960s, the belief arose that companies were responsible not only for their legal and economic obligations, but should be concerned with the welfare of the community and the education and happiness of their employees [16]. Moreover, in the 1970s, after numerous environmental disasters, such as the oil spill, the public was concerned about the rapid population growth, resource depletion and environmental pollution [29]. This

resulted in numerous social movements in the field of environmental protection, as well as human and labor rights. These movements continued in the 1980s, where, in addition to environmental issues, there were also protests against employment discrimination and consumer abuse. Moreover, there were protests against such activities of companies that threatened the health and safety of employees and worsened the quality of life in the city. The aggressive practices of international corporations were also publicized and criticized. Movements and social unrest led to the creation of social and environmental laws that imposed additional obligations on companies regarding environmental protection, product safety and labor rights [18].

It is not only the historical analysis of the CSR concept that gives an insight into the reasons why companies implement this idea. Interesting conclusions can also be drawn from the analysis of contemporary consumer and employee attitudes. Cone Communications/Ebiquity Global CSR research from 2015 [30] reveals that 91% of global consumers expect companies to be involved in solving social and environmental problems. Thus, the majority of global consumers (90%) would boycott the company if they found out about unethical business practices. Global consumers are also able to bear certain costs related to the CSR concept. Most of them (71%) declare that they would pay more for a socially or environmentally responsible product. 81% are able to consume less to conserve natural resources. 61% declare that they prefer to borrow or share goods rather than buy new ones. Moreover, 62% would prefer to work for a socially responsible company, even if it offered a lower salary than another socially unsuitable company. A total of 57% would accept a poorer quality product but more socially and environmentally responsible. A total of 61% of consumers use Social Media to engage in CSR. Additionally, research in Poland [6,13], and Georgia [31-33] has indicated an increase in social awareness of CSR as an important reason for the implementation of the CSR concept. Although research in Georgia [32] shows the perceived shortages by society in terms of access to information on CSR practices in Georgian companies. In Poland, some authors [13] directly refer to the fear of losing good reputation by companies as the reason for implementing CSR, thus referring to the theory of legality and its links with CSR. Additionally, the paper [34] shows that companies implementing CSR are an attractive workplace for employees, especially from the millennials generation. Thus, CSR is perceived as a way to achieve success in the modern world economy, one of the ways to build the company's image and a way to meet customer expectations [6].

The analysis of the historical approach to the development of the CSR concept and the observation of contemporary consumer attitudes, both global, as well as in Poland and Georgia, let us notice a constant increase in society's awareness of the responsibility of companies for their activities, not only in the context of law and economy. Therefore, we present a hypothesis corresponding to the first research question, in which we assume that:

Hypothesis 1 (H1). Changes in awareness and increased consumer expectations, as well as building a positive image of the company thanks to CSR are, in the opinion of students, the most common reasons for the implementation of the CSR concept by companies.

2.2. Characteristic Activities of Companies Implementing CSR

CSR research, conducted by scientists [20,35–37], and the historical view on the implementation of the CSR concept show, that we can now distinguish three distinct groups of CSR activities, resulting from the aforementioned historical approach. These are the areas of activity: (1) philanthropic, (2) pro-employee, (3) pro-ecological. Philanthropic activities mainly include: financial support for charity organizations, schools, clubs, associations, amusement and educational parks, fire brigades, IT centres, organizing internships for pupils and students. Pro-employee activities primarily include training for employees as part of encouraging career development and lifelong learning, supporting mothers' work through flexible working hours, promoting equal opportunities for women and people over 45 years, funding awards for academic achievement of employees' children, organizing employee trips. Pro-ecological activities include: implementation of environmental management systems, introduction of less energy-consuming technologies, closed production cycle, use of non-toxic, biodegradable products, recycling of local waste, recycling in cooperation with suppliers, use of recycled materials in production, use of ecological cleaning equipment and procedures, use of environmentally friendly materials (chemicals, intelligent coatings that are easy to clean), furnishing the company with heating panels, use of rainwater for washing, sponsoring environmental education programmes for children, partnership in environmental projects in the city and training employees in paying attention to ecological problems.

Some scientists subject these groups of CSR activities to a specific classification and form nomenclature for them. An example of this can be works [38,39], in which CSR activities are classified into: Corporate Operating Performance (COP) and Corporate Social Performance (CSP). COP activities primarily benefit the company's closest stakeholders, such as employees and suppliers. CSP activities are oriented towards further company stakeholders, such as local communities.

In order to organize CSR activities in Poland [20], has developed a CSR maturity model. This model makes it possible to identify incidental, tactical or strategic levels of involvement of Polish companies in CSR. The incidental level consists of activities of companies in the field of CSR defined as ad hoc, unplanned, random or non-existent. The tactical level consists of activities defined as inconsistent implementation of planned, repeated projects and instrumental motivation. The strategic level consists of activities, in which the CSR concept permeates every aspect of the organization and is integrated with the company's business strategy. Research [20] has shown that, in Poland, about half of the companies use CSR at the incidental level, about 30% at the tactical level and about 20% at the strategic level. Moreover, small and medium-sized enterprises currently show the lowest CSR maturity, which is in line with the trends prevailing also in the works of other researchers in Poland [13,37] and in the world [40-42]. Moreover, as in other countries, CSR maturity is industry-specific [40,42,43], but most industries in Poland do not have generally accepted CSR standards. The conclusion is that, although Poland has significantly improved its institutional CSR context over the last 20 years, the level of CSR practices in enterprises is still quite low. That is why CSR researchers in the Polish reality also focus on the issues of educating future managers in the field of business ethics, especially the issue of CSR [6]. In turn, Georgia lacks detailed information on what companies do in the field of CSR [32], which suggests the presence of CSR at an incidental level. What is more, research [44,45] indicates that Georgia obtained a low index in such areas as: gender egalitarianism, future orientation and performance orientation, which leads to the conclusion that the poor development of CSR in Georgia is conditioned by the cultural characteristics of the local society [44].

Despite the incidental level of CSR implementation in both Poland and Georgia, it should be emphasized that the undertaken activities in the field of CSR are accompanied by promotion. Publicizing corporate CSR initiatives has gained in importance and, according to [46], communication spending in this area are in third place in the budget positions of large corporations. Public relations tools [47], social media [48,49], websites, company reports, information on philanthropy [50], as well as corporate image and brand positioning strategies [51] are used to communicate CSR activities. Therefore, both environmental issues, as well as human and employee rights, have become a permanent fixture in the social consciousness. Therefore, we present a hypothesis corresponding to the second research question, in which we assume that:

Hypothesis 2 (H2). In the opinion of students, focusing on ethical values in human resource management and on pro-ecological solutions are the most common characteristics of companies that have implemented CSR.

2.3. Attitudes Regarding the Impact of CSR on the Profitability of Companies

CSR is considered a concept that, on the one hand, generates costs for companies, that could be borne by other institutions instead of CSR, and, on the other hand, leads to increased financial results, thanks to the competitive advantage achieved through CSR. Therefore, the question arises whether the revenues generated by CSR activity outweigh its costs. Many researchers have sought answers to this question [21–27]. For example, Hemingway and Maclagan [52] believe that CSR provides space and a cover for fraudulent activities of management, which imposes the negative nature of CSR. Similarly, [28,53,54] are sceptical about the concept of CSR, pointing to the possibility of some kind of manipulative and greenwashing practices under the guise of altruistic concern.

However, although some researchers point to the negative impact of CSR on financial results, due to manipulative practices in the field of CSR, there are also those, who perceive CSR as a source of competitive advantage for companies [55,56]. The main premise confirming the positive impact of CSR on the company's financial results is the stakeholder theory [25,26]. According to this theory, lasting relationships with stakeholders generate profit for the company, therefore managing these relationships is the basis for creating value for the company. The research [21] confirmed the validity of Freeman's theory. They consider CSR as a strategy creating legitimacy, reputation, and competitive advantages. The implementation of CSR creates satisfied key stakeholders, who ensure the effectiveness of the company's operations and reduce its costs. For example, satisfied employees provide a company with productive work and reduced employment and training costs. In turn, satisfied customers reward the company with repetitive purchasing decisions, providing it with income. Satisfied investors borrow capital at a lower rate, which lowers the cost of capital. Satisfied with ecological management, the company's community lowers advertising costs, while satisfied regular suppliers lower the costs of certification and quality. Moreover, research [46,57,58] has shown that the implementation of CSR causes the company's customers to identify with it and become branded diplomats. A similar result of the research on the company's reputation was obtained by [59-61]. The results of these studies are also confirmed by the report [30], according to which, if a company implements CSR, it can count on trust (90% of global consumers), loyalty (88%) and a positive image (93%).

Although literature studies indicate that the idea of CSR can be perceived as generating costs that reduce the profits of companies, scientists more and more often provide evidence that the implementation of the CSR concept also generates revenues that not only cover the costs of its implementation, but also allow to achieve a competitive advantage. Therefore, we present a hypothesis corresponding to the third research question, in which we assume that:

Hypothesis 3 (H3). *In the opinion of young people, the implementation of CSR principles in companies does not reduce the profitability of these companies.*

2.4. Attitudes towards the Use of Knowledge in the Field of CSR

Global emphasis on environmental, social and economic issues regulated by law, which is very often mentioned in publications concerning the study of CSR implementation, especially in Poland (cf. [6,20]) and the perspective that they should be incorporated into the processes functioning in companies, arouse in the society the need to educate its members in this area. The more so because, as mentioned in [14], incompetent CSR activities may be badly received by the society. Moreover, the more and more frequent presentation of the CSR concept as a way to achieve success in the contemporary world economy, building the company's image and meeting customer expectations increases the need for CSR activities, as it indicates a positive impact of CSR implementation on the financial results of companies. On the other hand, such works as [13,20], indicate a lack of strategy in activities related to CSR, which most probably result from the lack of knowledge on the subject. Therefore, we present a hypothesis corresponding to the fourth research question, in which we assume that:

Hypothesis 4 (H4). Most students agree that knowledge of the ethical aspects of business activities is useful in their professional life.

3. Materials and Methods

3.1. Sample and Data Collection

The research results were collected in 2019 via an electronic questionnaire. Data from Poland comes from the University of Economics in Katowice, where 482 students (307 women and 175 men) were surveyed. In Georgia, 118 students (61 women and 57 men) completed the questionnaire, from a total of four universities: Ilia Uni, Tbilisi State Uni, Technical University and Black Sea Uni. Due to the demographic differences between Poland and Georgia (the population of Georgia is much smaller than the population of Poland), the disproportion in the number of students from both countries was considered favourable—because it reflects the reality. The distribution of the number of respondents by age and gender is presented in the table below (Table 1).

Table 1. Structure of respondents.

Variable	Value	Poland	Georgia	Total
	Up to 20	59%	33%	54%
Age	21-24	38%	55%	41%
	Above 25	3%	12%	5%
C	Female	64%	52%	61%
Gender	Male	36%	48%	39%

3.2. Measures

The questions and measuring scales of the questionnaire used for the research are presented in Table 2. Moreover, the questionnaire contained a record with questions about gender and age.

Table 2. The relationship of research questions and questions in the questionnaire.

Research Question	Question	Scaling the Answers
	Select the sentences that, in your opinion, reflect the idea of Corporate Social Responsibility the best.	Multiple choice nominal scale
RQ1	What, according to you, determined the creation and development of the idea of Corporate Social Responsibility?	Multiple choice nominal scale
	According to you, what are the main reasons for companies to implement the principles of Corporate Social Responsibility	Multiple choice nominal scale
RQ2	What internal (inside a company) activities, according to you, reflect the idea of Corporate Social Responsibility in everyday business practice?	Multiple choice nominal scale
	What external (outside a company) activities, according to you, reflect the idea of Corporate Social Responsibility in everyday business practice?	Multiple choice nominal scale
RQ3	Does the introduction of the principles of Corporate Social Responsibility, in your opinion, cause lower profitability of the company?	5-point Likert scale
	Do you think that knowledge about ethical aspects of management is useful in future professional life?	5-point Likert scale
RQ4	Do you see the need to constantly develop and enrich knowledge from business ethics as an important element shaping the attitude of employer and employee?	5-point Likert scale
	Do you think that the students of the University should deepen their knowledge concerning the ethical aspects of corporate activities?	5-point Likert scale
3.3. Data Analysis

The obtained answers were subjected to statistical analysis in terms of obtaining answers to the hypotheses and research questions posed. The results were discussed taking into account the nationality and gender of the respondents, looking for hidden relationships. For this purpose, the Pearson chi-square test of independence was performed for the Likert scale questions. In cases, where the expected number obtained was too small, the Fisher exact test was applied. In turn, for questions with a multiple choice nominal scale, differences between the studied populations were searched for using the test for two proportions—separately for each possible answer. The assumed level of significance was 0.05 [62,63]. The results of the analyses are presented in charts, in accordance with the recommendations of Knafik [64,65]. The test results for multiple choice scales are presented in the butterfly chart, also known as the tornado diagram [66]. The results for the single choice scales are presented on the diverging bar chart [67].

4. Results

4.1. Reasons for Implementing the CSR Idea

In order to verify the H1 hypothesis, the respondents were asked about the main reasons for implementing CSR principles by the companies. It should be noted that the respondents chose a maximum of three answers. The frequencies of answers for Poland and Georgia were calculated and the significance of the difference between the fractions was calculated. Figure 1 presents the results.



Figure 1. The main reasons for companies to implement the principles of CSR according to students of Poland and Georgia: A—Expectation from customers; B—Developing ethical sensitivity; C—Companies realise that a purely economic approach to business is not enough to be successful in today's world; D—Companies are sensitive to social needs; E—Building the company's image; F—I have no opinion, I don't know. Note: Significant differences for the 0.05 significance level are marked in red.

The most frequently selected answer in Poland and Georgia was the statement that entrepreneurs realise that a purely economic approach in their activities is not enough to achieve success in today's world (69% and 65%). The second most frequently chosen answer was building the company's image (65% and 58%). In these respects, the respondents of both countries agreed. However, significantly more students from Poland than from Georgia marked the answer regarding developing ethical sensitivity (p = 0.0005). In Poland, no significant differences were found in the answers of men and women. The most common answers were C and E, although slightly more women chose the answer E first (68% and 63%), while men most often chose the answer C (72% and 64%). In Georgia, as in

Poland, there were no significant differences as to the substantive answers, but there were significantly more men who avoided answers (F: I have no opinion, I don't know; 18% and 5%; p = 0.0286).

Supplementary questions, verifying the H1 hypothesis, were the question about the reasons for the emergence and development of the CSR idea and the question about the understanding of CSR by the respondents. As before, these were multiple choice questions. The frequency of answers for Poland and Georgia and the significance of the difference between the fractions were calculated. Figure 2 presents the results.



Figure 2. What determined the creation and development of the idea of CSR in opinions of students of Poland and Georgia? A—Increase in social expectations; B—Changes in consumer awareness; C—The need for credibility of the EU economic policy; D—The need to create long-term economic efficiency of enterprises; E—Media operation; F—Don't know. Note: Significant differences for the 0.05 significance level are marked in red.

The most common answer in Poland was the answer B regarding changes in consumer awareness as the reason for the idea of CSR (73%). Secondly, the respondents indicated answer A, which meant an increase in social expectations (65%). Additionally, these two responses were selected—equally—in Georgia (53% and 55%). Despite a similar tendency, the respondents of both countries marked the answers A (p = 0.0382), B (p < 0.001) and C (p = 0.0106) with significantly different intensities. Proportionally more Polish students indicated A and B, while answer C (the need for credibility of the EU economic policy) was chosen more often in Georgia (40% and 28%). The distribution of answers by nationality and gender (Table 3) shows that, in Poland, the most frequent answer of women was B (75%) and A (61%), and men—A (72%) and B (69%). Notably more men chose answer A (p = 0.0142). However, significantly more women marked E (media activity; 40%: 28%; p = 0.0096). Georgian students had a slightly more gender-specific answers. The most common answer of women was B (64%), then A (54%), while men: A (56%), and then B and D equally, pointing to, apart from changes in consumer awareness, also the need to create long-term economic efficiency of enterprises (40%). Significantly more women marked the answer B (p = 0.0104), while significantly more men were undecided (F: I don't know; 21%; p = 0.0086).

The results of the answers to the question regarding the understanding of CSR by the respondents are illustrated in Figure 3.

In both countries, the most frequent answers were F and E. The majority of respondents understand CSR as running a business in such a way, as to take into account ethical values, law, respect for employees, society and the natural environment (76% in Poland and 55% in Georgia), as well as providing services and products in a way, that does not degrade the natural and social environment (58% and 53%). Although the results presented in Figure 1

show a significant difference between Poland and Georgia for the answer F (p < 0.001), it is the dominant answer in both countries. Moreover, the respondents of both countries differed significantly in terms of the frequency of selecting the answers: A (p = 0.0001), C (p = 0.0498), D (p = 0.0004) and G (p = 0.0354). In Poland, no major gender differences were noticed in the answers (Table 4). Only that more women marked answer A (p = 0.0486), while more men marked G (p = 0.0152). However, in Georgia, significant differences, confirmed by the test, concerned the answers: B (38% women and 21% men; p = 0.0478) and I (64% women and 35% men; p = 0.0017).

 Table 3. Significant gender differences: What determined the creation and development of the idea of CSR in opinion of students?

		Polanc	1	Georgia				
Responses	Percentage		Significance of the Difference	Perce	ntage	Significance of the Difference		
	Female	Male	<i>p</i> -Value	Female	Male	<i>p</i> -Value		
A: Increase in social expectations	60.91%	72.00%	0.0142	54.10%	56.14%	0.8237		
B: Changes in consumer awareness	74.59%	69.14%	0.1970	63.93%	40.35%	0.0104		
E: Media operation	39.74%	28.00%	0.0096	32.79%	24.56%	0.3242		
F: Don't know	14.01%	14.29%	0.9325	4.92%	21.05%	0.0086		

Note: Significant differences for the 0.05 significance level are marked in red.



Figure 3. Sentences that reflect the idea of CSR by nationality: A—Achieving sustainable profit, while wisely shaping relations with all stakeholders; B—A management tool that lets you use the process of building a dialogue with stakeholders, in order to improve the company's development strategy; C—The philosophy of running a business based on building lasting, transparent relationships with all interested parties: incl. with employees, customers, suppliers, shareholders, competition and the local community; D—Building a strategy of competitive advantage in the market based on ensuring lasting value for both shareholders and other stakeholders; E—Providing services and products in a way, that does not degrade the natural and social environment; F—Running a business in such a way, as to take into account ethical values, law, respect for employees, society and the natural environment; G—Building and implementing a social commitment strategy that exceeds legal obligations, for the benefit of all citizens, in accordance with socially accepted ethical standards; H—Contributing to sustainable development by working with employees, as well as local and global communities, to improve the quality of life for all citizens; I—Voluntary consideration of social, ethical and environmental aspects in business activities; J—I have no opinion, I don't know. Note: Significant differences for the 0.05 significance level are marked in red.

		Poland	1	Georgia					
Responses	Perce	ntage	Significance of the Difference	Perce	ntage	Significance of the Difference			
	Female	Male	<i>p</i> -Value	Female	Male	<i>p</i> -Value			
A: Achieving sustainable profit, while wisely shaping relations with all stakeholders	44.63%	35.43%	0.0486	19.67%	24.56%	0.5220			
B: A management tool that lets you use the process of building a dialogue with stakeholders, in order to improve the company's development strategy	26.38%	21.71%	0.2529	37.70%	21.05%	0.0478			
G: Building and implementing a social commitment strategy that exceeds legal obligations, for the benefit of all citizens, in accordance with socially accepted ethical standards	32.90%	44.00%	0.0152	44.26%	50.88%	0.4721			
I: Voluntary consideration of social, ethical and environmental aspects in business activities	49.51%	53.14%	0.4431	63.93%	35.09%	0.0017			

Table 4. Significant gender differences: Sentences that reflect the idea of CSR.

Note: Significant differences for the 0.05 significance level are marked in red.

Summarizing the presented results, it should be stated that the conducted research confirms the correctness of the H1 hypothesis. The most common reasons for implementing the idea in companies and for the development of CSR concepts in general are related, in the opinion of the respondents, to changes in awareness and increase in consumer expectations. Moreover, the respondents admit that, thanks to the implementation of the CSR concept, companies want to build their positive image.

4.2. Characteristic Activities of Companies Implementing CSR

In order to verify the H2 hypothesis, questions were asked about the internal and external activities characterizing the companies that implemented CSR. The respondents selected a maximum of 3 answers. The results are shown in figures (Figures 4 and 5)



Figure 4. Internal activities in everyday business practice that reflect the idea of CSR, according to students in Poland and Georgia: A—Human Resources management; B—Ethical programmes for employees; C—Occupational Health and Safety; D—Environmental protection management; E—Principles of corporate governance; F—I don't know. Note: Significant differences for the 0.05 significance level are marked in red.



Figure 5. External activities in everyday business practice that reflect the idea of CSR, according to students in Poland and Georgia: A—Maintaining contact with the local community; B—Good relationships with business partners and suppliers; C—Good relations with customers; D—Respecting human rights in action; E—The use of pro-ecological solutions; F—I don't know. Note: Significant differences for the 0.05 significance level are marked in red.

The most frequently selected answer regarding the company's internal activities, which, in everyday economic practice, are the means of implementation of the CSR idea in Poland, was answer B (60%), while the next ones (D, A and B) had a very similar frequency (58% and 57%). Thus, Polish students equally marked such activities as: ethical programmes for employees, human resources management, occupational health and safety, as well as environmental protection management. Whereas, in Georgia, the most common answer was occupational health and safety (C: 67%). However, significant differences in the marked responses in both countries showed D and E (for *p*-value = 0.0002 and *p* < 0.001). In terms of daily activities in enterprises, which can be seen in CSR, Polish students marked environmental management (D) much more often than their Georgian colleagues. For the latter, however, CSR was more often associated with the principles of corporate governance (E). There were no significant differences between answers and gender in Poland. However, in Georgia, definitely more women selected answer A (62%: 44%; *p* = 0.0448).

Analyzing the differences between the countries, it can be seen that, in Poland, the issue of ecology is treated with more attention than in Georgia, where the issues of occupational health and safety, as well as the principles of corporate governance come to the fore. Therefore, we have here a reference to the company's internal stakeholders and typical employee issues. The perception of CSR in Georgia in this aspect is also present in research [33], in which CSR focused on employee rights ensures a balance between work and private life and improves the health and safety of employees. The differences in terms of ecology may be explained by the observations contained in the research [44], which indicate that the low value of such an indicator as the future orientation, quite strongly related to the issue of ecology, leads to a weaker development of CSR in this area in Georgia, compared to other countries. In turn, the interest in ecological issues in Poland can be explained by research [68], which shows that the activities of Polish companies in the field of environmental protection are one of the most common and are oriented towards eco-education, optimization of resource use and elimination of waste, the use of new technologies and innovative solutions, in order to create new pro-ecological consumption patterns, formally strengthen all undertaken activities in the form of certification.

When asked what activities outside the company, in the everyday work, show the implementation of the CSR idea in both countries, the most common answer was D (67% and 69%). Respondents in both countries considered respect for human rights in action to be the most important aspect of external activities of enterprises. Significant differences were observed in answers A (p = 0.0044) and C (p = 0.0129). Definitely more Poles indicated

maintaining contact with the local community (A) as the external activities of companies under CSR, while more Georgians selected the option: good relations with customers (C). There were no significant gender differences in answers in Poland or Georgia.

Summarizing the results of the research, it can be said that, according to the opinions of students both in Poland and Georgia, orientation towards ethical values in the field of human resource management and pro-ecological solutions are the most common characteristics of companies that have implemented CSR. Thus, the obtained results confirm the second hypothesis. However, it is worth adding, when it comes to the verification of the second hypothesis, that the responses to the perception of external activities of companies implementing CSR indicate a great importance of respecting human rights in both analyzed countries. Therefore, this speaks in favour of enriching the original second hypothesis with this issue.

4.3. Attitudes Regarding the Impact of CSR on the Profitability of Companies

In order to examine the views of the respondents regarding the impact of CSR on the profitability of companies, they were asked whether the introduction of CSR principles reduces the company's profitability. They were asked to mark one of the five answers (according to the Likert scale). The results for both countries were gathered together in Figure 6. The chi-square test of independence confirmed the existence of dependence of the answers given on the respondents' country (p < 0.001).



Figure 6. Whether the introduction of the principles of CSR cause lower profitability of the company according to Georgian and Polish students. 1—No; 2—Rather no; 3—I don't know; 4—Rather yes; 5—Yes. In the circle, the average value obtained from the given question is shown.

When analyzing Figure 6, it should be noted that the question presented in it was inverted, i.e., the answers "no" and "rather no" should be treated as positive, while the answers "yes" and "rather yes" as negative. Thus, despite the fact that the respondents from both countries indicated different individual answers, taking into account their attitude: positive to CSR (answers: "no" and "rather no"), together 62% (Georgia) and 59% (Poland), and negative (answers: "yes" and "rather yes"), together respectively: 23% and 19%, their answers showed no significant difference (p > 0.05). Most of the respondents do not believe that the introduction of CSR would reduce the company's profitability. In this way, it is possible to confirm the correctness of the third hypothesis, although the authors expected that the results would be more unambiguous in the direction of the answers "no" and "rather no".

The Pearson chi-square test of independence was carried out by examining the dependence of the answers given on the gender of the respondents in individual countries. The test result indicated the existence of this relationship, both among students in Georgia (p = 0.0376) and in Poland (p = 0.007443).

4.4. Attitudes towards the Usefulness of Knowledge in the Field of CSR

In order to answer the fourth research question, a diverging bar chart was created, presenting the answers to three questions of the questionnaire (Figure 7).



Figure 7. Attitudes towards the usefulness of knowledge in the field of CSR. 1—No; 2—Rather no; 3—I don't know; 4—Rather yes; 5—Yes. In the circle, the average value obtained from the given question is shown.

The first of them was aimed at examining the awareness of whether the knowledge about ethical aspects of management is useful in later professional life. The chi-square test of independence showed significant differences in the answers in both countries (p = 0.0001588). The vast majority of students confirmed the awareness of the usefulness of this knowledge in their later professional life. There were significantly more such answers in Georgia (88%), but also in Poland they were the majority (77%). In turn, by examining the dependence of the answers given on the gender of respondents in individual countries, its existence was confirmed among students in Georgia (p = 0.03505). Definitely more women gave an affirmative answer (96%), while among the undecided there were more men (14%). Additionally, among students in Poland, the existence of a dependence between the answer and gender was confirmed (p = 0.007798). Additionally, more women (albeit with a much smaller share than in Georgia) answered affirmatively (82%), while the answers "rather no" and "I don't know" were mostly given by men.

The second question concerned the need to constantly develop and enrich knowledge of economic ethics as an important element shaping the attitude of the employer and employee. Additionally here, the vast majority of respondents felt the need to enrich their knowledge of economic ethics as an important aspect shaping the attitudes of employers and employees. As many as 74% of Georgians answered "yes" or "rather yes", while 63% of Poles chose the same answers. The chi-square test showed a relationship between the answers and nationality (p < 0.001). In Georgia, this relationship is also significant in terms of gender (p < 0.001), while in Poland it was not detected (p = 0.5364).

The third question concerned the opinion on the obligation to deepen knowledge about the ethical aspects of companies' activities. The independence test confirmed the existence of a dependence between the answers given and nationality (p < 0.001). Significantly more Georgians were in favour of the need to deepen their knowledge about the ethical aspects of business activities at universities (91%), however, the majority of Polish students also noticed this need (75%). Still, there was no relationship between the answers and gender in both countries (Georgia: p = 0.06449, Poland: p = 0.05395).

5. Discussion of Findings

Our research has confirmed the first hypothesis assuming that ethical values, changes in consumer awareness, and the willingness of companies to gain a competitive advantage thanks to CSR are the most common reasons for the implementation of the CSR concept by companies. The differences between the answers of students from both countries related to several issues. The first is the opinion on the developing ethical sensitivity, which was indicated by Polish students significantly more often than in Georgia. In turn, the issue of the need for credibility of the EU economic policy was more important for Georgian students than for Polish students. Moreover, in Poland, men more often attached importance to the issue of increasing social expectations, while women to the role of the media in developing the idea of CSR. In the case of Georgia, men more often than women indicated the issue of changing consumer awareness. Comparing the obtained results with the research [30], a certain convergence can be seen. The dominant number of answers indicating that CSR is perceived as running a business in such a way, as to take into account ethical values, law, respect for employees, society, and the natural environment, both in Poland and Georgia, shows that CSR and the ethical values it entails are perceived as a way to solve social and environmental problems. Students in both surveyed countries present a similar perception of CSR as the global consumers indicated in the research [30]. In turn, the high number of answers related to ecological issues is consistent with the research obtained by [18] and shows that, in both countries, students perceive the issues of providing products and services in a way, that does not degrade the natural environment as very important. The high percentage of answers regarding building lasting relationships with all stakeholders of the company proves the convergence of the results with the research [21], showing the importance of the stakeholder theory for the issue of generating profit for enterprises. An interesting aspect of additional research would be to answer whether students' responses from countries at a different economic level coincide with the presented ones. Therefore, we are considering the possibility of conducting similar research in countries at a lower and higher level of development.

Our research has also confirmed the second hypothesis claiming that orientation towards ethical values in the field of human resource management and pro-ecological solutions are the most common characteristics of companies that have implemented CSR in the analyzed countries. In the analysis of differences between countries, the difference in terms of the greater importance of maintaining contacts with the local community in the perception of Polish students is noticeable, and in the case of Georgian students-greater importance of relations with customers. From these differences, it can be concluded that, for students from Georgia, relations with close stakeholders of companies are more important, while for Polish students-relations with further ones. The approach of Polish students can be explained by the research described in the paper [13], in which employees of Polish branches of international corporations have a similar attitude as students. Therefore, it points to the value system currently functioning in Polish society. By planning further research in this area, we can indicate that retrospective studies could also provide worthy of note results whether CSR's perception is changing in both countries. For example, will the results from both countries be more similar in 5-10 years? In our opinion, this should as well be the direction of future research.

We have also confirmed our third hypothesis assuming a positive attitude of students as regards the impact of CSR on the company's profitability. The results of the research in both countries are consistent with the conclusions about the positive impact of CSR on the profitability of the work of companies of such researchers as [21,26,57], However, a puzzling factor is the large percentage of people in Poland, who are unable to determine the impact of CSR on profitability, which may be related to the research results presented in the paper [20], which emphasize the incidental nature of CSR activities of Polish companies.

Therefore, the lack of an unequivocal answer by 1/5 of the surveyed Polish students may indicate the awakening of awareness in this regard and the need to change the existing situation. This hypothesis, in turn, can be confirmed by similar studies carried out in a few years. Research at regular intervals could identify a trend in the development of consciousness.

Eventually, our research has also confirmed the fourth hypothesis assuming that in the opinion of students in both countries, knowledge of the ethical aspects of companies' activities is useful in professional life. The explanation of the need to enrich the knowledge and skills in the field of CSR, declared by Polish and Georgian students, can be based on research [13,14], which shows how difficult it is to build a civic community and how much knowledge is needed, so that CSR activities are not perceived as a smokescreen, greenwashing, or simply as incompetent, but in good faith [14]. Considering the results, the question arises what results would be obtained from the research on the introduction by academic teachers of economic courses of CSR content to their subjects. Is the perception of this need by academic teachers similar to the perception of students in both countries? What is the answer to the growing interest on the part of students in these issues? Should it be the adaptation of study programs, or what should be done in this matter?

Implication for Research and Practice

So far, research in the field of CSR has covered developed or developing countries. Comparative studies between the countries that are between this continuum were in the minority and usually involved closely neighboring countries. Thus, by researching countries such as Poland and Georgia, we filled the identified research gap. Our research has shown that countries slower-moving towards a market economy may have a slightly different perception of the concept of CSR. The concept of CSR can be perceived as activities directed towards closer than further stakeholders of enterprises.

Our research shows that students' awareness of CSR results from all the socially and environmentally responsible activities of companies on the market. Therefore, in the case of a megatrend related to the climate crisis and the need to adapt business to this crisis, it may happen that actions without a clear focus on ecology will be assessed as unsatisfactory. Therefore, companies should take various actions on a voluntary basis. They may concern, for example, the replacement of plastic packaging with more ecological ones, the implementation of the postulates of the zero waste concept or the circular economy.

The obtained results indicate that, although the concept of CSR is known and accepted by the societies of both countries, the complexity of issues related to its implementation in companies suggests the need to increase educational activities and educate personnel in skilfully implementing CSR principles in enterprises. It will be possible, inter alia, thanks to the expansion of the educational offer in this field at economic universities. The need to learn more about the ethical aspects of running a business reported by students in our research indicates that academic curricula should perhaps be directed to the more practical aspects of CSR. The implementation of project forms of teaching can bring satisfactory results, especially in combination with providing students with access to real problems faced by companies that want to be socially responsible. Such forms of teaching might be very well perceived by students.

6. Limitation

Although our research has shown that countries slower-moving towards a market economy may have a slightly different perception of the concept of CSR, we realize that comparative studies conducted in only two countries are insufficient to make such conclusions. Thus, we do not generalize our findings but we consider them only as a hypothesis that needs to be verified. Bearing in mind the exploratory nature of our research, we used a simple form of a questionnaire. Investigating with a more sophisticated form of the questionnaire would allow finding a broader context for the issues we are researching.

7. Conclusions

The article poses four research questions and verifies four hypotheses. The research was conducted in the context of differences related to nationality and gender. First of all, we were looking for an answer to the question of what the students' awareness is of the contemporary reasons for implementing the idea of CSR by companies. The conducted research allowed for the positive verification of the first hypothesis and, therefore, both in Poland and in Georgia, ethical values, changes in consumer awareness and the willingness of companies to gain a competitive advantage thanks to CSR are the most common reasons for the implementation of the CSR concept by companies. Secondly, we were looking for an answer to the question about the characteristic activities of companies in the field of CSR. The obtained results, especially those in terms of the perception of internal activities of companies implementing CSR, confirm the second hypothesis. Therefore, both in Poland and in Georgia, orientation towards ethical values in the field of human resource management and pro-ecological solutions are the most common characteristics of companies that have implemented CSR. However, it is worth adding, when it comes to the verification of the second hypothesis, that the responses to the perception of external activities of companies implementing CSR indicate a great importance of respecting human rights in both analyzed countries. Therefore, the verified second hypothesis should also be supplemented with this issue. In Poland, the issue of ecology and relations with the local community (and, therefore, further stakeholders of companies) is treated with more attention than in Georgia, where relations with the closest stakeholders of companies, i.e., employees and customers, come to the fore. Therefore, referring to the history of the development of the CSR concept, it can be seen that Georgia is at a slightly earlier stage of implementing this concept. Thirdly, we looked for an answer to the question of whether, in the opinion of the respondents, the implementation of CSR principles affects the profitability of companies. The obtained results indicate that it is possible to confirm the correctness of the third hypothesis, although the authors expected that the results would more clearly indicate the positive attitude of students as regards the impact of CSR on the company's profitability. The large percentage of the answers "I don't know" in Poland shows, according to the authors, the awareness of young people of the incidental nature of CSR activities, which suggests the need for intensified actions, not only to awaken public awareness, but to educate personnel, who can professionally implement CSR principles in the activities of companies. This conclusion is also confirmed by the results of research related to the last research question, i.e., related to the attitude that students present towards the exploration and usefulness of knowledge about the ethical aspects of business. The fourth hypothesis was confirmed and, therefore, according to the students, knowledge of the ethical aspects of companies' activities is useful in professional life.

Author Contributions: Conceptualization, G.P., I.Z. and A.M.; methodology, I.Z. and A.M.; formal analysis, I.Z. and A.M.; investigation, G.P.; writing—original draft preparation, I.Z. and A.M.; visualization, I.Z. and A.M. All authors have read and agreed to the published version of the manuscript.

Funding: This paper was published as part of statutory research at the Silesian University of Technology, Department of Organization and Management: 13/010/BK_21/0057.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Acknowledgments: We would like to thank Salome Svanaga and Joanna Paliszkiewicz PhD Eng., thanks to whom it was possible to conduct research in Georgia.

Conflicts of Interest: The authors declare no conflict of interest.

References

- 1. GII 2020 Global Innovation Index. Available online: https://www.globalinnovationindex.org/Home (accessed on 29 April 2021).
- Sołtysik-Piorunkiewicz, A.; Zdonek, I. How Society 5.0 and Industry 4.0 Ideas Shape the Open Data Performance Expectancy. Sustainability 2021, 13, 917. [CrossRef]
- Zdonek, D.; Król, K. The Impact of Sex and Personality Traits on Social Media Use during the COVID-19 Pandemic in Poland. Sustainability 2021, 13, 4793. [CrossRef]
- Hysa, B.; Karasek, A.; Zdonek, I. Social Media Usage by Different Generations as a Tool for Sustainable Tourism Marketing in Society 5.0 Idea. Sustainability 2021, 13, 1018. [CrossRef]
- European Commision 2001 Greenpaper. Available online: https://ec.europa.eu/transparency/regdoc/rep/1/2001/EN/1-2001-366-EN-1-0.Pdf (accessed on 29 April 2021).
- Polok, G. Rola Wyższej Szkoły Ekonomicznej w Propagowaniu Wiedzy o Społecznej Odpowiedzialności Biznesu w: Społeczna Odpo-Wiedzialność–Aspekty Teoretyczne i Praktyczne; Wyd. Uniwersytetu Ekonomicznego w Katowicach. 2011, 64, 185–205.
- Roszkowska-Menkes, M.T. Integrating strategic CSR and open innovation. Towards a conceptual framework. Soc. Responsib. 2018, 14, 950–966. [CrossRef]
- Mirvis, P.; Herrera, M.E.B.; Googins, B.; Albareda, L. Corporate social innovation: How firms learn to innovate for the greater good. J. Bus. Res. 2016, 69, 5014–5021. [CrossRef]
- Osburg, T. Social Innovation to Drive Corporate Sustainability. In Social Innovation; Springer Science and Business Media LLC: Berlin, Germany, 2013; pp. 13–22.
- Habek, P. Development of Corporate Social Responsibility in Visegrad Group Countries. Sci. Pap. Silesian Univ. Technol. Organ. Manag. Ser. 2019, 2019, 71–82. [CrossRef]
- Habek, P.; Sujova, E.; Cierna, H. Technology Comparative analysis of CSR reporting practices in Poland and Slovakia. *Sci. Pap. Silesian Univ. Technol. Organ. Manag. Ser.* 2018, 2018, 51–65. [CrossRef]
- Global Finance Magazine the World's Richest and Poorest Countries. Available online: https://www.gfmag.com/global-data/economic-data/worlds-richest-and-poorest-countries (accessed on 29 April 2021).
- Dancewicz, B.; Struve, F. The Influence of MNEs on CSR in Poland. In Corporate Social Responsibility in Poland CSR Sustainability Ethics & Governance; Długopolska-Mikonowicz, A., Przytuła, S., Stehr, C., Eds.; Springer International Publishing: Cham, UK, 2019; pp. 149–165, ISBN 978-3-030-00439-2.
- Szumniak-Samolej, J. Social Media for Corporate Social Responsibility Strategy Creation and Communication in Poland. In Corporate Social Responsibility in Poland CSR Sustainability Ethics & Governance; Długopolska-Mikonowicz, A., Przytuła, S., Stehr, C., Eds.; Springer International Publishing: Cham, UK, 2019; pp. 273–286. ISBN 978-3-030-00439-2.
- Gallup Inc. 2016 Global Engagement Report. Available online: http://www.shqiperiajone.org/sites/default/files/pdf/2016%20 Global%20Civic%20Engagement%20Report.pdf (accessed on 29 April 2021).
- Agudelo, M.A.L.; Johannsdottir, L.; Davídsdóttir, B. A literature review of the history and evolution of corporate social responsibility. Int. J. Corp. Soc. Responsib. 2019, 4, 1. [CrossRef]
- Chaffee, E.C. 2017 The Origins of Corporate Social Responsibility. Available online: https://heinonline.org/HOL/LandingPage ?handle=hein.journals/ucinlr85&div=18&id=&page= (accessed on 29 April 2021).
- Carroll, A.B. A history of corporate social responsibility: Concepts and practices. In *The Oxford Handbook of Corporate Social Responsibility*; Andrew, C.M., Matten, D.J., Moon, D.S., Eds.; Oxford University Press: New York, NY, USA, 2008; pp. 19–46.
- 19. Harrison, B. Philanthropy and the Victorians. *Vic. Stud.* **1966**, *9*, 353–374.
- Witek-Crabb, A. CSR Maturity in Polish Listed Companies: A Qualitative Diagnosis Based on a Progression Model. Sustainability 2019, 11, 1736. [CrossRef]
- Maqbool, S.; Zameer, M.N. Corporate social responsibility and financial performance: An empirical analysis of Indian banks. *Futur. Bus. J.* 2018, *4*, 84–93. [CrossRef]
- Wang, Z.; Sarkis, J. Corporate social responsibility governance, outcomes, and financial performance. J. Clean. Prod. 2017, 162, 1607–1616. [CrossRef]
- Sánchez-Torné, I.; Morán-Álvarez, J.C.; Perez-Lopez, J.A. The importance of corporate social responsibility in achieving high corporate reputation. *Corp. Soc. Responsib. Environ. Manag.* 2020, 27, 2692–2700. [CrossRef]
- Margolis, J.D.; Elfenbein, H.A.; Walsh, J.P. Does It Pay to Be Good. And Does It Matter? A Meta-Analysis of the Relationship between Corporate Social and Financial Performance. SSRN Electron. J. 2009. [CrossRef]
- 25. Freeman, R.E. Strategic Management: A Stakeholder Approach; Pitman Publishing: Boston, MA, USA, 1984.
- Freeman, R.E.; Dmytriyev, S. Corporate Social Responsibility and Stakeholder Theory: Learning From Each Other. Symphonya. Emerg. Issues Manag. 2017, 7–15. [CrossRef]
- 27. Hammann, E.-M.; Habisch, A.; Pechlaner, H. Values that create value: Socially responsible business practices in SMEs-empirical evidence from German companies. *Bus. Ethic Eur. Rev.* **2009**, *18*, 37–51. [CrossRef]
- 28. Moon, J. The Social Responsibility of Business and New Governance. Gov. Oppos. 2002, 37, 385–408. [CrossRef]
- 29. Du Pisani, J.A. Sustainable development–historical roots of the concept. *Environ. Sci.* 2006, *3*, 83–96. [CrossRef]
- Cone 2015 Global CSR Study. Available online: https://www.conecomm.com/research-blog/2015-cone-communications-ebiqui ty-global-csr-study (accessed on 29 April 2021).
- 31. Gigauri, I. Impact of Corporate Social Responsibility on Consumer Purchase Decision. J. Educ. Soc. Res. 2012, 2, 106–111.

- Gigauri, I. Communication Needs for Corporate Social Responsibility in Developing Countries: The Case of Georgia. World J. Soc. Sci. 2015, 5, 127–141.
- Gigauri, I. Corporate Social Responsibility and COVID-19 Pandemic Crisis. Int. J. Sustain. Entrep. Corp. Soc. Responsib. 2021, 6, 30–47. [CrossRef]
- Klimkiewicz, K.; Oltra, V. Does CSR Enhance Employer Attractiveness? The Role of Millennial Job Seekers' Attitudes. Corp. Soc. Responsib. Environ. Manag. 2017, 24, 449–463. [CrossRef]
- Dyczkowskaa, J.; Krasodomskab, J.; Michalakc, J. CSR in Poland: Institutional Context, Legal Framework and Voluntary Initiatives. Account. Manag. Inf. Syst. 2016, 15, 206.
- Wolczek, P. Development of the CSR Concept in Poland–Progress or Stagnation? Soc. Responsib. Organ. Chang. Their Dir. Res. Pap. Wroc. Univ. Econ. 2015, 387, 200–214. [CrossRef]
- Wołczek, P. Strategic CSR in Poland. In Corporate Social Responsibility in Poland CS, Sustainability Ethics & Governance; Długopolska-Mikonowicz, A., Przytuła, S., Stehr, C., Eds.; Springer International Publishing: Cham, UK, 2019; pp. 289–309, ISBN 978-3-030-00439-2.
- Harrison, K.; Huang, L. Not all corporate social responsibility (CSR) is created equal: A study of consumer perceptions of CSR on firms post fraud. J. Mark. Theory Pr. 2020, 1–18. [CrossRef]
- Huang, L. Quality or Responsibility? The Impact of Two Categories of Corporate Social Responsibility on the Consumer-Brand Relationship. Mark. Manag. J. 2015, 25, 27–41.
- Deegan, C.; Gordon, B. A Study of the Environmental Disclosure Practices of Australian Corporations. Account. Bus. Res. 1996, 26, 187–199. [CrossRef]
- Ali, W.; Frynas, J.G.; Mahmood, Z. Determinants of Corporate Social Responsibility (CSR) Disclosure in Developed and Developing Countries: A Literature Review. Corp. Soc. Responsib. Environ. Manag. 2017, 24, 273–294. [CrossRef]
- Holder-Webb, L.; Cohen, J.R.; Nath, L.; Wood, D. The Supply of Corporate Social Responsibility Disclosures Among U.S. Firms. J. Bus. Ethic 2008, 84, 497–527. [CrossRef]
- Cowen, S.S.; Carolina, N. The Impact of Corporate Characteristics on Social Responsibility Disclosure: A Typology and Frequency-Based Analysis. Account. Organ. Soc. 1987, 12, 111–122. [CrossRef]
- Shekriladze, M.; Stehr, C. Corporate Social Responsibility in Rising Economies: Georgia. Glob. Chall. CSR Sustain. Dev. 2020, 141–161. [CrossRef]
- House, R.J.; Hanges, P.J.; Javidan, M.; Dorfman, P.W.; Gupta, V. Culture, Leadership, and Organizations. In *The Globe Study of 62* Societies; Sage Publications: New York, NY, USA, 2004; ISBN 1-4522-0812-3.
- Parguel, B.; Benoît-Moreau, F.; Larceneux, F. How Sustainability Ratings Might Deter 'Greenwashing': A Closer Look at Ethical Corporate Communication. J. Bus. Ethic. 2011, 102, 15–28. [CrossRef]
- Grunig, L.A.; Grunig, J.E.; Vercic, D. Are the IABC's excellence principles generic? Comparing Slovenia and the United States, the United Kingdom and Canada. J. Commun. Manag. 1998, 2, 335–356. [CrossRef]
- Castronovo, C.; Huang, L. Social Media in an Alternative Marketing Communication Model. J. Mark. Dev. Compet. 2012, 6, 117–134.
- Huang, L.; Clarke, A.; Heldsinger, N.; Tian, W. The communication role of social media in social marketing: A study of the community sustainability knowledge dissemination on LinkedIn and Twitter. J. Mark. Anal. 2019, 7, 64–75. [CrossRef]
- 50. Ziek, P. Making sense of CSR communication. Corp. Soc. Responsib. Environ. Manag. 2009, 16, 137–145. [CrossRef]
- 51. Du, S.; Bhattacharya, C.; Sen, S. Reaping relational rewards from corporate social responsibility: The role of competitive positioning. *Int. J. Res. Mark.* 2007, 24, 224–241. [CrossRef]
- Hemingway, C.A.; Maclagan, P.W. Managers' Personal Values as Drivers of Corporate Social Responsibility. J. Bus. Ethic 2004, 50, 33–44. [CrossRef]
- Janney, J.J.; Gove, S. Reputation and Corporate Social Responsibility Aberrations, Trends, and Hypocrisy: Reactions to Firm Choices in the Stock Option Backdating Scandal. J. Manag. Stud. 2011, 48, 1562–1585. [CrossRef]
- Jeon, H.J.; Gleiberman, A. Examining the Role of Sustainability and Green Strategies in Channels: Evidence from the Franchise Industry. J. Mark. Theory Pr. 2017, 25, 189–199. [CrossRef]
- Falkenberg, J.; Brunsael, P. Corporate Social Responsibility: A Strategic Advantage or a Strategic Necessity? J. Bus. Ethic 2011, 99, 9–16. [CrossRef]
- 56. McWilliams, A.; Siegel, D.S. Creating and Capturing Value. J. Manag. 2010, 37, 1480–1495. [CrossRef]
- Sen, S.; Bhattacharya, C. Does Doing Good Always Lead to Doing Better? Consumer Reactions to Corporate Social Responsibility. J. Mark. Res. 2001, 38, 225–243. [CrossRef]
- Maignan, I.; Ferrell, O.C. Corporate Social Responsibility and Marketing: An Integrative Framework. J. Acad. Mark. Sci. 2004, 32, 3–19. [CrossRef]
- 59. Sagar, P.; Singla, A. Trust and corporate social responsibility: Lessons from India. J. Commun. Manag. 2004, 8, 282–290. [CrossRef]
- Fukukawa, K.; Balmer, J.M.T.; Gray, E.R. Mapping the Interface Between Corporate Identity, Ethics and Corporate Social Responsibility. J. Bus. Ethic 2007, 76, 1–5. [CrossRef]
- Hur, W.-M.; Kim, H.; Woo, J. How CSR Leads to Corporate Brand Equity: Mediating Mechanisms of Corporate Brand Credibility and Reputation. J. Bus. Ethic 2014, 125, 75–86. [CrossRef]

- Hogg, R.V.; McKean, J.W.; Craig, A.T. Introduction to Mathematical Statistics, 8th ed.; Pearson: Boston, MA, USA, 2019; ISBN 978-0-13-468699-8.
- 63. Peck, R. Introduction to Statistics and Data Analysis, 6th ed.; Cengage Learning: Boston, MA, USA, 2018; ISBN 978-1-337-79361-2.
- Knaflic, C.N. Storytelling with Data: A Data Visualization Guide for Business Professionals; John Wiley & Sons: Hoboken, NJ, USA, 2015; ISBN 1-119-00225-7.
- 65. Tufte, E.R. The Visual Display of Quantitative Information. J. Heal. Qual. 1985, 7, 15. [CrossRef]
- Kaley, A. 2020 Building Butterfly/Tornado Charts—Storytelling with Data (Part II). Available online: https://medium.datadrive ninvestor.com/building-butterfly-tornado-charts-storytelling-with-data-part-ii-ba443648b40e (accessed on 29 April 2021).
- Weitz, D. 2020 Diverging Bars, Why & How. Storytelling with Divergences. Available online: https://towardsdatascience.com/d iverging-bars-why-how-3e3ecc066dce (accessed on 29 April 2021).
- Hys, K. The Practice of CSR in Poland in the Realm of Environment Action. In Proceedings of the 17th International Multidisciplinary Scientific GeoConference (SGEM2017), Albena, Bulgaria, 29 June–5 July 2017; pp. 1035–1041. [CrossRef]





Article The Green Side of Industry: The Drivers and the Impacts of ECO-Innovations in Brazil⁺

Fernanda Q. Sperotto 1,2,* and Iván G. P. Tartaruga 1,2,*

- ¹ Centre of Studies in Geography and Spatial Planning (CEGOT), 4150-564 Porto, Portugal
- ² Faculty of Arts and Humanities, University of Porto, Via Panorâmica s/n, 4150-564 Porto, Portugal
- * Correspondence: fsperotto@letras.up.pt (F.Q.S.); itartaruga@letras.up.pt (I.G.P.T.)
- † This is a modified version of a conference document. In Proceedings of the V Industrial Economy and Innovation Meeting (ENEI): Innovation, Sustainability and Pandemic, Belo Horizonte, Brazil, 10–14 May 2021.

Abstract: This study aimed to provide an overview of eco-innovations in the Brazilian industry. To address this issue, we analyzed specific data of eco-innovative companies. In addition, we applied the cluster heatmap technique, which allowed us to analyze the different drivers and impacts of eco-innovations in different sectors. According to the results, companies that stated that innovation made it possible to reduce their environmental impact represent a third of all innovators. Moreover, they are companies that have shown greater effort to innovate and greater susceptibility to the benefits and obstacles of innovation. Furthermore, the eco-innovation strategy is mainly driven by market factors, such as reputation and codes of good practice. The impacts are mainly associated with the use of more widespread and less complex technologies, such as recycling. In addition to these results, the study considers some alternatives to guide the innovation policy, especially related to eco-innovations in semi-peripherical countries.

Keywords: sustainability; green technologies; environmental innovation; industry; Brazil

1. Introduction

The climate change scenario requires the application of a set of actions capable of tackling ecological, social, and economic problems. Harmonizing the economic, social, and environmental dimensions is at the core of the 2030 Agenda and the Sustainable Development Goals (SDGs).

The perspective that will allow the development of a sustainable technological model is mainly connected with two SDGs. The first, SDG 7 (affordable and clean energy), aims to ensure universal access to affordable, reliable, and modern energy services. Therefore, its purposes include: substantially increasing the share of renewable energy in the global energy mix; improving international cooperation to facilitate access to clean energy research and technology; and widening infrastructure and upgrading technology for supplying modern and sustainable energy services for all nations, especially in developing countries. The second, SDG 9 (industry, innovation, and infrastructure), aims to build resilient infrastructures, to promote inclusive and sustainable industrialization and to encourage innovations. Its included targets aim to: modernize infrastructures and to retrofit industries for strengthening sustainability, with increased resource-use efficiency and greater adoption of clean and environmentally sound technologies and industrial process; intensify scientific research, for improving the technological capabilities of industrial sectors; and encouraging innovation and increasing the number of workers and R&D resources (public and private), particularly in developing countries [1].

The preview targets in two SDGs are interconnected to the purposes of eco-innovations (EIs). The EIs follow practically the same features as general innovations—products, processes, and organizational techniques, which can be new to a market or company—but with the differential of reducing the environmental risks throughout its life cycle [2–4].

Citation: Sperotto, F.Q.; Tartaruga, I.G.P. The Green Side of Industry: The Drivers and the Impacts of ECO-Innovations in Brazil . *Sustainability* **2021**, *13*, 8065. https://doi.org/10.3390/su13148065

Academic Editor: Iwona Zdonek

Received: 23 June 2021 Accepted: 14 July 2021 Published: 19 July 2021

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2021 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/).

39

In discussions about technological changes, there are high expectations regarding the potential of green technologies [5–7] and the competitive advantages of EIs [8]. Investment targeting renewable energies, biotechnologies, and more technically and ecologically efficient products and processes is key to a new technological model [9–11]. In this respect, it is worthwhile to note that the principal economic policies today are closely tied to sustainable development, such as the European Green Deal in the European Union or the Green New Deal in the United States. In line with this paradigm shift in technology, development opportunities also arise for developing countries, specifically, in Latin America [12,13].

At the same time, there are challenges, especially for developing countries [14]. In the case of Brazil, the transition to a green economy and the encouragement of EIs has outlines that are well delineated to that of a developing country, requiring, above all, technological, financial, and institutional efforts. The ideal path presupposes that the country can promote structural change that increases productivity in a socially inclusive [15] and environmentally sustainable manner. However, in fact, this nation has provided a dual response to this situation in terms of government and public policy in innovation [16]. On the one hand, Brazilian innovation systems have some strengths built on the existence of all the components of a significant innovation system, for example, educational, research and regulation subsystems are important for our purposes here, the strategic natural resources, such as land, mineral and water supplies as well as biodiversity from land and maritime biomes. On the other hand, there are key system weaknesses, such as the frequent negative macroeconomic interventions in innovative productive activities and, perhaps the most critical, the absence of a consistent long-term vision or agenda, i.e., fragmentation and discontinuity in public policies and institutional configurations, that give direction and coherence to scientific research and public and private innovation efforts.

In this context, an important question is how developing countries, such as Brazil, have responded to the transition to a more sustainable development model. To illuminate this issue, the analysis combined elements of innovation, production and environmental protection from a sectorial perspective. The study examined specific EI data—drivers and effects—in Brazilian companies, considering two delimitations. The first concerns the identification of the group of eco-innovative companies, through the application of a specific selection criterion, directing the analysis to innovative companies that observed an environmental benefit from their generated innovation. The second delimitation relates to the sectorial profile, namely, the extractive and manufacturing sectors. The database is from the latest edition of Pintec (Pesquisa de Inovação), 2015–2017, which included specific issues about EIs for the first time.

Thus, the objective of this paper was to provide an overview of eco-innovations in Brazilian industry. Therefore, a comparative profile of two groups of companies (ecoinnovators and innovators) was elaborated and, in the case of eco-innovators, the drivers and impacts of eco-innovations were analyzed. We believe that analyzing eco-innovative companies, especially in developing countries, is extremely relevant for challenges of the 2030 Agenda. The panorama of EIs in Brazilian extractive and manufacturing sectors indicated that the eco-innovative companies represent just over a third of the innovators. In general, these companies show a greater effort to innovate when compared to other innovators [17,18]. Associated with this greater commitment, eco-innovators demonstrate a greater perception of the benefits generated by innovation. Likewise, they also exhibit greater sensitivity to its obstacles. The analysis of drivers and impacts allowed us to enrich this profile, highlighting some central elements of environmental innovations, such as the fact that EIs in the Brazilian extractive and manufacturing activities are mainly induced by market factors, and that environmental regulations are not part of the most relevant determinants. Moreover, the most recurrent effects are related to the use of more widespread and less complex technologies, such as the recycling of materials, waste, and wastewater [19].

The article is structured as follows: Section 2 is dedicated to the theme of EI, with an emphasis on drivers and the effects that environmental innovation potentially generates.

Section 3 describes the methodology, in particular, the selection criteria, of the companies and the elaboration of the graphic elements of analysis. Section 4 focuses on results, with special attention to information from Pintec sustainability and the EI set. Section 5 discusses the main findings. Finally, Section 6 concludes by indicating possibilities to guide innovation policy, especially regarding eco-innovations in semi-peripheral countries.

2. Literature Review

For a long time, environmental protection actions were seen as an impediment to the economy, reinforcing the existence of a permanent trade-off between these two dimensions. Although the attachment to this idea is still manifested, technological advances and institutional conduct strategies (government, companies, civil society, NGOs, etc.) already allow us to make viable alternative and sustainable models of production and consumption for which EIs and green technologies are crucial.

In this direction, the proposition of Porter and Line [20] was a precursor in demonstrating that pollution was equivalent to inefficiency. When analyzing some sectors that were strongly influenced by environmental regulations, they found that the adoption of innovations and other factors that promote competitiveness allowed companies to reduce costs with environmental actions. Among other aspects, environmental standards were able to indicate inefficiencies, encouraging new solutions; sensitize companies about their production processes; reduce uncertainty about investments for environmental improvement; and pressure companies to eco-innovate [20].

Technological innovation policies play a primary role for the purposes of sustainable development. Therefore, the substitution of unsustainable production methods and consumption patterns by innovations capable of mitigating impacts on the environment is part of a broader context of technological evolution [21]. Thus, it is especially important that innovation and sustainability are developed together, as the needs foreseen in the meaning of sustainable development are dynamic and, consequently, susceptible to changes in the current context [22].

The concept of EI adopted in this analysis is from the report *Measuring ECO-innovation*: "[...] the production, assimilation or exploitation of a product, production process, service or management or business method that is novel to the organization (developing or adopting it) and which results, throughout its life cycle, in a reduction of environmental risk, pollution and other negative impacts of resources use (including energy use) compared to relevant alternatives." [2] (p. 7).

Thus, a company can become an eco-innovator when it applies environmental technology, e.g., by developing equipment or processes that use inputs in a sustainable and efficient manner and/or reduces the effects of production on the most varied natural ecosystems. The manufacture of "green" products, the shared use of goods and the creation of environmental innovation systems, combining production and consumption to reduce impacts on the environment—such as organic products and renewable energies—are other examples. Another strategy is to adopt managerial or organizational methods, such as environmental management and certification systems, and prevention and cooperation programs, with other entities aimed at reducing the environmental impact [2,4].

A particular characteristic of EIs is the presence of the problem of double externality. As with in innovations in general, there are market failures due to knowledge externalities; in EIs, there are also failures attributed to the environmental benefits generated, which increases the risk, discouraging the development of EIs [23,24]. Therefore, the application of environmental standards is essential to encourage EIs and to reduce the double externality for companies that develop innovations.

As highlighted, our focus in this study was the impacts and drivers of EIs. In relation to the effects, or impacts, of EIs, these are commonly associated with the notion of performance. Studies that analyze impacts emphasize both the results directly perceived in the environment—for instance, the replacement or more efficient use of energy and inputs, and

the reduction in contamination in soil, water, and air—and others of economic character, such as financial performance, competitiveness, and market value [25–28].

In addition, there is a classification that links the impacts of EIs to a kind of innovation. The effects of product EIs refer to improving the recycling of products after use; reducing energy consumption; and reducing emissions in water, soil, and noise. The impact of process EIs corresponds, in addition to the last two products' effects, to the reduction in materials per product unit; the decrease in energy use per product unit; the decrease in CO₂ and other pollutants in the air; recycling waste, water or materials; and replacing hazardous substances [8].

The factors that induce companies to seek solutions to reduce their environmental impact, either reactively or proactively [25,27], are among the most analyzed aspects in the literature [26,29].

To analyze the drivers of EIs, we adopted the theory of environmental innovation [17], which includes both general aspects of innovations and particular aspects of EIs. Just as Horbach [8,17] and Triguero et al. [30], we categorized the determinants into three main groups: supply-side factors, demand-side factors, and institutional and political influences of an environmental characteristic. While the first two groups are widely known in innovation studies, the third—environmental and policy influences—is specific to EIs [31,32].

Hence, based on the factors listed by the theory of environmental innovation [17], it was possible to associate in each group of determinants the drivers present in the special survey by Pintec (Table 1). Pintec listed nine specific factors that contributed to the company's eco-innovation decision.

Determinants	Nature of Factors	Inducers (Pintec)
Supply-side	Technology push Cost-savings	Government support High costs of energy, water, raw materials
Demand-side	Market pull	Voluntary actions Codes of good practice Market demand Reputation
Environmental and policy influences	Regulatory pull/push	Existing environmental standards Future environmental standards Public contract requirements
Comment Advantad former [0 17 (101	

Table 1. Determinants, nature of factors and inducers (Pintec) of environmental innovation.

Source: Adapted from [8,17,30].

The supply side factors are those related to the technological capabilities available for the company to innovate [33]. As with innovations in general, the factors correspond to the company's technological and organizational capabilities [17,34]; the cooperative relations with suppliers, research institutes, agencies and universities [8,30,32,35,36]; the access to external information and knowledge [35,36]; and the access to subsidies and fiscal incentives [31]. Other factors related to the supply-side are energy and material cost-saving strategies [8,34,37].

In turn, demand-side elements highlight market forces as the main determinants of technical change and, therefore, of stimulating (eco)innovations [33]. Among the main factors, the market pull is the possibility of expanding market share [8,17,38]; the increased market demand for green products and eco-label adoption strategies [15,39,40]; and the social awareness actions of the need for clean production [26,38], such as voluntary actions or codes of good practice, mainly resulting from pressure from stakeholders and consumers [22,31,34]. In particular, the institutional pressure can trigger a technological change within the company and therefore redirect R&D efforts for projects related to the environment [41].

A specificity of the determinants related to environmental regulation is the notion of regulatory pull/push. The empirical verification shows that the regulatory structure and environmental policy generates significant effects on EIs. This is due to the fact that these, unlike those in the fields of microelectronics or telecommunications, are not self-applicable; that is, the factors of market pull and technology push are not strong enough to develop them. Therefore, EIs need specific regulatory support, capable of inducing (pull) and boosting them (push) [23]. In this group of drivers, there are environmental standards and regulations, including the expectation of stricter future standards; environmental taxes and duties; and indirect instruments, such as the imposition of requirements for public procurement. It is worth mentioning that the effects of environmental innovation drivers [8,18,20,22,31,37,42].

3. Material and Methods

The research was structured in two stages (Figure 1). The first focused on the selection criteria for eco-innovative companies. A significant part of quantitative studies on innovation uses surveys, such as the Community Innovation Survey (European Union) or Business R&D and Innovation Survey (United States). In particular, the surveys are a relevant base of information about drivers of EIs [8,15,43]. In the case of Brazil, the most well-known base of innovation statistics is Pintec, from the Instituto Brasileiro de Geografia e Estatística (IBGE). Pintec is a sample survey, conducted over a three-year period, and applied to companies with more than 10 employees belonging to the sectors of industry, electricity and gas, and selected services [44]. In addition to contemplating the unprecedented information from the last Pintec about the drivers and impacts of EIs, the objective of this study was to specifically analyze eco-innovative companies. To this end, questions related to the theme of innovation and the environment were identified in the general questions block. The question chosen to differentiate companies between innovators and eco-innovators was N° 105, which asks about the effects of product and process innovations in reducing the impacts on the environment [44,45]. In this question, companies must manifest this effect, indicating one of four levels of importance: high, medium, low, and not relevant. The issue, as proposed in some studies [8,17,18,46,47], is a measure of environmental performance and, therefore, a proxy to differentiate eco-innovative companies from others.



Figure 1. Design of the stages of the study methodology.

Nevertheless, it is pertinent to make two brief considerations about the data. However, the survey also probes about the reduction in the consumption of other inputs (energy, water, and materials); this effect results much more often from cost saving, rather than from an initial guideline of environmental conduct [2,47]. Another point is the bias present in approaches on the environmental impact. The responses of companies may be subject to corporate rhetoric and, consequently, may be overestimated [48].

Based on the definition of the criteria for identifying eco-innovative companies, two databases were requested: one formed by innovative companies that declared "high" or "medium" importance in reducing the impact on the environment, hereinafter identified as ECO, and another composed of companies that answered "low" or "not relevant" relevance, identified as INO. The objective here was to draw a comparative profile between ECO and INO, based on a selection of variables from Pintec's general block, which were considered relevant for the purpose of the investigation. Data were derived from the extractive and manufacturing sectors, for the period 2015–2017.

The second stage involved the development of cluster heatmaps to examine information from the special EI set. The cluster heatmap statistical tool was developed computationally by Galili et al. [49], using the R language [50]. In simple terms, the method is applied to large data matrices, for viewing patterns and creating dendrograms for the definition of clusters.

The heatmap cluster method is based on the application of an algorithm, which provides a structure in a series of clusters, and clusters of clusters, of data examination. In other words, unlike other clustering methods, here the algorithm determines the number of clusters. These algorithms can follow different logics, the most common is hierarchical cluster analysis (HCA). This logic combines objects that are closest to each other until all objects are joined in one single cluster, which will be separated from other furthest objects. To analyze which drivers and impacts were more (or less) important for eco-innovative companies of different sectors, we applied the "complete linkage" method (for HCA) to calculate distances between aggregates (or dissimilarity between clusters). This method determines the distance (or dissimilarity) between two clusters as the distance between the furthest two objects from each cluster. Thus, the ordering method allows similar rows (or columns) to be placed together in the graph. The "complete linkage" method results in compact and homogeneous clusters [51], which we consider adequate for our study.

For the construction of these clusters, two questions from Pintec were selected: N°196, which asked companies to highlight the factors that contributed to their decision to introduce innovations that would generate environmental benefits, and N°105 (sub-items 1 to 5), which requested companies to indicate the relevance of the impact on the environment of the innovations implemented. For these sub-items, only "high" or "medium" graduations were considered.

The heatmap method is performed in three phases [49,51]: treatment of the data matrix; definition of similarity; and determination of the serialization of rows and columns matrix (dendrogram). The first concerns the organization of data. In this phase, two matrices of standardized data in "z" scores were composed by economic activity (two sections and 24 NACE divisions). As the number of companies in the sample is highly varied between activities, it was necessary that the data be standardized. Thus, the matrices were as follows: EI drivers, composed of 10 EI inducers and 26 economic activities, and EI impacts, formed by 5 observed effects and 26 economic activities (Figure 2). In the second, the type of similarity between the observations in each cell (row and column) was defined. In the third and last phases, the ranking of the data in the rows and columns was determined. Furthermore, the ordering of the data (scale) was carried out according to its distribution in the columns. Therefore, the colors and the order of the observations were distributed to highlight the sectors that had the highest values in each variable.

Industrial activities (2 sections + 24 divisions NACE)	Eco-innovation drivers matrix	Eco-innovation impact matrix
 Extractive (mining and quarrying) Manufacturing Man. food products Man. beverages Man. tobacco products Man. tobacco products Man. toxtiles Man. wearing apparel Man. wearing apparel Man. wood and prod. wood and cork Man. wood and prod. wood and cork Man. paper and paper products Printing and reproduction of recorded media Man. coke and refined petroleum products Man. other mon-metallic mineral products Man. ther non-metallic mineral products Man. other non-metallic mineral products Man. fabricated metal products Man. an electrical equipment Man. motor vehicles, trailers and semi-trailers Man. furniture Other manufacturing Repair and install. machinery and equipment 	 Voluntary actions (VACT) Meet the necessary requirements for the consolidation of public contracts (RQPC) Codes of good environmental practices in the sectors which it operates (COGP) Market demand (real or potential) for environmental innovation (MKTD) Availability of government support, subsidies or other government financial incentives for environmental innovation (GOVS) High costs or energy, water or raw materials (HCTS) Improve the company's reputation (REPT) Exvisting environmental standards or taxes on contamination (EEST) Environmental standards or taxes that may be introduced in the future (FEST) Others (OTRS) 	 Recycling of waste, wastewater or materials for sale or reuse (RECY) Reduced contamination of soil, water, noise or air (RCON) Reduction of the company's CO₂ footprint (CO₂ production) (RCO2) Total or partial replacement on energy of fossil fuel with renewable energy sources (RENG) Substitution, in whole or in part, of materials with others that are less pollutants or hazardous (SBSM)

Figure 2. Drivers and impacts of eco-innovation, and industrial activities to generate heatmap matrices.

4. Results

As highlighted in the concept of EI, environmental innovation occurs within the company and follows the characteristics of innovations in general, except for the positive effect it has on the environment. In this sense, according to the guidelines of the Oslo Manual [52] applied at Pintec [44], (eco)innovation corresponds to any product, process, or organizational form that is new to the company, even if it is known in the market. This section has two objectives: to draw a comparative profile between the ECO and INO companies, and to examine the drivers and impacts of EIs on ECOs.

4.1. Comparative Profile of ECO and INO Companies

The last edition of Pintec surveyed 39,329 innovative companies (all economic sectors covered by the survey); out of these, 13,365 (34.0%) said that innovation allowed them to reduce the environmental impact with "high" and "medium" importance. In extractive activity, there were 143 eco-innovative companies, while in manufacturing, there were 12,334 [45].

As shown in Table 2, the rates of innovation—total and sectorial—for ECOs were lower, as expected, since they represent a little more than a third of innovative companies. However, when analyzing the rate of innovative intensity (the ratio between the amount spent on innovation activities and the company's current net revenue), a slightly higher effort is observed in the ECOs, mainly in-house R&D activities [18,42,53]. As EIs represent a very dynamic field in which knowledge is rapidly changing, it is observed that R&D activities, as well as cooperation with research centers and universities, are more important compared to other innovations without environmental benefits [35,37,53–55]. Furthermore, as EIs, particularly products and processes, are characterized by a high degree of innovation, the cost of new equipment, software, and marketing activities is more relevant when compared to other innovations [8,54].

Sectors Innovation Rate Innovative Ir ECO INO ECO INO				Innova	ive Intensity Ra					
	Innovat	ion Kate	То	tal	In-House R&	D Activities	N ² of Companies			
	ECO	INO	ECO	INO	Total					
Extractive	6.2	8.4	1.2	1.9	0.7	0.5	143	193	336	
Manufacturing	12.3	22.0	1.6	1.1	0.6	0.4	12,334	22,062	34,396	
Total ⁽¹⁾	11.4	22.2	1.8	1.6	0.9	0.5	13,365	25,964	39,329	

Table 2. Number, innovation rate, and innovative intensity rates of eco-innovative (ECO) and innovative (INO) companies, by selected economic sector, Brazil, 2015–2017.

Source: Own elaboration, based on IBGE [45]. Note: (1) Includes all sectors of the survey.

The degree of novelty of product and process innovation is an important measure, as it reflects the radical nature of innovation. While national rates refer to innovations for the domestic market already developed in the world market, world rates refer to the most substantial innovations, which are also new abroad (Figure 3).



Figure 3. Rate of product and process innovation in industrial sectors of eco-innovative (ECO) and innovative (INO) companies, according to the degree of novelty for market, Brazil, 2015–2017. Source: Own elaboration, based on IBGE [45].

When comparing these rates in groups of companies, there are interesting elements. In turn, in the world segment, although they present a higher index only in product innovation in the extractive sector, the difference between the two groups is smaller when compared to the national segment, indicating that the ECO and INO are practically at the same level of novelty. On the other hand, in manufacturing, the superiority of the INO in both markets is undeniable. In summary, the 2015–2017 portrait shows that Brazilian EIs generated a higher impact on extractive activities. This is positive, given that this sector is notoriously known for its high polluting potential and its ability to deplete non-renewable resources. Furthermore, incremental technologies predominate in Brazilian EIs, which has been observed in studies of environmental innovation in other countries [8].

The benefits observed with the innovations are relevant aspects for the discussion about the drivers and impacts of EIs, as they highlight a double stimulus effect: the inducing element reflects on good results that corroborate with the induction and encourage it. To address this issue, only companies that expressed "high" or "medium" importance were considered. In addition, these gains were grouped into three categories: competition (related to better market positioning), production (related to greater efficiency in the production process) and protection (linked to compliance with norms and standards). The point that draws attention is the greater perception of ECO about the benefits generated by innovation (Figure 4). The only benefit in which INO had greater relevance was the opening of new markets in the extractive activities.



Figure 4. Benefits highlighted by eco-innovative (ECO) and innovative (INO) companies, in the extractive (**a**) and manufacturing (**b**) sectors, according to category, Brazil, 2015–2017. Source: Own elaboration, based on IBGE [44]. Note: data of companies that attributed "high" or "medium" importance to the benefit.

The main differences in favor of ECO were found in the categories of production and protection. In manufacturing (Figure 4b), the gains from the reduction in the consumption of inputs, energy, water and raw materials were the most relevant, and, in the protection category, both benefits from compliance with norms and standards were accentuated. In extractive sector (Figure 4a), the advantages of competition stood out, particularly the gains from the expansion and maintenance of the market share. In the case of the INOs, it is worth emphasizing that, despite manifesting benefits of reducing the consumption of water, energy, and materials, they declared that the innovation had a low or non-relevant impact on the environment. Here, the hypothesis is that the benefits of the reductions were, above all, the result of a cost saving strategy. Therefore, ECOs expressed the benefits of (such as the decrease in the quantity of inputs), but also others capable of promoting them to more competitive environments. Although it may seem contradictory, eco-innovative companies appear to be less dependent on stricter regulation than more passive companies. Therefore, the adoption of more subtle measures—voluntary actions, eco-audits, eco-labels, and codes of good practice-may be sufficient for pioneering EI companies [23]. However, Els depend more on environmental rules than other innovations [37,54,56].

Finally, other pertinent information concerns the obstacles highlighted by companies. These data are particularly useful to develop, apply, and monitor policies to foster innovation. To assess them, the problems were grouped into three types of barriers: economic (emphasizing the availability and sources of monetary resources), technique (focusing on the direct factors of production and innovation) and institutional (bringing together a variety of actors and forms of interaction) (Figure 5).

The results of barriers to innovation were quite different among industrial activities. In extractive sector, ECOs were shown to be less susceptible to problems, unlike INO (Figure 5a). The exception was in the excessive economic risks item, in the economic barrier. Despite being the most recurrent problem among ECOs, the other two obstacles to the same barrier—high costs of innovation and scarcity of appropriate sources of financing were not perceived with the same intensity. One speculation is that this phenomenon may be associated with the application of measures to support EIs in extractive activities, such as cheaper access to credit or other financing options, which provide, in return, better environmental performance, which is unknown or even devalued by the sector's



INO. In general, financial barriers (imperfect financial markets, absence of financial assets, scalability, and even weak environmental rules) hinder the development and the diffusion of EIs [28].

Figure 5. Obstacles highlighted by eco-innovative (ECO) and innovative (INO) companies, in the extractive (**a**) and manufacturing (**b**) sectors, according to category, Brazil, 2015–2017. Source: Own elaboration, based on IBGE [44]. Note: data of companies that attributed "high" or "medium" importance to the obstacles.

When examining the manufacturing industry, ECOs indicated the greatest sensitivity to barriers to innovate (Figure 5b). The economic barrier was the most accentuated and the high costs of innovation stood out. In the technical barrier, the most prominent problems were the lack of qualified personnel (also shared by the INO) and the lack of information about technology. This last problem, added to the lack of information on markets and the lack of adequate external technical services, expressed the greatest differences between the two groups of companies. Here, the hypothesis was that ECOs constitute a more specific group of companies, which demand more advanced knowledge and technologies, aimed at a green vanguard [35]. Regarding institutional barriers, the intensity was practically the same in ECO and INO, with a greater signal for the scarce possibilities of cooperation with other companies or institutions in the case of eco-innovators. However, as reported in the literature, in general, the EIs depend more on external sources of knowledge and information than other innovations [54,57].

4.2. Analysis of ECO-Innovation Drivers

Although the literature review exposed a broader set of EI drivers, the investigation analyzed the factors defined by Pintec [45] as follows: voluntary actions (VACT); government support (GOVS); codes of good environmental practices (COGP); market demand (MKTD); high costs of energy, water or raw materials (HCTS); existing environmental standards or taxes on contamination (EEST); environmental standards or future taxes (FEST); company reputation (REPT); public contract requirements (RQPC); and others (OTRS) (Table 1 and Figure 2). The data refer to the extractive and manufacturing industries, as well as the 24 sectors of the latter.

Figure 6 (see interactive heatmap in Figure S1) exposes two clusters: the first, indicated by the column dendrogram (horizontal-axis), groups the EI drivers according to their frequency in the industrial sectors, while the second, line dendrogram (vertical-axis), brings together industrial sectors with a similar distribution of drivers. In both cases, the intensity of the orange color indicates the greater influence of the drive, with the scale in number of standard deviations.



Figure 6. Drivers of eco-innovation on eco-innovative companies by sectors of the extractive and manufacturing industries, Brazil, 2015–2017. Source: Own elaboration, based on IBGE [45].

The analysis of the driver cluster shows distinct scales of value among the drivers of EI. The first group, to the left of the heatmap, places the drivers that contributed least in the decision to develop innovations with environmental gains. Both the OTRS (for which no additional information is given) and the GOVSs had close weights; in turn, the RQPCs were mentioned a slightly more. Government support, subsidies, and other incentives for EIs (technology push) seek to boost the development and application of technologies, processes, products, and organizational procedures that are less aggressive to the environment. Regulatory pull requirements are an environmental policy measure that conditions aspects of the company's ecological conduct to the conclusion of business with the government.

The second group, on the right, was divided into two branches. The first branch grouped EEST and FEST (regulatory pull/push), which presented motivations such as the MKTD driver (market pull). These, in turn, were like those of VACT (market pull), which represent a set of company actions, with the purpose of creating internal practices of good environmental conduct (such as the proper disposal of electronic waste). Here, there is a point to be highlighted regarding the importance given to environmental standards, whether existing or to come. Studies have indicated that EIs are more dependent on regulations than other innovations [37,56–59]. However, in the case of Brazil, this was not verified, either by the environmental norms or by the requirements for public contracts, which are both instruments of environmental policy. The second branch reflects the combination of the three main drivers: REPT and COGP (market pull), and HCTS (costsavings). The reputation is related to the adoption of a green communication strategy, such as the creation of an eco-friendly image, which is also very conducive to stimulate marketing innovations. The codes of good practice comprise tacit measures and agreements that, although not mandatory, act as a type of rule of behavior for companies in each sector, or that are part of a production chain. These drivers showed the highest values. The REPT and COGP are drivers influenced by institutional pressures, which help guide R&D efforts for environmental protection strategies [34]. In turn, the HCTS is a very prominent driver in the literature, indicating that EIs apply far more cost-saving or energy-saving strategies than other innovations [23,30,37,54]. Furthermore, if, on the one hand, firms tend to initiate EI projects in response to market pressures, on the other hand, greater investments in EIs are encouraged by other determinants, such as HCTS [38].

The analysis of the cluster of EI drivers, according to the sector (line dendrogram), allows us to raise, albeit with caution, some hypotheses. It is important to note that the sectorial composition involves differentiated activities included within the same economic division; that is, there are different characteristics regarding the intensity of technology/labor, international market orientation [18,60,61], degree of transformation, position in the production chain [39,62], and the size and share capital of the company [18], among other aspects.

The distribution of industrial activities was segmented into two sub-clusters, each formed by two secondary groups, which branched into small groups. In the first subcluster—formed by the first 11 activities arranged in the upper part of the heatmap particularities were identified that may elucidate the weight of certain EI drivers. An example was the first three sectors: tobacco; paper; and motor vehicles, trailers and bodies. In these activities, EIs were motivated mainly by market pull, REPT, and COGP drivers. Similarly, they are exporting sectors, subject to different types of pressures, such as customers, stakeholders and other actors that relate the activity to others downstream in the production chain (for example, tobacco and forest monocultures). The company's position in the production chain can influence the willingness to eco-innovate, as well as its orientation towards the foreign market [17,18,39]. Another divergent sector was the preparation of leathers and the manufacture of leather goods, travel goods, and footwear, with a strong weight of the driver of voluntary actions. One hypothesis is that, although Pintec does not provide data on the type of product, EIs were potentially of an organizational type, given the relevance of internal environmental protection practices. Based on studies focused on sectors with high polluting potential, it is possible to highlight some points. This is the case for the manufacture of chemicals and energy-intensive industries, which often carry out more EIs due to regulatory pressure or high energy consumption [26,37,42]. Two other activities more sensitive to the regulatory pull/push determinants was the manufacturing of machinery and equipment and the manufacturing of coke, petroleum products, and biofuels.

The second sub-cluster connected 13 activities, plus the extractive and processing industries. In the case of extractive activities, there was a prevalence of existing environmental standards (EEST), (regulatory push/pull), reputation (REPT) and codes of good practice (COGP) (market pull), while in manufacturing, the main drivers were reputation and codes of good practice (both market pull), and the high costs of energy, water, and raw materials (HCTS) (cost-saving). The sector that came closest to the average for manufacturing was metal products. In turn, the activities most influenced by the cost-saving driver were furniture, wearing apparel, printing and reproduction of recordings, and electrical equipment. Strategies to reduce costs and use resources more efficiently are highly significant for eco-innovative companies, even more than for other innovators [29]. The codes of good practice mainly guided EIs in the sectors of rubber and plastics, pharmaceutical products and preparations, and textiles. Finally, basic metals were the most induced to eco-innovate by existing environmental rules.

4.3. Analysis of ECO-Innovation Impacts

Like the analysis of drivers, the examination of the impacts was based on the list of effects on companies (Table 1 and Figure 2), as follows: substitution of materials with others that are less contaminating (SBSM); replacement of energy from renewable fossil fuels (RENG); reduction in the contamination of soil, water, waste, or air (RCON); recycling of waste, wastewater, or materials, for sale or reuse (RECY); and reducing the CO₂ footprint (RCO₂). For each effect, the company indicated the degree of company, being that they were computed only as "high" or "medium" manifestations on impact.

Figure 7 (see interactive heatmap in Figure S2) associates two clusters: one, referred to by the column dendrogram (horizontal-axis), which groups the impacts of EIs according to their occurrence in the industrial sectors, while the other, line dendrogram (vertical-axis), approximates industrial activities with a similar arrangement of impacts. In both clusters, the stronger green color corresponds to the greater relevance of the impact, with the scale being the number of standard deviations.



Figure 7. Impacts of eco-innovation on eco-innovative companies, by sectors of the extractive and manufacturing industries, Brazil, 2015–2017. Source: Own elaboration, based on IBGE [44].

The scale of relevance of environmental impacts perceived by eco-innovative companies was segmented into two sub-clusters. The grouping to the right of the heatmap highlights the most observed impacts. The industrial companies realize that their environmental innovations have made it possible, above all, to recycle waste, including wastewater and materials for reuse (RECY). The reduction in the contamination of soil, water, noise, or air (RCON) was the second most relevant. A common element in the two effects was the similar level of importance across sectors. This is the characteristic that links the two effects on the same branch. At another level was the substitution of raw materials for others that are less harmful to the environment (SBSM). The second sub-cluster brought together the two less relevant effects: the reduction in total CO_2 production (carbon footprint), RCO₂, and the replacement of fossil energies with renewable sources (RENG). Although both impacts are important to mitigate damage to the environment, this drew attention to the low level of energy change, by green technologies for energy generation. Furthermore, it is pertinent to mention that EIs that reduce CO_2 emissions are highly dependent on subsidies, basic research, and public technology programs [8].

The examination of the impacts by sector of activity indicated a very distinct cut, formed by a large group with three branches and a group of only one activity—in this case, that of wood products. This isolated position of the sector is due to the high frequency of impact in four of the five predicted. In addition, this activity showed the greatest manifestation regarding energy transformation, using cleaner sources of energy generation (RENG). In general, the other sectors had between two or three more important impacts. Regardless of this provision, it is important to reinforce the reservations regarding the characteristics of each sector, which make it possible to establish only a few deductions. One of them was the lesser relevance of the effect of substituting materials for others that are less harmful to the environment (SBSM) in some sectors, such as wood products; coke and refined petroleum products; food and beverage products; basic metals; and extractive activities. All these activities depend heavily on specific materials for which there are no direct substitutes. This is the case for water for the beverage and food industry; wood for products of this nature; coke, oil, and sugar cane (or other materials used for this purpose, such as soy, corn, and biomass) for fuels; and minerals for extraction activities.

4.4. Drivers, Impacts and Innovation Variables: Correlated Effects

To complete the heatmap analysis, we calculated the correlations among the drivers (10), impacts (5), and other selected innovation variables (91) captured by Pintec (Appendix A). The purpose of this exercise was to establish a profile of eco-innovators, which could guide actions to encourage EI. The results of this exercise are shown in Figure 8.

VARIABLEC	DRIVERS									IMPACTS				
VARIABLES	EEST	FEST	GOVS	MKTD	REPT	VACT	COGP	HCTS	RQPC	SBSM	RENG	RCON	RECY	RCO ₂
Innovative intensity rate	No	No	No	No	No	No	No	No	No	0.521 *	No	No	No	No
Main responsible product innovation - the company	No	-0.750 **	No	No	No	No	No	No	-0.549 *	No	No	No	0.545 *	No
Main responsible product innovation - another company and institute	No	0.601 *	No	No	No	No	No	No	0.652 **	No	No	No	No	No
Main responsible process innovation - the company	No	-0.565 *	No	No	No	No	No	No	No	No	No	No	No	No
Main responsible process innovation - another company and institute	No	-0.527 *	No	No	No	No	No	No	No	No	No	No	No	No
Benefits of innovation - maintenance of market share	No	No	No	No	No	No	No	No	No	No	No	No	No	0.525 *
Benefits of innovation - expansion of market share	No	No	No	No	No	No	No	No	No	0.525 *	No	No	No	No
Benefits of innovation - reduced labor costs	No	No	0.509 *	No	No	No	No	0.681 **	No	No	No	No	No	No
Benefits of innovation - reduction of energy consumption	No	No	0.554 *	No	No	No	No	0.649 **	0.567 *	No	No	No	No	No
Benefits of innovation - expansion of the control of aspects related to health and safety	No	No	No	No	No	No	No	No	No	No	No	0.647 **	No	No
Benefits of innovation - compliance with standard regulations and standards	No	No	No	No	No	No	No	No	No	No	No	No	0.610 *	No
Barriers of innovation - scarcity of adequate external technical services	No	No	No	No	No	No	No	No	0.520 *	No	No	No	No	No
Information sources - another company in the group	No	No	No	No	No	No	0.560 *	No	No	No	No	No	No	No
Information sources - professional training centers and technical assistance	No	No	No	No	No	No	No	0.528 *	No	No	No	No	No	No
ooperation - suppliers	No	No	No	No	No	0.530 *	No	No	No	No	No	No	No	No
Organizational Innovation - management technique	No	No	No	No	No	No	No	No	No	0.510 *	No	No	No	No
Organizational Innovation - environmental management technique	No	No	No	0.528 *	No	No	No	No	No	No	No	No	No	No
Eco-innovators that produced renewable energy	No	No	No	No	No	No	No	No	No	No	0.564 *	No	No	No
EEST	No	0.604 *	No	No	No	No	No	No	No	No	No	No	No	No
FEST	0.604 *	No	No	No	No	No	No	No	No	No	No	No	-0.606 *	No
REPT	No	No	No	No	No	No	0.828 **	No	No	No	No	No	No	No
COGP	No	No	No	No	0.828 **	No	No	No	No	No	No	No	No	No
HCTS	No	No	No	No	No	No	No	No	0.616 *	No	No	No	No	No
RQPC	No	No	No	No	No	No	No	0.616 *	No	No	No	No	No	No
RCON	No	No	No	No	No	No	No	No	No	No	No	No	No	0.519 *
RECY	No	-0.606 *	No	No	No	No	No	No	No	No	No	No	No	0.560 *
RCO ₂	No	No	No	No	No	No	No	No	No	No	No	0.519 *	0.560 *	No

Figure 8. Correlation coefficient—selected variables. Note: No—Not significant; * significant at 1%; ** significant at 0.1%.

Among the aspects observed, one was that eco-innovators motivated by REPT were also influenced by COBP, both drivers of market pull. In EEST and FEST, and in the HCTS (of energy, water, or raw materials) and RQPC, the same happened. In the latter—drivers of a different nature—it is likely that when the company eco-innovates looking for cost-saving, it is also more attentive to other market opportunities, such as meeting environmental conditions to supply products or services to the public sector.

In terms of impacts, the reduction in CO₂ emissions (RCO₂) was linked to two other effects: the RCON (grouping soil, water, waste, or air) and the RECY (contemplating waste, wastewater, or materials). ECO-innovators, who showed a decrease in their carbon footprint, also exhibited a reduction in contamination, in addition to the effects of recycling.

In RECY, there was an inverse correlation with FEST. This result was interesting and corroborates with previous studies. As already noted, recycling is generally associated with the use of more widespread and less complex technologies [37,63]. In this sense, when an eco-innovative company is guided by environmental standards or future taxes, it expects regulations to become more restrictive, and will probably prefer to develop innovations with greater effect in reducing contamination, which recycling practices are more limited in achieving.

The correlations with other innovation variables highlighted pertinent elements. In drivers, eco-innovators induced by FEST normally have another company and institute as the main responsible for product innovation. This relationship reinforces the fact that eco-innovators are more dependent on external agents, especially when there is an expectation of greater rigidity of environmental standards [37,42,57]. In contrast, these standards or taxes were inversely related when the main actor responsible for product innovation is the company. In process EIs, regardless of the main innovation agent, the relationship with

FEST remains the opposite. In other words, companies guided by these standards tend not to eco-innovate in the process.

Government encouragement (GOVS) was associated with reduced labor and energy costs. Moreover, those motivated by market demand (MKTD) tended to adopt environmental management techniques. Recalling that these techniques involve, among other aspects, an environmental communication policy aimed mainly at answering questions from consumers and stakeholders about the company's environmental behavior, this result reinforces the importance of environmental management systems for EIs [34]. In turn, companies that were guided by voluntary efforts (VACT) tended to cooperate with suppliers. A similar observation was made between the COGP and the relevance of another company in the group in the sources of information. In these two situations, there was a greater willingness to establish links downstream of the production chain or with companies in the same sector. It was interesting to note that the high cost driver (HCTS) was related to the benefits of reducing labor and energy costs, and to sources of information with professional training and technical assistance centers. Regarding RQPC, these are linked to the condition that the main executor of the product EI is another company and institute, the benefits of decreased energy consumption, the absence of adequate external technical services, and conversely, when the main executor of product innovation is the company.

In respect of impacts, the replacement of materials (SBSM) was associated with the rate of innovative intensity of the companies, their expansion in the market and the application of management techniques. The substitution of energy (RENG) was related to the production of renewable energy. The RCON tended to increase the control of aspects related to health and safety. The recycling effect was mainly observed when it led to the framing of regulations and standards, and when the main developer of the product innovation was another company and institute. Finally, eco-innovators who saw a reduction in their CO_2 footprint more directly perceived the company's maintenance gains in the market.

5. Discussion

To promote the development of EIs, some aspects must be considered. The first is the fact that they need much more stimulation than other innovations, mainly regarding access to external sources of knowledge and information, the granting of subsidies and the support of public technology programs [8,54]. In the case of Brazilian eco-innovative companies, the scarce possibilities for cooperation with other companies or institutions, the lack of information about technology and the market, the scarcity of external technical services and the high costs of innovation were recurrent impediments in manufacturing.

Although this relevance has not been observed in extractive activities, an addressing investment to promote EIs would be very auspicious. In developing countries such as Brazil, the support of universities and research centers, especially in the environmental and technological areas, can simultaneously help to overcome economic and environmental problems, and can provide opportunities for EIs [37]. As the study indicated, the proximity to other external agents proved to be relevant for eco-innovators motivated by the reduction in high costs. A similar observation was made in the relationship between voluntary actions and cooperation with suppliers.

However, the effectiveness of actions such as this depends greatly on the degree of articulation and commitment between the actors involved. The government has a key role in this process, mainly the public institutions responsible for industrial, innovation and environmental policies [15]. As highlighted in the study, even with the encouragement of some actions, there are important corrections to be made. A good example is the eco-innovators who, even induced by the requirements of public contracts, face the lack of adequate external technical services.

The synergy between the instruments of these policies can increase environmental engagement in the economy, involving companies and consumers, the public and private sectors, universities and research centers, institutional and civil society representatives. The

goals set out in SDG 9 are linked to this environmental commitment. The modernization of infrastructures and the improvement of the industry to make them sustainable, the increase in scientific research and technological capabilities of the industrial sectors, and the encouragement of innovation are goals that are part of the same transformation circuit.

From the perspective of companies, they need to develop skills in the scope of sustainability, improving products, processes, and organizational and management forms. For example, initiatives related to organizational innovations have indicated interesting links with the drivers of market demand and the exchange of materials for others polluting less.

Despite being an internal task that must be undertaken in the long term [22], the benefits achieved by EIs are an incentive both for eco-innovators to advance in new projects and for other companies to realize the advantages of eco-innovating, valuing in its innovation strategy the reduction in environmental impacts. As the study revealed, although EI requires greater effort on the part of eco-innovators—as indicated in the index of in-house R&D activities—they perceived with greater emphasis the gains in the three categories analyzed, competition, production, and protection.

The analysis of drivers and the impacts of EIs provided interesting insights. Examining the drivers, the conclusion was that EIs in Brazil still receive scarce attention from the public sphere. This was evident in the results of government support drivers (technology push), requirements for public contracts (regulatory pull), and environmental standards, existing or expected (regulatory pull/push), which were less relevant than the other inducers. This result is in line with other studies, such as Mazzucato and Penna [16], who pointed out a lack of cohesiveness in the Brazilian innovation system, as we have seen at the beginning of this article, namely through aspects that show the difficulties in creating innovation policy instruments aimed at environmental innovation, in dialogue with industrial and environmental policies. Here, public financial support is essential to encourage EIs. On the other hand, the greater representativeness of the company's reputation drivers and codes of good practice (both market pull), and high cost of materials (cost-saving), signals that the development of EIs is highly associated with a market response. The analysis allowed us to make some hypotheses based on the formation of sub-clusters, such as the connection of certain drivers with the sector's level of exposure in the foreign market, or the position sector in a broader production chain.

The examination of the impacts resulting from EIs was equally interesting, either by complementing the analysis of the inductors, or by indicating the types of EI results most observed in the country. The most relevant effects were the recycling of waste, the reduction in contamination, and the replacement of materials with others that are less harmful to the environment. In contrast, the least verified were the reduction in carbon footprint and the replacement of non-renewable sources of energy with cleaner sources.

Although Pintec does not inform the purpose of the EI, it is desirable to establish some considerations. Among them is the small relevance of adopting clean energy. The transition to a more sustainable model is strongly linked to this substitution, as stated in SDG 7, which aims to increase the share of renewable energy by 2030. Efforts to reduce energy consumption require more advanced technologies and higher R&D expenditures. In Brazil, an interesting initiative is the adoption of a solar energy system for public lighting, an EI implemented by the public sector [63]. For this reason, the development and use of clean energy occurs mainly in central countries, while less technologically demanding options, such as recycling, predominate among semi-peripheral and peripheral countries [19]. This last situation was verified in Brazil. In EIs, the effects associated with more widespread and less complex technologies, such as the recycling of waste, wastewater, or materials, prevailed. This consideration can be extended to the effects of reducing the contamination of soil, water, air and noise, which tend to represent a first stage of EI [39], associated with the use of end-of-pipe, which seeks to improve the quality levels of these elements, through the application of filters or other devices for this purpose. It is important to note that EIs are most often associated with cleaner production measures, unlike end-of-pipe technologies, that reduce environmental damage without changing the production process [64].

Focusing on the sector, a feature of semi-peripheral countries such as Brazil is the high representativeness of sectors of medium or low technological intensity, many of which also have high levels of polluting potential. Despite sharing these characteristics, this does not mean that they are homogeneous; quite the contrary, there is a variety of aspects that differentiate them—market orientation, technological intensity, labor intensity and qualification, and production chain, among others. Hence, the technological options and EI opportunities will differ from sector to sector. Thus, it is presumed that an available part of the stock of clean technologies belongs to a given sector. Therefore, the characteristics of each production activity guides both existing technological opportunities and influences their degree of innovativeness and, likewise, affect the chances that EIs are developed and adopted [39]. This is a key point to be considered in the measures and actions to promote EIs.

Here, it is interesting to note the relationship, or lack thereof, between the sectors with the same profile in drivers and EI impacts. In groups of companies with the same profile of drivers and impacts, it was not possible to detect similarities in technological terms. Particularly in the drivers, it was expected that the sectors of the same group had the condition of related variety [65], i.e., the firms of different sectors had some cognitive proximity and, consequently, intersectoral complementarity that could explain the similar innovative profile. However, the possibility must be raised that this diversity is simply random and, therefore, there is no justification for this. ECO-innovative companies follow an "ecological" technological trajectory for specific reasons, shared in their sector, and which are not present in units from other sectors of the same profile defined by heatmaps.

6. Conclusions

It is increasingly urgent to establish a new model of production and consumption that is sustainable. An alternative to achieve this objective and advance in the development of a sustainable technological model is to encourage EIs. However, semi-peripheral countries such as Brazil will have to deal with their technological limits to access this new green model. Given this scenario, the study set out to establish an overview of EI in the country, with the aim of bringing elements to expand the dialogue among innovation, industry and environment. The study indicated that eco-innovators represent a third of all innovators. As highlighted, these companies have shown greater effort to innovate and greater susceptibility to the benefits and obstacles of innovation; their eco-innovation strategy are mainly driven by market factors; and the impacts are mainly associated with the use of more widespread and less complex technologies.

At the global level, numerous ways to encourage more sustainable production and consumption are discussed. One way is to rethink the economic incentives for EI. An interesting measure is the implementation of industrial certifications and credits that guarantee that the production processes comply with environmental requirements [15].

Semi-peripheral countries such as Brazil can overcome the technological gap by relying on emulation from mission-oriented innovation policies [66] that, for example, can promote an assignment, such as diminishing carbon emissions in a specific period, or a transformative innovation policy [67] that highlights an integrated and systemic approach, considering the negative social and environmental effects of innovation in its framework. Both cases highlight the promotion of investments in innovation (especially in EIs) with the leading role in this process [68].

These types of policy can be carried out through various instruments, which can be on the supply side, such as tax incentives for R&D, or on the demand side, such as public procurement policies, or even by both orientations (supply and demand), such as technical standards [69].

However, the analysis undertaken here may highlight some interesting preferential paths. Returning to the results of EI drivers, among the most valued inductors were reputation and codes of good practice. Both inductors—with a strong correlation as indicated by the study—are tacit and, therefore, not mandatory. Thus, two policy instruments seem

ideal for raising awareness and impelling (or compelling) more companies to innovate in environmental terms. Exercising influence on both the supply and demand sides, these instruments can be used for technological forecasting and regulation.

Technological forecasting aims to sensitize and alert public and private managers about trends in technological trajectories at national and global levels and, at the same time, about the benefits of considering these trends. As previously mentioned, the theme of EIs or sustainable technological development is in the list of global megatrends, highlighting the strong link between technological changes and the environmental dimension [5,7,10,11,70]. It is important to emphasize that this instrument also intends to provoke improvements in the discourse in general, which is fundamental for changing an outdated technological trajectory. On the other hand, the regulation instrument—also correlated—has an important role in the same direction as technological forecasting, but on a compulsory basis. In short, regulations seek to increase the demand for (environmental) innovations and improve the overall technological structure.

In addition to this pair of policy instruments, many others would apply. Among those that should be highlighted are those linked to the promotion of interaction and learning at regional and national scales [69]. Directed to the offer, three instruments are highlighted here: agglomeration policies, support for collaboration, and innovation networks. All of these emphasize the importance of the agents' interdependence to innovate in new and disruptive themes to reach more advanced levels of general technological development. For example, it is worth mentioning the case of eco-industrial clusters, which can be fostered by policies, where the waste from one plant is used as input by another plant, etc. [71], forming clusters of sustainable industries, whose production is associated with the circular economy. The EIs represent a type of transformation process, capable of creating a socioeconomic system based on the idea of the circular economy [72]. In this sense, it is very pertinent to examine whether innovative institutional contexts—regional and national—encourage environmental technological change and, therefore, the EIs [73].

Another interesting perspective is to analyze the phenomenon of the development and diffusion of EIs from the territory [74,75]. Although (eco)innovation requires an analytical knowledge base [76], which results in closer links with universities and the formation of complex international knowledge exchange networks [77], it also requires a synthetic knowledge base [76], which is linked to knowledge and skills based on local, regional and national experience [77].

As a research agenda, we consider that this study can stimulate further analysis. The expectation is that the presented sectorial portrait will serve as an initial stage to investigate groups of sectors that share similarities, such as high polluting potential or belonging to certain production chains. Here, case studies can provide valuable information, particularly to guide eco-innovation policies. Another further research agenda is the analysis of related variety, or systems of related firms in eco-innovation issues from a regional development perspective [65], for example: occupational structures as information networks and also through the flow of labor among industries; the technological relatedness among industries, a fundamental issue for developing countries [78]; or research relationship among individuals or scientific institutions. In all these cases, the relatedness analysis typically looks for the probability that a region establishes an economic activity within the scope of eco-innovation as a function of related activities in that place.

Finally, we believe that EIs have the intrinsic characteristic of avoiding one of the main negative effects of innovations in general, one of the components of the "dark side of innovation" [79], namely, that of environmental degradation. Indeed, from the Industrial Revolution to the present day, many innovations have contributed to the environmental changes that are currently threatening the planet through pollution, loss of biodiversity, climate change, etc. Therefore, in view of a necessary and urgent sustainable global development, EIs are essential in all productive activities.

Supplementary Materials: The following are available online at https://www.mdpi.com/article/ 10.3390/su13148065/s1, Figure S1: Drivers of eco-innovation on eco-innovative companies, by sectors of the extractive and manufacturing industries, Brazil, 2015–2017; and Figure S2: Impacts of ecoinnovation on eco-innovative companies, by sectors of the extractive and manufacturing industries, Brazil, 2015–2017.

Author Contributions: Conceptualization, F.Q.S. and I.G.P.T.; methodology, F.Q.S. and I.G.P.T.; validation, F.Q.S. and I.G.P.T.; formal analysis, F.Q.S. and I.G.P.T.; investigation, F.Q.S. and I.G.P.T.; visualization, F.Q.S. and I.G.P.T.; writing—original draft preparation, F.Q.S. and I.G.P.T.; writing—review and editing, F.Q.S. and I.G.P.T. All authors have read and agreed to the published version of the manuscript.

Funding: This research received support from the Centre of Studies in Geography and Spatial Planning (CEGOT), funded by national funds through the Foundation for Science and Technology (FCT) under the reference UIDB/04084/2020.

Acknowledgments: The authors would like to thank the special tabulations provided by the Pintec/IBGE team, as well as the participants of the study presentation session (in V Meeting of Industrial Economics and Innovation) and the reviewers for their comments and suggestions.

Conflicts of Interest: The authors declare no conflict of interest.

Appendix A

List of variables from the correlation analysis.

- 1. Innovation rate
- 2. Innovative intensity rate
- 3. Innovative intensity rate R&D internal
- 4. Product innovation rate—national market
- 5. Product innovation rate—world market
- 6. Process innovation rate-national market
- 7. Process innovation rate—world market
- 8. Main responsible product innovation—the company
- 9. Main responsible product innovation—another company in the same group
- 10. Main responsible product innovation—in cooperation with another company
- 11. Main responsible product innovation—another company and institute
- Main responsible process innovation—the company
- 13. Main responsible process innovation—another company in the same group
- 14. Main responsible process innovation—in cooperation with another company
- 15. Main responsible process innovation—another company and institute
- 16. Number of workers exclusively dedicated to R&D
- 17. Number of researchers in R&D activities
- 18. Number of graduate workers in R&D activities
- 19. Number of undergraduate workers in R&D activities
- 20. Expenditure on innovative activities—internal R&D activities
- 21. Expenditure on innovative activities-external R&D activities
- 22. Expenditure on innovative activities-external knowledge
- 23. Expenditure on innovative activities-software
- 24. Expenditure on innovative activities-machinery and equipment
- 25. Expenditure on innovative activities—training
- 26. Expenditure on innovative activities—market innovation
- 27. Expenditure on innovative activities—industrial and other projects
- 28. Innovative activities-Internal R&D (high or medium relevance)
- 29. Innovative activities-external acquisition of R&D *
- 30. Innovative activities-acquisition of another knowledge *
- 31. Innovative activities-acquisition of software *
- 32. Innovative activities—acquisition of machinery and equipment *
- 33. Innovative activities-training *

- 34. Innovative activities-introduction of new technical innovations *
- 35. Innovative activities-industrial design and other techniques *
- 36. Benefits of innovation-improving the quality of goods and services *
- 37. Benefits of innovation-expansion of the range of goods and services *
- 38. Benefits of innovation-maintenance of market share *
- 39. Benefits of innovation—expansion of market share *
- 40. Benefits of innovation—opening new markets *
- 41. Benefits of innovation—increase in productive capacity *
- 42. Benefits of innovation-increased production flexibility *
- 43. Benefits of innovation-reduction of production costs *
- 44. Benefits of innovation-reduced labor costs *
- 45. Benefits of innovation-reduction of raw material consumption *
- 46. Benefits of innovation—reduction of energy consumption *
- 47. Benefits of innovation-reduction of water consumption *
- 48. Benefits of innovation—expansion of the control of aspects related to health and safety *
- 49. Benefits of innovation-compliance with standard regulations and standards *
- 50. Barriers of innovation-excessive economic risks *
- 51. Barriers of innovation-high costs of innovation *
- 52. Barriers of innovation-lack of adequate sources of financing *
- 53. Barriers of innovation-organizational rigidity *
- 54. Barriers of innovation-lack of qualified personnel *
- 55. Barriers of innovation-lack of information about technology *
- 56. Barriers of innovation—lack of market information *
- 57. Barriers of innovation-difficulty adapting to standards, rules and regulations *
- 58. Barriers of innovation—scarcity of adequate external technical services *
- 59. Barriers of innovation—centralization of innovation activity in another group company *
- 60. Barriers of innovation—few possibilities for cooperation with other companies/ institutions *
- 61. Barriers of innovation-weak consumer response to new products *
- 62. Information sources-R&D Department *
- 63. Information sources—other areas in the company *
- 64. Information sources—another company in the group *
- 65. Information sources-suppliers *
- 66. Information sources-customers or consumers *
- 67. Information sources-competitors *
- 68. Information sources-independent consultants and consultants *
- 69. Information sources-universities or other higher education centers *
- 70. Information sources-research institutes or technology centers *
- 71. Information sources-professional training centers and technical assistance *
- 72. Information sources—testing, assay and certification institutions *
- 73. Information sources—conferences, meetings and specialized publications *
- 74. Information sources-trade fairs and exhibitions *
- 75. Information sources-computerized information networks *
- 76. Cooperation—customers and consumers *
- 77. Cooperation-suppliers *
- 78. Cooperation-competitors *
- 79. Cooperation—another company in the group *
- 80. Cooperation-consultancy firms *
- 81. Cooperation-universities and research institutes *
- 82. Cooperation-professional training centers and technical assistance *
- 83. Cooperation-testing, testing and certification institutions *
- 84. Organizational Innovation-management technique

- 85. Organizational Innovation-environmental management technique
- 86. Organizational Innovation-work organization
- 87. Organizational Innovation-external relations
- 88. Marketing Innovation-concepts/strategies
- 89. Marketing Innovation-aesthetics, design and others
- 90. ECO-innovators that publish Sustainability Reports
- ECO-innovators that produced renewable energy
 * high or medium relevance.

References

- 1. United Nations. The Sustainable Development Goals. 2018. Available online: https://sdgs.un.org/goals (accessed on 17 December 2019).
- Kemp, R.; Pearson, P. Final Report MEI Project about Measuring ECO-Innovation; UM-Merit: Maastricht, The Netherlands, 2007. Available online: https://www.oecd.org/env/consumption-innovation/43960830.pdf (accessed on 9 October 2016).
- Arundel, A.; Kemp, R. Measuring ECO-Innovation. Working Paper Series; United Nations University, 2009. Available online: http://collections.unu.edu/eserv/unu:324/wp2009-017.pdf (accessed on 16 September 2016).
- 4. Kemp, R. ECO-innovation: Definition, Measurement and Open Research Issues. Econ. Politica 2010, 27, 397–420. [CrossRef]
- Mazzucato, M.; Semieniuk, G.; Watson, J. What Will It Take to Get Us to a Green Revolution? Policy Paper 2015, U. Sussex. Available online: https://www.sussex.ac.uk/webteam/gateway/file.php?name=what-will-it-take-to-get-us-a-green-revolution. pdf&site=264 (accessed on 16 January 2016).
- Perez, C.; Leach, T. A Smart Green 'European Way of Life': The Path for Growth, Jobs and Wellbeing. Beyond the Technological Revolution, Working Paper Series. 2018. Available online: http://beyondthetechrevolution.com/wp-content/uploads/2014/10/ BTTR_WP_2018-1.pdf (accessed on 9 August 2018).
- Lema, R.; Fu, X.; Rabellotti, R. Green windows of opportunity: Latecomer development in the age of transformation toward sustainability. *Ind. Corp. Chang.* 2021, 29, 1193–1209. [CrossRef]
- Horbach, J.; Rammer, C.; Rennings, K. Determinants of eco-innovations by type of environmental impact—The role of regulatory push/pull, technology push and market pull. *Ecol. Econ.* 2012, 78, 112–122. [CrossRef]
- 9. Freeman, C. The greening of technology and models of innovation. Technol. Forecast. Soc. Chang. 1996, 53, 27–39. [CrossRef]
- 10. Mazzucato, M.; Semieniuk, G. Public financing of innovation: New questions. Oxf. Rev. Econ. Policy 2017, 33, 24–48. [CrossRef]
- Perez, C. Unleashing a golden age after the financial collapse: Drawing lessons from history. *Environ. Innov. Soc. Transit.* 2013, 6, 9–23. [CrossRef]
- 12. Perez, C. Cambio tecnológico y oportunidades de desarrollo como blanco móvil. *Rev. CEPAL* 2001, 75, 115–136. Available online: http://www.eclac.org/publicaciones/xml/3/19323/perez.pdf (accessed on 7 April 2013). [CrossRef]
- Perez, C. Innovation Systems and Policy: Not only for the Rich? Working Papers in Technology Governance and Economic Dynamics 2012, n. 42. pp. 1–23. Available online: http://technologygovernance.eu/files/main/2012071005451212.pdf (accessed on 13 April 2013).
- Kemp, R.; Never, B. Green transitions, industrial policy, and economic development. Oxf. Rev. Econ. Pol. 2017, 33, 66–84. [CrossRef]
- Altenburg, T.; Rodrik, D. Green industrial policy: Accelerating structural change towards wealthy green economies. In *Green Industrial Policy. Concept, Policies, Country Experiences*; Altenburg, T., Assamann, C., Eds.; UN Environment, German Development Institute: Bonn, Germany, 2017; pp. 1–20. Available online: https://www.un-page.org/files/public/green_industrial_policy_book_aw_web.pdf (accessed on 27 February 2020).
- Mazzucato, M.; Penna, C. The Brazilian Innovation System: A Mission-Oriented Policy Proposal; Centro de Gestão e Estudos Estratégicos (CGEE): Brasília, Brazil, 2016. Available online: https://www.cgee.org.br/documents/10195/909424/The_Brazilian_ Innovation_System-CGEE-MazzucatoandPenna-FullReport.pdf. (accessed on 8 April 2016).
- 17. Horbach, J. Determinants of environmental innovation—New evidence from German panel data sources. *Res. Policy* 2008, 37, 163–173. [CrossRef]
- Triguero, Á.; Cuerva, M.C.; Álvarez-Aledo, C. Environmental Innovation and Employment: Drivers and Synergies. Sustainability 2017, 9, 2057. [CrossRef]
- Horbach, J. Empirical determinants of eco-innovation in European countries using the community innovation survey. *Environ.* Innov. Soc. Transit. 2016, 19, 1–14. [CrossRef]
- Porter, M.E.; Linde, C. Toward a New Conception of the Environment-Competitiveness Relationship. J. Econ. Perspect. 1995, 9, 97–118. Available online: http://www.jstor.org/stable/2138392 (accessed on 5 February 2013). [CrossRef]
- 21. Freeman, C.; Soete, L. The Economics of Industrial Innovation, 3rd ed.; Routledge: London, UK, 2004.
- Bossle, M.B.; de Barcellos, M.D.; Vieira, L.M.; Sauvée, L. The drivers for adoption of eco-innovation. J. Clean. Prod. 2016, 113, 861–872. [CrossRef]
- Rennings, K. Redefining innovation—ECO-innovation research and the contribution from ecological economics. *Ecol. Econ.* 2000, 32, 319–332. [CrossRef]

- Jaffe, A.B.; Newell, R.G.; Stavins, R.N. A tale of two market failures: Technology and environmental policy. Ecol. Econ. 2005, 54, 164–174. [CrossRef]
- González-Benito, J.; González-Benito, Ó. Environmental proactivity and business performance: An empirical analysis. Omega 2005, 33, 1–15. [CrossRef]
- Díaz-García, C.; Moreno, A.G.; Martínez, F.J.S. Eco-innovation: Insights from a literature review. Innovation 2015, 17, 6–23. [CrossRef]
- Ryszko, A. Proactive environmental strategy, technological eco-innovation and firm performance-case of Poland. Sustainability 2016, 8, 156. [CrossRef]
- Ruiz, J.D.G.; Botero-Botero, S.; Duque-Grisales, E. Financial ECO-Innovation as a Mechanism for Fostering the Development of Sustainable Infrastructure Systems. Sustainability 2018, 10, 4463. [CrossRef]
- Barbieri, N.; Ghisetti, C.; Gilli, M.; Marin, G.; Nicolli, F. A survey of the literature on environmental innovation based on main path analysis. J. Econ. Surv. 2016, 33, 596–623. [CrossRef]
- Triguero, A.; Moreno-Mondéjar, L.; Davia, M.A. Drivers of different types of eco-innovation in European SMEs. Ecol. Econ. 2013, 92, 25–33. [CrossRef]
- Hojnik, J.; Ruzzier, M. What drives eco-innovation? A review of an emerging literature. *Environ. Innov. Soc. Transit.* 2016, 19, 31–41. [CrossRef]
- Del Río, P.; Peñasco, C.; Romero-Jordán, D. Distinctive Features of Environmental Innovators: An Econometric Analysis. Bus. Strat. Environ. 2015, 24, 361–385. [CrossRef]
- Dosi, G. Technical Change and Industrial Transformation: The Theory and an Application to the Semiconductor Industry; Macmillan Press: London, UK, 1984; p. 338.
- Kesidou, E.; Demirel, P. On the drivers of eco-innovations: Empirical evidence from the UK. *Res. Policy* 2012, 41, 862–870. [CrossRef]
- De Marchi, V. Environmental innovation and R&D cooperation: Empirical evidence from Spanish manufacturing firms. *Res. Policy* 2012, 41, 614–623. [CrossRef]
- Pereira, R.M.; MacLennan, M.L.F.; Tiago, E.F. Interorganizational cooperation and eco-innovation: A literature review. Int. J. Innov. Sci. 2020, 12, 477–493. [CrossRef]
- Horbach, J. Do eco-innovations need specific regional characteristics? An econometric analysis for Germany. Rev. Reg. Res. 2014, 34, 23–38. [CrossRef]
- Paraschiv, D.M.; Nemoianu, E.L.; Langa, C.A.; Szabó, T. Eco-innovation, responsible leadership and organizational change for corporate sustainability. *Amf. Econ.* 2012, 32, 404–419. Available online: https://www.amfiteatrueconomic.ro/ArticolEN.aspx? CodArticol=1133 (accessed on 25 January 2018).
- Carrillo-Hermosilla, J.; Del Río, P.; Könnölä, T. Eco-Innovation: When Sustainability and Competitiveness Shake Hands, 1st ed.; Pal-Grave Macmillan: London, UK, 2009.
- Wagner, M. Empirical influence of environmental management on innovation: Evidence from Europe. Ecol. Econ. 2008, 66, 392–402. [CrossRef]
- Berrone, P.; Fosfuri, A.; Gelabert, L.; Gomez-Mejia, L.R. Necessity as the mother of 'green' inventions: Institutional pressures and environmental innovations. *Strat. Manag. J.* 2013, 34, 891–909. [CrossRef]
- Del Río, P.; Peñasco, C.; Romero-Jordán, D. What drives eco-innovators? A critical review of the empirical literature based on econometric methods. J. Clean. Prod. 2016, 112, 2158–2170. [CrossRef]
- Kemp, R.; Pontoglio, S. The innovation effects of environmental policy instruments—A typical case of the blind men and the elephant? Ecol. Econ. 2011, 72, 28–36. [CrossRef]
- 44. Instituto Brasileiro de Geografia e Estatística. Pesquisa de Inovação 2017: Notas Técnicas; IBGE: Rio de Janeiro, Brazil, 2020.
- 45. Instituto Brasileiro de Geografia e Estatística. Pesquisa de Inovação 2017: Tabulação Especial; IBGE: Rio de Janeiro, Brazil, 2020.
- 46. Conceicao, P.; Heitor, M.V.; Vieira, P.S. Are environmental concerns drivers of innovation? Interpreting Portuguese innovation data to foster environmental foresight. *Technol. Forecast. Soc. Chang.* 2006, 73, 266–276. [CrossRef]
- Queiroz, J.M.; Podcameni, M.G.V.B. Estratégia inovativa das firmas brasileiras: Convergência ou divergência com as questões ambientais? *Rev. Bras. Inovação* 2013, 13, 187–224. [CrossRef]
- Chapman, K. Environmental policy and industrial location in the United States. In Institutions and Geographical Patterns; Flowerdew, R., Ed.; Croom Helm: London, UK, 1982; pp. 141–168.
- Galili, T.; O'Callaghan, A.; Sidi, J.; Sievert, C. heatmaply: An R package for creating interactive cluster heatmaps for online publishing. *Bioinformatics* 2018, 34, 1600–1602. [CrossRef] [PubMed]
- Rstudioteam. RStudio: Integrated Development for R [Computer Software]; RStudio: Boston, MA, USA, 2020. Available online: https://rstudio.com/ (accessed on 17 July 2020).
- Storme, T.; Derudder, B.; Dörry, S. Introducing cluster heatmaps to explore city/firm interactions in world cities. *Comput. Environ.* Urban. Syst. 2019, 76, 57–68. [CrossRef]
- 52. Organization for Economic Co-Operation and Development (OECD). Oslo Manual: Guidelines for Collecting and Interpreting Innovation Data, 3rd ed.; OECD Publishing: Paris, France, 2005; p. 166. [CrossRef]
- Leitão, J.; De Brito, S.; Cubico, S. Eco-Innovation Influencers: Unveiling the Role of Lean Management Principles Adoption. Sustainability 2019, 11, 2225. [CrossRef]

- Horbach, J.; Oltra, V.; Belin, J. Determinants and Specificities of ECO-Innovations Compared to Other Innovations—An Econometric Analysis for the French and German Industry Based on the Community Innovation Survey. Ind. Innov. 2013, 20, 523–543. [CrossRef]
- Hájek, P.; Stejskal, J. R&D Cooperation and Knowledge Spillover Effects for Sustainable Business Innovation in the Chemical Industry. Sustainability 2018, 10, 1064. [CrossRef]
- Carrillo-Hermosilla, J.; del Río, P.; Könnölä, T. Diversity of eco-innovations: Reflections from selected case studies. J. Clean. Prod. 2010, 18, 1073–1083. [CrossRef]
- Leitão, J.; Pereira, D.; De Brito, S. Inbound and Outbound Practices of Open Innovation and ECO-Innovation: Contrasting Bioeconomy and Non-Bioeconomy Firms. J. Open Innov. Technol. Mark. Complex. 2020, 6, 145. [CrossRef]
- Oltra, V. Environmental Innovation and Industrial Dynamics: The Contributions of Evolutionary Economics. W. P. GREThA. 2008. Available online: https://core.ac.uk/download/pdf/6297007.pdf (accessed on 27 October 2017).
- Ghisetti, C.; Pontoni, F. Investigating policy and R&D effects on environmental innovation: A meta-analysis. Ecol. Econ. 2015, 118, 57–66. [CrossRef]
- 60. Horbach, J.; Jacob, J. The relevance of personal characteristics and gender diversity for (eco-)innovation activities at the firm-level: Results from a linked employer-employee database in Germany. *Bus. Strat. Environ.* **2018**, *27*, 924–934. [CrossRef]
- Choi, H.; Yi, D. Environmental innovation inertia: Analyzing the business circumstances for environmental process and product innovations. Bus. Strat. Environ. 2018, 27, 1623–1634. [CrossRef]
- Galera-Quiles, M.; Piedra-Muñoz, L.; Galdeano-Gómez, E.; Carreño-Ortega, A. A Review of ECO-Innovations and Exports Interrelationship, with Special Reference to International Agrifood Supply Chains. *Sustainability* 2021, 13, 1378. [CrossRef]
- Koeller, P.; Miranda, P.; Lustosa, M.; Podcameni, M. *Ecoinovação: Revisitando o Conceito*; Texto para Discussão 2556; IPEA: Rio de Janeiro, Brasil, 2020; p. 60. Available online: http://repositorio.ipea.gov.br/bitstream/11058/9960/1/td_2556.pdf (accessed on 22 October 2020).
- Frondel, M.; Horbach, J.; Rennings, K. End-of-pipe or cleaner production? An empirical comparison of environmental innovation decisions across OECD countries. *Bus. Strat. Environ.* 2007, 16, 571–584. [CrossRef]
- Asheim, B.T.; Boschma, R.; Cooke, P. Constructing Regional Advantage: Platform Policies Based on Related Variety and Differentiated Knowledge Bases. *Reg. Stud.* 2011, 45, 893–904. [CrossRef]
- Mazzucato, M. Mission-Oriented Innovation Policy: Challenges and Opportunities; Working Paper; UCL—Inst. for Innov. and Public Purpose: London, UK, 2017. Available online: https://www.ucl.ac.uk/bartlett/public-purpose/sites/public-purpose/files/ moip-challenges-and-opportunities-working-paper-2017-1.pdf (accessed on 27 April 2019).
- Schot, J.; Steinmueller, W.E. Three frames for innovation policy: R&D, systems of innovation and transformative change. *Res. Policy* 2018, 47, 1554–1567. [CrossRef]
- Zagato, L. Ainda é possível que os países em desenvolvimento façam seu catching up no século XXI? Braz. J. Political Econ. 2019, 39, 527–543. [CrossRef]
- 69. Edler, J.; Fagerberg, J. Innovation policy: What, why, and how. Oxf. Rev. Econ. Policy 2017, 33, 2–23. [CrossRef]
- Schot, J.; Kanger, L. Deep transitions: Emergence, acceleration, stabilization and directionality. *Res. Policy* 2018, 47, 1045–1059. [CrossRef]
- 71. Aoyama, Y.; Murphy, J.T.; Hanson, S. Key Concepts in Economic Geography; SAGE Publications: London, UK, 2011.
- De Jesus, A.; Mendonça, S. Lost in Transition? Drivers and Barriers in the ECO-innovation Road to the Circular Economy. *Ecol. Econ.* 2018, 145, 75–89. [CrossRef]
- González, P.D.R. The empirical analysis of the determinants for environmental technological change: A research agenda. Ecol. Econ. 2009, 68, 861–878. [CrossRef]
- Tartaruga, I.; Sperotto, F. Rethinking Clusters in the Sense of Innovation, Inclusion, and Green Growth. In *Rethinking Clusters*; Sedita, S., Blasi, S., Eds.; Sustainable Development Goals Series; Springer: Cham, Switzerland, 2021; pp. 101–110. [CrossRef]
- Podcameni, M.G.; Cassiolato, J.E.; Lustosa, M.C.; Marcellino, I.; Rocha, P. Exploring the convergence between sustainability and local innovation systems from a southern perspective: What Brazilian empirical evidence has to offer. *Local Econ. J. Local Econ. Policy Unit* 2019, 34, 825–837. [CrossRef]
- 76. Asheim, B. Differentiated knowledge bases and varieties of regional innovation systems. Innovation 2007, 20, 223–241. [CrossRef]
- Binz, C.; Truffer, B. Global Innovation Systems—A conceptual framework for innovation dynamics in transnational contexts. *Res. Policy* 2017, 46, 1284–1298. [CrossRef]
- Hidalgo, C.; Balland, P.A.; Boschma, R.; Delgado, M.; Feldman, M.; Frenken, K.; Glaeser, E.; He, C.; Kogler, D.; Morrison, A.; et al. The Principle of Relatedness. In Proceedings of the Unifying Themes in Complex. Systems IX, ICCS 2018, Springer Proceedings in Complexity, Cambridge, MA, USA, 22–27 July 2018; Morales, A., Braha, D., Minai, A., Bar-Yam, Y., Eds.; Springer: Cham, Switzerland, 2018. [CrossRef]
- 79. Coad, A.; Nightingale, P.; Stilgoe, J.; Vezzani, A. Editorial: The dark side of innovation. Ind. Innov. 2021, 28, 102–112. [CrossRef]




Article The Impact of Succession Management on Small and Medium Enterprises' Sustainability in Lagos State, Nigeria

Francis Ezieshi Monyei ¹, Wilfred I. Ukpere ^{2,*}, Emmanuel Kalu Agbaeze ¹, Solomon Omonona ³, Lovlyn Ekeowa Kelvin-Iloafu ¹ and Happiness Ozioma Obi-Anike ¹

- ¹ Department of Management, Faculty of Business Administration, Enugu Campus, University of Nigeria, Nsukka 410001, Nigeria; monyei_francis@yahoo.com (F.E.M.); agbaeze.emmanuel@unn.edu.ng (E.K.A.); lovlyn.kelvin-iloafu@unn.edu.ng (L.E.K.-I.); Ozioma.obi-anike@unn.edu.ng (H.O.O.-A.)
- ² Department of Industrial Psychology and People Management, School of Management,
- College of Business & Economics, University of Johannesburg, Johannesburg 2092, South Africa
- ³ Business Management Department, University of Fort Hare, Alice 5700, South Africa; Somonona@ufh.ac.za
- Correspondence: wiukpere@uj.ac.za

Abstract: This study investigated the impact of succession management on small and medium enterprises' (SMEs') sustainability in Lagos, Nigeria. The study adopted a descriptive survey research method and its participants were selected from five SMEs across Lagos State, based on their number of years of operating, staff strength, and branch locations. The enterprises included Gnakk Enterprises, Dreamville School, Vitus Cosmetics, Osi Ventures, and Chijioke Frozen Foods. The Pearson product moment correlation coefficient was used to analyze the hypotheses, which revealed that succession management had a positive correlation (r = 0.934, *p*-value < 0.05) with corporate sustainability amongst SMEs in Lagos State, Nigeria. Hence, the study concluded that in respect of strategic recruitment, succession management attracts and assists to retain a high performing skilled labour force.

Keywords: capability; leadership; organization; survival; workforce

1. Introduction and Theoretical Background

The present economic climate regards businesses as the most important driver of national economic development and growth. Organizations compete to gain and maintain a competitive edge over rivals. To preserve an optimum level of corporate performance, a variety of resources are employed, including finances, workforce, and machinery [1,2]. However, amongst these resources, human resources (workforce) is the most critical in any organization [3,4]. Currently, many organizations, especially small and medium enterprises (SMEs), expend significant energy trying to attract customers, while they spend much less time on the workforce to retain these customers. Almost all organizations work hard to attain corporate sustainability [5,6]. However, there seems to be a significant gap between what they wish to attain and what they eventually receive [7]. Finding a long-term solution to the situation and sustaining a competent workforce to deliver corporate sustainability, remains a challenging aspect and process for firms, as it requires effective strategic and succession management. Dauda [8] and Mohrman and Lawler [9] claim that succession management enables SMEs to identify a high potential workforce and talented prospects that possess leadership capabilities and business acumen, which are required to execute long term strategic business plans, while the way in which business executives respond to succession could influence corporate sustainability [10,11].

Bansal and Song [12] argue that the term corporate sustainability is both current and provides a foundation for empiricism. The authors point out that corporate sustainability's early emphasis was on society and social issues, which were portrayed as obligations to interested parties at that time. Contrarily, they mention that early corporate sustainability

Citation: Monyei, F.E.; Ukpere, W.I.; Agbaeze, E.K.; Omonona, S.; Kelvin-Iloafu, L.E.; Obi-Anike, H.O. The Impact of Succession Management on Small and Medium Enterprises' Sustainability in Lagos State, Nigeria. Sustainability 2021, 13, 13489. https://doi.org/10.3390/ su132313489

Academic Editors: Iwona Zdonek, Adam R. Szromek and Marc A. Rosen

Received: 12 October 2021 Accepted: 13 November 2021 Published: 6 December 2021

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2021 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/).

63

focused on environmental management, which is distinct from the environmental protection debate that currently surrounds economic development, now known as sustainable development [13,14]. Furthermore, they emphasize that the corporation is nested in social and environmental systems while discussing corporate sustainability. From a nomothetic standpoint, Bansal and Song [12] also argue that the terms corporate and sustainability have different antecedents and results, rendering their current homogeneity a challenge. They also state that the introduction of strategic approaches to the problem of corporate sustainability and the environment resulted in their confluence. It generates strategic options that are more easily aligned with the dominant neo-classical conception of SMEs as a profit-driven business.

In Nigeria, particularly in Lagos State, SMEs are currently experiencing prolonged volatility as a result of a high incidence of staff turnover, nepotism, corrupt practices, and inconsistent succession management [3,15]. These have become worrisome, especially owing to the nation's current economic and financial recession. As ineffective, inconsistent, and skewed succession planning continues to infect and saturate every part of their business, the impacts of this on SMEs remain unresolved and unacceptable. Hence, it has had a significant impact on their overall performance, as well as on the morbidity of several. Collings and Mellahi [16] claim that succession management has thus far been premised on the idea of a declining workforce base, whilst owing to the present economic situation, SMEs now understand that to gain and maintain a competitive edge, they must manage their workforce effectively. To do this, they must confront the realities of global succession management (GSM) and its several challenges, while developing human capital management activities to meet those challenges [17,18]. Hence, this enquiry, which investigates the impact of succession management on SMEs' sustainability in Lagos State, Nigeria, is justified.

1.1. The Resource-Based View (RBV)

Although others have contributed to development of the resource-based view of strategy, Wernerfelt is credited with coining the term as we know it today [19–21]. The said author opines that a corporation is a collection of resources that is tethered to it semipermanently [22]. A firm's RBV comprises the resources that it has created to make it viable within the sector, whilst emphasizing its internal environment as a driver of competitive edge. Since the 1980s, research focus has switched from industry structure, including the five forces model and the structure–conduct–performance (SCP) paradigm to the firm's internal structure, comprising both resources and experiences (as the key components of the RBV). The RBV also uncovered that an enterprise's resources, which it owns, deploys, and uses, are significantly more important than industry structure [21,23].

Only the most strategic and valuable resources and competencies should be considered as competitive advantage sources, according to scholars who subscribe to the RBV [21,24–26], and who have used terms such as core competencies. Competitors cannot reproduce, substitute, or recreate core competencies, which are unique, scarce, and highly valued firm-based resources. Huang [26] posits that there is a distinction between tangible and intangible resources, with intangible resources being the most essential in terms of strategy. They believe that intangible assets are more likely to be used to create a long-term competitive edge than material resources. The RBV combines the two variables (succession planning and corporate sustainability) by offering insight into a firm's resources, competencies, and strategies, as well as how these might be utilized for a better, more sustainable and long-lasting business.

1.2. Research Question

The above background triggered the research question presented below.

1. To what extent does succession management impact SMEs' sustainability in Lagos State, Nigeria?

1.3. Research Objective

The objective of the study is stated below.

 To investigate the impact of succession management on SMEs' sustainability in Lagos State, Nigeria.

2. Literature Review

2.1. Succession Management

The concept of succession management refers to the process that business owners implore to attract, evaluate, and strengthen their workforce to guarantee that they are capable of playing important roles in the firm [27,28]. It is the process of finding and evaluating potential successors, whilst preparing them for present and future employment prospects in the company. In addition, succession planning necessitates identifying successors and engaging key personnel in the firm to guarantee their involvement and commitment on a long-term basis [8]. Succession planning is a dynamic, continuing process of identifying, evaluating, and developing leadership candidates for the future. It is also used to evaluate, develop, and recognize key contributors who can satisfy future strategic and operational requirements.

Ahsan [28] asserts that succession management is a way of determining vital staff potential and having a backup strategy for their smooth replacements. There is a constant suspicion or anxiety in the minds of SMEs' management concerning a shortage of future successors for replacements at all administrative levels [5,15]. This creates a lacuna that mitigates continuity of business operations, viability and sustainability, which introduces the need for an established succession management system and structure in the enterprise. Preparedness for an unexpected shortage in an enterprise's workforce base and corporate sustainability signals the importance of succession management, which is a creative strategy and means to build competence, career advancement, and staff promotion [7,8]. As a strategy, succession management should be majorly considered in these aspects, including equipping the workforce with pre-requisite knowledge and capability for the new task. The successor's effective training and development demands follow and, finally, management accepting the successor. Rothwell [29] asserts that if succession management is optimally and effectively executed, it will enable the corporation's sustainability and will ensure fulfilment of the workforce's aspirations. The author maintains that steps or efforts are engineered for succession management to have a positively significant impact on managerial operations, if well integrated [30,31].

2.2. Strategy for Succession Management

A thorough succession management strategy includes several actions that serve as a road map to create succession plans. Ioannou and Serafeim [30] and Dauda [8] aver that there is no such thing as a one-size-fits-all solution for succession planning; hence, different organizations will use diverse components of any succession planning strategy. To begin, critical jobs must be assessed in terms of the skills and experience that leadership roles require. Then, based on management's assessments of their performance and potential for advancement, they should identify notable talent at the company's top two or three levels. Secondly, through formal education, leadership and management training, coaching or mentoring, and job improvement, management should analyze key talents, the primary development needs of skills suited for the next level, and strategies for future successors. Finally, an annual assessment should be used to track the employee development plan's implementation, as well as adoption of a transition strategy to select leaders from the talent pool to fill vacancies. Internal capabilities, talent pool planning, and future succession and replacement training are the three types of succession planning that are commonly used in this respect [27,29].

2.3. Models of Succession Management

A succession management model is a framework, structure or roadmap that depicts the integral aspects of a viable, unbiased, and successful identification, development, and retention of high performers [28,31]. Corporations typically use one of three succession management models, as shown below:

- 1. Emergency substitutions or short-term planning;
- 2. Long-term talent management or planning; and
- 3. A mix of the above-mentioned strategies.

In the first model emergency or short-term planning substitutes are the most typical succession management, while they are also a crucial point for all kinds of firms. This strategy focuses on an immediate need, such as the need to replace personnel who leave the company. Second, the long-term personnel management or planning model is built on a strategic framework for a company's future goals, key occupations needed for development, and the ideal people to perform those jobs. Some companies seek the participation of all employees in the required evaluation process, which helps them to find the best candidate to train for the company's future needs [27,29]. If a firm intends to nurture its leadership from within its existing talent pool, succession planning becomes a critical component of its long-term human capital strategy. Specialized personnel or a talent pool, identifying and expanding future talents that are vital to the enterprise's performance, and engaging and retaining people through engagement in an internal capacity, are all advantages of this method. Finally, the mixed model combines the first and second plans by allowing senior management to plan for the company's long-term growth and its personnel, while also preparing to guarantee that their corporate activities are not disrupted by emergency replacements or are not hampered by the loss and scarcity of qualified individuals [8,29]. The benefits of succession management are discussed next.

2.4. Benefits of Succession Management

Regardless of the ownership structure, succession management is crucial for a company's long-term viability and competitive advantage. Hence, succession management aids a company's deliberate approach to develop leaders and staff competence evaluation. It is extremely important when a government regulation forces mass retirements to keep a company's critical knowledge base intact [8]. According to Huang [26], succession management allows employee movement within the company. This has the potential to promote employee confidence. Rothwell [29] notes that succession planning is a talent management technique that is essential to attract, maintain, and advance specific capabilities that help a company remain competitive. Internal resourcing, reducing personnel turnover caused by high-fliers who change jobs frequently, and preparing suitable candidates for top management roles, are all seen to benefit from succession management [26].

Dauda [8] reiterates that consistent succession management provides broader insight into not only focusing on switching top officials, but also forces firms to examine all staff for the enterprise's future. The main benefit of succession planning is the proactive construction of a strong future staff resource base, which is critical to attract and retain the best and most important people who will contribute to the company's current and future growth. Hence, suitable employees with pre-requisite expertise in specific jobs doing specific duties, is critical for corporate sustainability [8,26,32].

2.5. Corporate Sustainability

The ability of a company to persist is referred to as corporate sustainability [20,33,34]. Sustainability is defined as long-term company success that also assists with advancing the economy and society in terms of promoting a stable society and a healthy environment. It concerns contributing constructively to society, while also safeguarding the company's operations and the communities in which it operates [3,35]. The three pillars of business sustainability that have been identified are environmental, socio-cultural, and economic sustainability. The use of resources to meet human, economic, social, and environmental

demands in a way that respects the interests of current and future generations, is known as sustainable development [26,36], which is discussed next.

In essence, corporate sustainability is the belief that businesses and their directors have a responsibility that extends beyond profit [33]. Dyllick and Hockert's [36] widely accepted definition of business sustainability is helpful to understand its current usage. The authors describe business sustainability as a company's ability to meet the needs of its immediate stakeholders without jeopardizing its potential to meet the needs of future stakeholders. This explanation's scope, which is clearly borrowed from sustainable development, is restricted to the organization. The company's stakeholders, or those who have direct control over its operations, finances, or market, are a source of concern. There is no mention of ethics in the definition. Although others, such as Hahn and Figge [33], aver that there appears to be some implicit pragmatic consensus that corporate sustainability refers to a composite and multi-faceted construct that includes environmental, social, and economic organizational outcomes, there is no explicit mention of the environment. It is also worth noting that the sustainability's intergenerational aspect is confined to the next set of stakeholders. Hence, within a finance context, it has a fundamental and organizationally focused idea [32].

Amini and Bienstock [34] attempted to synthesize and operationalize the debate recently. They validate the term corporate sustainability as being organizationally particular in terms of scope and its principal purpose as being ecological in their well-developed work, which recognizes the complexity and far-reaching features of sustainability. While Vermeulen and Witjes [37] include innovation as a new dimension, the policy goal is corporate sustainability's strategic profit potential. Finally, they propose a slightly different interpretation of the term, claiming that it refers to particular organizations implementing plans to attain long-term sustainability [37,38].

2.6. Sustainable Development Practices

The commerce industry in Lagos State, Nigeria, was selected for this study owing to reports of their involvement in sustainable development, as published in the National Bureau of Statistics [37,39]. The data reviews reported by the National Bureau of Statistics show the following common sustainability practices amongst SMEs.

Waste reduction and recycling: Staff are encouraged to use online approvals, electronic communications, and other web-based programs rather than printing documents, except when necessary [20,40]. For certain types of transactions, online business transactions have replaced cash and cheque books in SMEs, which has resulted in significant paper savings and the transmission of electronic transactional documents.

Diversity: The organizations attract high-flying employees by promoting a diverse workforce and by treating all employees equally and without discrimination on the basis of gender, ethnicity, color, nationality, religious beliefs, or any other distinguishing trait, whilst ensuring that under-represented groups continue to have access to open positions within the organization.

Community investment projects: The SMEs invest in the communities in which they operate by providing basic social services, such as clinic rehabilitation, school renovation, reequipment of craft workshops, book donations for school libraries, and science laboratory construction, as well as by increasing long-term employment through a community service effort inside the company [12,39,41].

3. Empirical Review

Akinyele et al. [27] considered the impact of succession planning on business persistence. According to empirical research from Covenant University, the study's purpose was to investigate the impact of succession planning on organizational persistence. A survey and an individual interview were utilized to gather the required data for the investigation. Pearson's correlation coefficient was used to test the hypotheses. The findings showed that succession planning significantly affects organizational survival. This reveals techniques for sharing institutional information and retaining institutional data and, accordingly, guaranteeing the firm's presence. The study likewise uncovers that career advancement largely affects Covenant University's organizational longevity. Thus, proficiency amongst staff is a requirement for them to advance towards addressing the university's succession requirements, which then guarantees the institution's long-termed survival.

Maguta [42] examined the impact of succession planning techniques on the daily operations of non-governmental organizations (NGOs) in Kenya, investigating the link between succession planning approaches and attrition rates, organizational friction, and shareholder expectations. A descriptive strategy geared towards Nairobi-based registered NGOs was ideal for the study. The study gathered both qualitative and quantitative data from 15 non-governmental groups, and presented its findings, using illustrative frequency distributions. Finally, the study found that succession planning is a continuous worry for NGOs seeking to maintain their performance. In Nairobi, there was a gap between the traditional management of non-governmental organizations and implementation of succession planning approaches. Maintaining relevance, as well as strengthening other successful managerial talents, also proved to be a key contributor towards NGOs' long-term success.

Syeda and Abida [43] studied the succession planning impact on employee commitment in the telecommunications industry. For scholars in the field, the study provided an overview of the industry's succession planning impact on employee commitment, focusing on the role of talent management in the contemporary global economy. Only 250 copies of the questionnaire were returned after being personally distributed at the head offices and franchises of five different telecommunication companies in Rawalpindi, Islamabad, for the purpose proposed. The statistics were examined, using SPSS (Statistical Procedures for Social Sciences). The study's findings revealed that employee commitment and effective succession planning have a significant relationship. This result centres on successful succession planning assisting with developing successful leadership and their role as the most crucial enabler of both workplace and employee engagement, as well as work teams, particularly in the telecommunications industry, at large.

Hanaysha [44] considered staff productivity to be one of the key management themes that have received significant scholarly attention and is thought to be a vital mechanism for advancing business sustainability. To ensure long-term success, it was also necessary to have a fundamental awareness of the basic components that drove sustainability. To achieve the purpose, primary data was collected, using an online survey, where a sample of 242 staff from public universities in Northern Malaysia were polled. Structural equation modeling was used to analyze the data, which showed that work engagement has an affirmative impact on staff productivity. Furthermore, the research shows that all aspects of work engagement such as enthusiasm, dedication, and engagement, have a significant impact on employee outcomes.

3.1. Research Hypothesis

From the above literature review and theoretical foundation, the following null and alternate hypotheses were formulated:

Hypotheses 1: Succession management does not relate positively to corporate sustainability.

Hypotheses 2: Succession management relates positively to corporate sustainability.

3.2. Knowledge Gaps

The literature shows a lack, dearth, and paucity of local material and studies on how small and medium businesses in Lagos State, Nigeria, can achieve a competitive edge by implementing consistent succession management that propels their sustainability and expansion of operations, both nationally and globally.

4. Research Methodology

The research adopted a descriptive survey method. The total number of SMEs in Lagos has been quoted as 3,224,324 [15]; however, only those who fulfilled the selection criteria were selected for this study. The selection criteria were based on their number of years in operation, capital base, and branches across Lagos State. Five SMEs were selected randomly, and these included Gnakk Enterprises, Dreamville School, Vitus Cosmetics, Osi Ventures, and Chijioke Frozen Foods. The research participants were staff at the selected SMEs, comprising a total of 3126 staff from the five (5) SMEs. A sample size of 359 was received, using the Bill Godden formula. Based on the population being finite, a proportionate stratified sampling approach, using Bowley's proportional formula, was used to choose respondents in each of the selected firms under investigation. The questionnaire was designed, using the 5-point Likert scale and was administered manually to respondents. Content validity of the instrument was conducted by three management experts from both the industry and academia. The instrument's reliability was tested using Cronbach alpha, which yielded a coefficient of 0.83, suggesting strong instrument consistency, and at a 5% probability threshold of significance. Lastly, the study used the Pearson's product-moment correlation to analyze the data.

5. Data Analysis and Results

5.1. Results

Table 1 indicates that a total of 359 questionnaires were distributed: 54 were distributed to Gnakk Enterprises (7 administrative staff and 47 low-cadre staff) and 46 were returned; 125 questionnaires were distributed to Dreamville School (10 to administrative staff and 115 to low-cadre staff) and 100 were returned; 103 were distributed to Vitus Cosmetics (6 to administrative staff and 97 to low-cadre staff) and 93 were returned; 48 were distributed to Osi Ventures (7 to administrative staff and 41 to low-cadre staff) and 29 were returned; and 54 were distributed to Chijioke Frozen Foods (5 to administrative staff and 24 to low-cadre staff) and 23 were returned.

Omerications	Distribut	ed Number	Returne	d Number	Number Not Returned		
Organizations	Administrative Staff	Low-Cadre Staff	Administrative Staff	Low-Cadre Staff	Administrative Staff	Low-Cadre Staff	
Gnakk Enterprises	7	47	4	42	3	5	
Dreamville School	10	115	8	92	2	23	
Vitus Cosmetics	6	97	5	88	1	9	
Osi Ventures	7	41	6	39	1	2	
Chijioke Frozen Foods	5	24	3	20	2	4	
Total	35	324	26	281	9	43	
Grand Total	3	359	3	307		52	

Table 1. Questionnaire distribution and return.

Source: Field Survey, 2021.

5.2. Descriptive Statistics Related to the Research Objective

Table 2 indicates that 46 of the respondents (15%) strongly agreed, 91 (30%) agreed, 32 (10%) were undecided, 97 (32%) strongly disagreed, and 41 (13%) strongly disagreed with the assertion that attracting and retaining high-performing employees is crucial for an organization's success.

Table 3 shows that 76 (25%) of the 307 respondents strongly agreed, 146 (48%) agreed, 22 (7%) were undecided, 16 (5%) strongly disagreed, and 47 (15%) disagreed with the premise that strategic recruitment has a significant bearing on firm survival.

	Gnakk En	terprises	Dreamvil	le School	Vitus Co	smetics	Osi Ve	ntures	Chijioke Foo	Frozen ds		Percentages
	Admin. Staff	L-C. Staff	Total	(%)								
SA	2	10	3	52	2	15	1	9	-	18	46	15
А	2	12	1	15	2	43	2	15	2	2	91	29.6
UD	-	5	3	5	-	10	1	3	-	-	32	10.4
SD	-	4	1	5	-	-	2	10	1	-	97	31.6
D	-	11	-	15	1	20	-	2	-	-	41	13.4
Total	4	42	8	92	5	88	6	39	3	20	307	100

Table 2. It is vital for an organization's success to attract and retain high-performing employees.

Source: Field Survey, 2021. Where: SA-Strongly agree; A-Agree; UD-Undecided; SD-Strongly disagree; D-Disagree.

	Gnakk En	iterprises	Dreamville School		Dreamville School Vitus Cosmetics Osi V		Osi Ve	tures Chijioke Froze Foods		Frozen ds		Percentages
	Admin. Staff	L-C. Staff	Admin. Staff	L-C. Staff	Admin. Staff	L-C. Staff	Admin. Staff	L-C. Staff	Admin. Staff	L-C. Staff	Total	(%)
SA	2	10	3	20	1	18	5	15	-	2	76	24.8
А	2	10	4	30	3	60	-	19	3	15	146	47.6
UD	-	5	-	10	1	5	-	-	-	1	22	7.2
SD	-	2	-	12	-	-	-	-	-	1	16	5.2
D	-	15	1	20	-	5	1	5	-	-	47	15.3
Total	4	42	8	92	5	88	6	39	3	20	307	100

Table 3. Strategic recruitment significantly affects firm survival.

Source: Field Survey, 2021. Where: SA-Strongly agree; A-Agree; UD-Undecided; SD-Strongly disagree; D-Disagree.

Table 4 displays the descriptive statistics for succession management and its relationship with corporate sustainability. Corporate sustainability received a mean response rate of 2.9 with a standard deviation of 1.7, while succession management received a mean response rate of 3.3 with a standard deviation of 1.5. Based on standard deviation values of 1.7 and 1.5, it may be deduced that the dependent and independent variables have almost the same variability of data points. Hence, succession management is a larger part of the equation when it comes to factors that affect a company's long-term viability.

Table 4. Descriptive statistics.

	Mean	Std. Deviation	Ν
Corporate sustainability	2.9348	1.71791	307
Succession management	3.2600	1.47450	307

The Pearson correlation coefficient result shows the link between succession management and long-term corporate sustainability. Table 5 above shows the number of cases, the correlation coefficient, and the significant values. The correlation value of 0.934, indicates that the association is significant at a 0.05 level (2-tailed) and demonstrates a high positive link (r = 0.934, *p*-value = 0.000) between succession management and corporate sustainability. Hence, the null hypothesis is rejected and the alternate hypothesis that succession management relates positively to corporate sustainability, is accepted.

Table 5. Correlations.

		Corporate Sustainability	Succession Management
Corporato sustainability	Pearson's Correlation	1	0.934 **
Corporate sustainability	N	307	307
Succession management	Pearson's Correlation	0.934 **	1
ouccession management	N	307	307

**. at the 0.05 level, the correlation is significant (2-tailed).

6. Discussion

The findings relate to determining how succession management influences the corporate sustainability of small and medium enterprises (SMEs) in Lagos State, Nigeria. The study's hypotheses were tested using the Pearson product moment correlation coefficient, which indicated that succession management is associated with corporate sustainability of SMEs in Lagos State, Nigeria (r = 0.934, *p*-value < 0.05). The stated result and statistical figures show that every succession management strategy conducted or practiced in SMEs follows with the enterprise's sustainability. These research findings support Syeda and Abida's [43] investigation, which offers a link between efficient succession planning and employee satisfaction exists {F (1, 43) = 24.070, p 0.001}, positive succession planning supports effective leadership, and leadership is the most significant enabler of employee engagement on the job, in the business, and at work. Akinyele et al. [27] found that succession planning had a significant impact on organizational survival in a study that observed the stated variables. While based on Toliver's [45] findings, managers established and maintained a leadership position for high-potential individuals in imminent management roles by tracking in-house personnel with fundamental leadership abilities, re-engineering the in-house succession pipeline, and establishing replacement design mechanisms. Furthermore, this study's findings concur with Maguta [42], who asserts that succession planning, which involves retaining employees and managing organizational problems, is essential, and that meeting shareholder expectations has a positive relationship with an NGO's performance (r = 0.743, r = 0.641, and r = 0.866). In addition, it proffers that succession planning reduces employee turnover rates by enhancing staff retention, which influences NGOs' performance and sustainability. This study's result aligns with findings from oral interviews that: succession management helps SMEs to reduce the potential for costly errors by incentivizing employees to invest in firm-specific human capital roles to meet future needs, strengthens the internal leadership bench, minimizes potential negative effects that it may cause, supports diversity in the enterprise, and accelerates development.

However, the outcomes of this investigation contradict the findings of Garg and Weele [46] who posited that in small and micro enterprises investigated in Johannesburg, South Africa, there was a disconnect between how succession development was viewed and how it actually worked, stressing that these small and micro enterprises did not have established procedures to groom, train, and generate qualified successors.

6.1. Contribution of the Study to the Body of Knowledge

This study focuses on succession management and corporate sustainability, assessing its impact on SMEs in Lagos State, Nigeria. In line with the call for research and the paucity in scholarly investigations on the construct variables (succession management and corporate sustainability), sector (small and medium enterprises) and geographical scope of the study (Lagos State, Nigeria), this research establishes a theoretical foundation and empirical evidence for a link between succession management and corporate sustainability in SME's, particularly in the current economic downturn.

6.2. Limitations of the Study

There was a paucity of research materials on the subject matter under investigation, coupled with an unwilling attitude among respondents. However, this was curbed by sourcing relevant related literature online to buttress the discourse, while regular meetings were held with respondents to elicit their feedback relating to the research.

7. Conclusions

Succession management is arguably human capital development's brainbox and has been at the heart of corporate sustainability. Hence, in a fast-paced business environment, succession planning makes it easier to integrate all systems and make well-informed decisions regarding new or familiar changes in company and workforce management. During the period of this present economic recession, the small and medium enterprises (SMEs) under study, namely Gnakk Enterprises, Dreamville School, Vitus Cosmetics, Osi Ventures, and Chijioke Frozen Foods have expanded in every facet of their operations, either in terms of market share or customer patronage, thereby advancing their productivity. Thus, this study concludes that succession management, in respect of strategic recruitment, attraction, and retention of high-performing staff, is a pivotal strategy that delivers corporate sustainability. This is a consequence of the study's population responses who are stakeholders of the SMEs in Lagos State, Nigeria, as they agreed and strongly agreed with the research data that formed the basis for inference making and result generalization. Hence, SMEs should adopt succession management that is proper, efficient, and consistent by proactively ascertaining, evaluating, and implementing succession plans and training employees to take on or succeed in critical leadership positions, as this ensures the company's longevity and sustainability in any business environment.

Author Contributions: Conceptualization, F.E.M.; data curation, S.O., L.E.K.-I. and H.O.O.-A.; formal analysis, F.E.M. and S.O.; funding acquisition, W.I.U.; investigation, F.E.M. and E.K.A.; methodology, E.K.A. and H.O.O.-A.; project administration, E.K.A.; resources, W.I.U., L.E.K.-I. and H.O.O.-A.; software, W.I.U., S.O. and L.E.K.-I.; supervision, E.K.A.; validation, W.I.U.; writing—original draft, F.E.M.; writing—review and editing, W.I.U. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Institutional Review Board Statement: The study was conducted according to the guidelines of the Declaration of Helsinki, and approved by the Ethics Committee of the Department of Management, University of Nigeria, Enugu Campus (protocol code: DM-2018-086 and date of approval: 10 September 2018).

Informed Consent Statement: Informed consent was obtained from all subjects involved in the study.

Data Availability Statement: Primary data was used for the study which was provided by persons in the study. As such, there are no links to publicly archived datasets analyzed or generated during the study.

Conflicts of Interest: The authors declare no conflict of interest.

References

- Nwaise, N.D. Empowering Nurses through Knowledge and Technology to Decrease Fall Rates. A Thesis Submitted for the Degree
 of Doctor of Nursing Practice at Walden University, USA. 2017. Available online: http://scholarworks.waldenu.edu/dissertations
 (accessed on 18 July 2020).
- Sjåfjell, B.; Mähönen, J.; Novitz, T.; Gammage, C. Securing the Future of European Business: SMART Reform Proposals. 2020. Available online: https://www.accountancyeurope.eu/wp-content/uploads/Draft-SMART-report-Business-4-March-2020.pdf (accessed on 2 April 2021).
- Monyei, E.F.; Agbaeze, K.E.; Isichei, E.E. Organisational paranoia and employees' commitment: Mediating effect of human resources policies. Int. J. Sci. Technol. Res. 2020, 9, 2277–8616.
- 4. Kehinde, J. Talent management: Effect on organizational performance. J. Manag. Res. 2012, 4, 10–52. [CrossRef]
- Sheehy, B. CSR and Environmental Law. In *The Oxford Handbook of Corporate Social Responsibility*; Crane, A., Matten, D., McWilliams, A., Moon, J., Siegel, D.S., Eds.; OUP: Oxford, UK, 2019.
- 6. Hakovirta, M.; Denuwara, N. How COVID-19 Redefines the Concept of Sustainability. Sustainability 2020, 12, 3727. [CrossRef]
- Jyoti, J.; Rani, R. Exploring talent management practices: Antecedents and consequences. Int. J. Manag. Concepts Philos. 2014, 8, 220. [CrossRef]
- Journals, I.; Dauda, A. Business Continuity and Challenge of Succession in Nigeria: What happens when the CEO leaves. *IOSR J. Bus. Manag.* 2013, *8*, 88–101. [CrossRef]
- Mohrman, A.S.; Lawler, E.E., III. Designing organizations for sustainable effectiveness. A new paradigm for organizations and academic researchers. J. Organ. Eff. People Perform. 2014, 1, 14–34.
- Lazear, E.P.; Shaw, K.L. Personnel economics: The economist's view of human resources. J. Econ. Perspect. 2007, 21, 91–114. [CrossRef]
- 11. Aithal, P.; Kumar, P.M.S. Theory A for optimizing human productivity. IRA-Int. J. Manag. Soc. Sci. 2016, 4, 526–535. [CrossRef]
- 12. Bansal, P.; Song, H.C. Similar but not the same: Differentiating corporate sustainability from corporate responsibility. *Acad. Manag. Ann.* 2017, *11*, 105–149. [CrossRef]

- 13. Ruban, D.A.; Yashalova, N.N. Society and environment in value statements by hydrocarbon producers. *Extr. Ind. Soc.* 2021, *8*, 100873. [CrossRef]
- 14. Meseguer-Sánchez, V.; Gálvez-Sánchez, F.J.; López-Martínez, G.; Molina-Moreno, V. Corporate Social Responsibility and Sustainability. A Bibliometric Analysis of Their Interrelations. *Sustainability* **2021**, *13*, 1636. [CrossRef]
- SMEDAN. SMEDAN and National Bureau of Statistics Collaborative Survey and Selected Findings of 2013. 2015. Available online: https://nigerianstat.gov.ng/pdfuploads/SMEDAN%202013_Selected%20Tables.pdf (accessed on 12 October 2021).
- Collings, D.; Mellahi, K. Strategic talent management: A review and research agenda. Hum. Resour. Manag. Rev. 2009, 19, 304–313. [CrossRef]
- 17. Bebbington, K.J.; Unerman, J. Achieving the United Nations Sustainable Development Goals: An enabling role for accounting research. *Account. Audit. Account. J.* 2018, *31*, 2–24. [CrossRef]
- 18. Bass, B.M. The future of leadership in learning organizations studies. J. Leadersh. Organ. 2000, 7, 18-40.
- 19. Hoskisson, R.E.; Eden, L.; Lau, C.M.; Wright, M. Strategy in emerging economies. Acad. Manag. J. 2000, 43, 249-267.
- 20. Atkinson, A.B. Economics as a Moral Science. Economica 2009, 76, 791–804. [CrossRef]
- Muñoz-Pascual, L.; Curado, C.; Galende, J. The Triple Bottom Line on Sustainable Product Innovation Performance in SMEs: A Mixed Methods Approach. Sustainability 2019, 11, 1689. [CrossRef]
- 22. Wernerfelt, B. A resource-based view of the firm. Strateg. Manag. J. 1984, 5, 171-180. [CrossRef]
- El-Haggar, S.; Samaha, A. Roadmap for Global Sustainability—Rise of the Green Communities; Springer: Berlin/Heidelberg, Germany, 2019.
- Shim, J.; Moon, J.; Lee, W.; Chung, N. The Impact of CSR on Corporate Value of Restaurant Businesses Using Triple Bottom Line Theory. Sustainability 2021, 13, 2131. [CrossRef]
- Rahdari, A.; Rostamy, A.A.A. Designing a general set of sustainability indicators at the corporate level. J. Clean. Prod. 2015, 108, 757–771. [CrossRef]
- 26. Huang, T. Succession management systems and human resource outcomes. Int. J. Manpow. 2001, 22, 736–747. [CrossRef]
- Akinyele, S.T.; Ogbari, M.; Akinyele, F.A.; Dibia, K. Succession planning and its impact on organizational survival. J. Res. Nat. Dev. 2015, 13, 1596–8303.
- Ahsan, M. Effective recruitment and selection along with succession planning towards leadership development, employee retention, and talent management in Pakistan. J. Entrep. Organ. Manag. 2018, 7, 233.
- Rothwell, W.J. Effective Succession Planning: Ensuring Leadership Continuity and Building Talent from Within; Amacom: New York, NY, USA, 2005.
- Ioannou, I.; Serafeim, G. Yes, Sustainability Can Be a Strategy. Harvard Business Review. 11 February 2019. Available online: Hbr.org/2019/02/yes-sustainability-can-be-a-strategy (accessed on 12 April 2020).
- Gadenne, D.; Mia, L.; Sands, J.; Winata, L.; Hooi, G. The influence of sustainability performance management practices on organisational sustainability performance. J. Account. Organ. Chang. 2012, 8, 210–235. [CrossRef]
- Rahdari, A.; Sepasi, S.; Moradi, M. Achieving sustainability through Schumpeterian social entrepreneurship: The role of social enterprises. J. Clean. Prod. 2016, 137, 347–360. [CrossRef]
- Hahn, T.; Figge, F. Beyond the Bounded Instrumentality in Current Corporate Sustainability Research: Toward an Inclusive Notion of Profitability. J. Bus. Ethic. 2011, 104, 325–345. [CrossRef]
- Amini, M.; Bienstock, C.C. Corporate sustainability: An integrative definition and framework to evaluate corporate practice and guide academic research. J. Clean. Prod. 2014, 76, 12–19. [CrossRef]
- 35. Kuhlman, T.; Farrington, J. What is Sustainability? Sustainability 2010, 2, 3436–3448. [CrossRef]
- 36. Dyllick, T.; Hockerts, K. Beyond the business case for corporate sustainability. Bus. Strategy Environ. 2002, 11, 130–141. [CrossRef]
- 37. Vermeulen, W.J.V.; Witjes, S. On addressing the dual and embedded nature of business and the route to-wards corporate sustainability. J. Clean. Prod. 2016, 112, 2822–2832. [CrossRef]
- 38. Lawler, E.E.; Worley, C.G. Designing organizations for sustainable effectiveness. Organ. Dyn. 2012, 41, 265–270. [CrossRef]
- 39. Oyewo, B.M.; Badejo, S.O. Sustainable development reporting practices by Nigerian banks. Med. J. Soc. Sci. 2014, 5, 23–35.
- Crane, A.; Matten, D. Business Ethics: Managing Corporate Citizenship and Sustainability in the Age of Globalization; Oxford University Press: Oxford, UK, 2016.
- Gallego-Sosa, C.; Gutiérrez-Fernández, M.; Fernández-Torres, Y.; Nevado-Gil, M.T. Corporate Social Responsibility in the European Banking Sector: Commitment to the 2030 Agenda and Its Relationship with Gen-der Diversity. *Sustainability* 2021, 13, 1731. [CrossRef]
- 42. Maguta, H.M. Effects of Succession Planning on the Performance of Non-Governmental Organizations in Kenya. Int. J. Sci. Res. Publ. 2016, 6, 2250–3153.
- Syeda, S.G.; Abida, D. Impact of Succession Planning on Employee Engagement in Telecommunication Sector in Rawalpindi, Pakistan. Eur. J. Bus. Manag. 2014, 6, 2222–2839.
- Hanaysha, J. Improving employee productivity through work engagement: Empirical evidence from higher education sector. Int. J. Indust. Eng. Comput. 2016, 6, 61–70. [CrossRef]

- Toliver, Y.W. Succession Planning for Next Generation Business Leaders. Doctorate Degree Thesis Presented to the College of Management and Technology. Minneapolis, United States. Walden University. 2017. Available online: http://scholarworks. waldenu.edu/dissertations (accessed on 10 January 2018).
- Garg, A.K.; Weele, E.V. Succession planning and its impact on the performance of small micro medium enterprises within the manufacturing sector in Johannesburg. Int. J. Bus. Manag. 2012, 7, 1833–8119. [CrossRef]



Article



The Influencing Factors on Impulse Buying Behavior of Consumers under the Mode of Hunger Marketing in Live Commerce

Zhenfang Zhang, Nan Zhang and Jiguang Wang *

School of Economics and Management, Shanxi University, Taiyuan 030006, China; zhangzhenfang@sxu.edu.cn (Z.Z.); zn17835396231@163.com (N.Z.)

* Correspondence: wangjg@sxu.edu.cn

Abstract: The sustainability of live commerce is closely dependent on the impulse buying behavior of consumers because of its live characters compared to the traditional e-commerce mode. Hunger marketing is a widely adopted mode in live commerce, however, the influencing factors on impulse buying behavior of consumers under the mode of hunger marketing in live commerce is currently lacking. Here, based on SOR theory, we carefully studied how external stimulus variables (anchor characteristics, online comments, logistics service quality, promotion incentive information, and promotional time limit) affected consumers' impulse buying behavior under the inner mechanism reaction factors (perceived trust and perceived value). By collecting questionnaires from 533 Chinese online consumers, we processed these data using SPSS23.0 and analyzed them using the structural equation model (SEM) generated by AMOS 25.0. The results showed that extrinsic stimulus variables (except for the promotional time limit variable) have significant positive effects on impulse buying behavior through partial or total positive effects on intrinsic mechanisms. We analyzed the reasons for all these results in detail, and thought the reason that the promotional time limit did not have a significant impact on the perceived value might be due to the rapid increase in the frequency of live commerce. Moreover, we also put forward the suggestions on the basis of these influencing factors to improve the sustainable development of live commerce.

Keywords: live commerce; hunger marketing; perceived trust; perceived value; impulse buying behavior

1. Introduction

The sustainable development of e-commerce requires enterprises to innovate marketing forms and models constantly. There are many problems in the traditional retail industry, such as high cost, low realizable profit, a poor precipitation effect and a small radiation range. With the rapid development of network technology and mobile terminals in recent years, the mode of live commerce provides new opportunities for the sustainable and healthy development of e-commerce. Live commerce is a new type of social interaction platform that provides a way to promote sustainable product consumption online [1]. It breaks through the limitations of time and space, and creates a refined live commerce marketing system of "private sphere + retail + full scene" for consumers [2]. This model combines a stereo environment, preferential price, diversified products, fast payment methods, convenient logistics services and other aspects, giving consumers a strong sense of presence, and stimulating consumers' purchase intention, thus driving product sales and obtaining rich profits. 2016 was the first "Year of Live Commerce" in China [3]. The ability of e-commerce platforms to cash in goods has been truly realized since the rise of the "Double 11" online shopping festival in 2017. Since the outbreak of the epidemic in 2020, offline consumer demand has been suppressed, making the facilitating role of online live streaming of goods more obvious. How to maintain the sustainable development of live commerce is a problem that should be paid attention to at present, and this requires the common attention of both industry practitioners and academia.

Citation: Zhang, Z.; Zhang, N.; Wang, J. The Influencing Factors on Impulse Buying Behavior of Consumers under the Mode of Hunger Marketing in Live Commerce. *Sustainability* **2022**, *14*, 2122. https://doi.org/10.3390/ su14042122

Academic Editors: Iwona Zdonek and Adam R. Szromek

Received: 11 January 2022 Accepted: 9 February 2022 Published: 13 February 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). It can be said that live commerce as an emerging business model not only brings huge profits but also attracts a large amount of user traffic. Although it is difficult to find a replacement in a short period of time, live commerce has also encountered some shortcomings while prospering. Its sustainable development has become the focus of attention from all walks of life, and only a profound analysis of the problems aimed at developing a sound operation mechanism that plans the right direction of development will help with these issues. Live commerce has a high acceptance among contemporary consumer groups, so research based on the perspective of users is more conducive to revealing the industry's shortcomings, helping it break through the bottleneck period and achieve long-term sustainable development.

In order to maintain the sustainable development of e-commerce live commerce, it has become a hot topic to study the purchasing behavior of consumers. Through situational experiments and relevant secondary data, Huang et al. found that the recommendation of hedonic products by reputable anchors can help trigger the recognition mechanism of consumers and improve their purchase intention and product sales [4]. Yan et al., (2021) believed that the popularity of anchors, interactivity and personalized service in live commerce had a positive impact on consumers' purchasing behavior to varying degrees [5]. In this process, consumers' perceived value plays a mediating role between the characteristics of live commerce and purchasing behavior. Liu et al., (2013) proved that commodity availability and website ease of use in online shopping features significantly affect consumers' impulse buying [6]. Zheng (2021) found in his research that the higher the degree of interaction with users in the broadcast room, the more likely users will have flow experience and the more likely they will form cross-purchase intention [7].

Previous research has shown that different marketing models can lead to different purchasing behaviors of consumers. In the traditional marketing mode, consumers' purchasing behavior is attracted by advertising marketing, direct marketing, and other strategies [8]. For live commerce, there are also various types of goods. For example, in the traditional ecommerce marketing model, some brands simply introduce their products for the purpose of doing so. Specifically, the anchors show, explain and try on the products in a full range in the form of live commerce so as to deepen consumers' understanding of the products and brands. However, there are no preferential activities. In this single mode, it is difficult to arouse consumers' shopping intentions. In addition, when e-retailers adopt the influence of marketing incentives, situational influence and individual characteristics, live commerce is more likely to trigger consumers' impulsive buying behavior [9]. It can be seen that under different sales modes, consumers' purchasing behaviors are very different, so it is more accurate to study the purchasing behaviors under specific modes.

Currently, the single live commerce mode has gradually entered a soft state. In order to break through the sales bottleneck, "hunger marketing" has become a common marketing mode in live commerce. The "hunger marketing" strategy comes from the "utility theory" in economic theory, through making consumers feel satisfied [10] in order to build a solid psychological foundation for marketing. Hunger marketing strategy, in the final analysis, is a kind of aggregation promotion, mainly manifested as the combination of artificially low price control and supply restriction [11]. In live commerce, businesses rely on the platform's flow base to create a tense atmosphere and artificial phenomenon of short supply, which runs through three stages: early warm-up, middle empathy and late feedback [12]. For e-retailers, adopting the hunger marketing model can bring many advantages. First, by controlling retail price and total sales volume, the total profit of the supply chain is increased and the double marginalization effect is aggravated. Yu and Zhang (2018) studied the impact of the hunger marketing strategy on supply chain pricing, and coordinated the supply chain through a two-period pricing model [13]. Tolety (2016) studied hunger marketing, time-limited sales and cost optimization strategies of Xiaomi [14]. Chen (2016) studied the application analysis of hunger marketing in the mobile phone industry [15]. However, as a new strategy of live commerce, there are few studies on hunger marketing

strategy as the preset background. As a new development trend, it is more meaningful to study the impact of the hunger marketing strategy adopted by live commerce on consumers' purchasing behavior.

The buying behavior of consumers can be divided into purported buying and impulsive buying [16]. Intentional purchase refers to the planned purchase behavior of consumers. Impulse buying behavior tends to be unplanned but is carried out because consumers are aroused by positive emotions and have an immediate desire to have [17] a particular item. Wells and Veena (2011) defined impulse buying as the consumer being spontaneous and temporarily out of control, and having a strong desire to buy. Impulse buying behavior is very common [18]. Relevant (2018) studies show that modern consumers clearly pay attention to hedonism and realizable consumption styles [19]. Impulse buying accounts for about 80% of product sales, and the sales of new products are basically impulse purchases [20]. Jiang and Zhao (2013) found that the images of goods, services, convenience and safety have a significant impact on consumers' intentional purchases, and a virtual experience can stimulate consumers' impulsive purchase behavior [21]. Compared with offline shopping, online shopping consumers, as a result of time and space constraints, showed more obvious characteristics of impulse buying, especially with regard to the process of a live business platform., Businesses take advantage of these tendencies to motivate consumers to make impulsive buying decisions, and push them further into impulse buying behavior. In combination with the merchant hunger marketing model in live commerce, this article defines impulse buying behavior as the consumer's strong and eager purchase behavior under the stimulation of external factors on the premise of the absence of any clear purchasing plan on behalf of the consumer.

The hunger marketing model is often used in traditional marketing, but it rarely gets attention under the background of live commerce with goods. However, under the marketing environment of limited time and limit, most consumers are more likely to engage in impulsive consumption behavior. In addition, the existing studies adopt a relatively independent and decentralized approach, that is, they only study related factors separately, such as store environment [22], promotion intensity [23], time scarcity [24], consumer trust [25], etc. Few studies integrate these interrelated concepts. Taking SOR as the theoretical basis, we established a model and set the research background in the case of hunger marketing strategy in the mode of live commerce to comprehensively explore the influence factors of external stimulus factors and internal mechanism on consumers' impulse buying behavior. This paper enriches the research field of consumer behavior in the context of e-commerce, provides precise supplement for consumers to make impulsive behavioral decisions in the online shopping environment, and helps e-commerce platforms and enterprises to change their innovation mode and realize sustainable development of e-commerce platforms.

This paper is divided into the following parts: The first part is the introduction. The second part proposes a theoretical model and hypothesis research. The third part is questionnaire design and data collection, and the fourth part tests and analyzes the collected data. The fifth part analyzes the hypotheses and conclusions. The sixth part puts forward accurate suggestions on the sustainable development strategy of hunger marketing under the current e-commerce model and points out the limitations of this paper.

2. Theoretical Basis and Hypothesis Research

2.1. Theoretical Basis of SOR

The SOR (stimuli-organism-response) theory is a cognitive theory of psychology that studies human behavior by adding the intermediate process of "individual psychological activity" on the basis of "stimulus-response" theory, emphasizing the mediating role of individual physiological and psychological activities. The SOR theory holds that the behavioral decision-making (i.e., response) whereby consumers decide to approach or avoid comes not only from the attention to external environmental factors (i.e., stimulation), but also from the sensory cognition of consumers' internal state (i.e., organism). The SOR model is considered to be successful in explaining behavioral differences caused by various marketing stimuli and cognitive factors. The main advantage of the SOR model is that it is flexible and includes the opportunity to examine a variety of internal and external stimuli. The SOR theory is widely used to explain consumer behavior or consumer will, particularly based on the impact of external environmental characteristics on consumers' internal psychological state and behavioral response [26]. The SOR theory is widely used to explain consumer behavior or consumer will, especially based on the impact of external environmental characteristics on consumers' internal psychological state and behavioral response. Jing and Sung (2019) explored the effect of service recovery on customer behavioral intention in online shopping malls under the mediating effect of customer forgiveness based on the SOR model [27]. Hu et al., (2012) studied the influence of information content and information form on consumers' purchasing behavior under the mediation of channel mode through SOR theory [28].

Consumers' behavioral motivation is controlled by self-consciousness, and conscious cognition is the intermediary. When commodity browsers are stimulated by the external environment, they will rely on subjectivity to form a "cognitive structure". At this time, they do not act directly, but deliberately and actively obtain effective information from the stimulation process and consider whether to choose commodities, and then respond. "Stimulation" in the retail environment can often attract consumers' attention and cause changes in their psychological feelings. This stimulation is usually via external environmental factors, including marketing strategies and other objective effects; "Organism" is an individual's internal activity process. It is a cognitive process in which consumers consider themselves, their money and risk after receiving the information they see or hear; "Response" includes psychological and behavioral responses, that is, the decisions consumers will make after processing the information.

2.2. Research Assumptions

2.2.1. Research Assumptions of Anchor Characteristics

An e-commerce influencer is an important front in live commerce [29]. By virtue of a person's professionalism, popularity, interactivity and affinity, they interact with consumers in real time through live commerce platforms, serving as a bridge between businesses and consumers. Influencers help merchants to promote products and deliver product information and enhance their brand image. Meanwhile, as the executer of the hunger marketing strategy, e-commerce influencers help consumers to know more about products and stimulate consumer demand. Enthusiasm is based on the personality characteristics of the influencer, bringing consumers an enhanced shopping experience, and influencing their attitudes and bolstering the strength of the products they represent. They also enhance the scarcity of the product and add a sense of immediacy to the impending purchase. This is all designed to strengthen the perceived value of a product and guide consumers to make impulsive shopping decisions [30]. Zhang and Zhang (2021) pointed out that influencers play an important role in sales performance in network broadcasts, and categorized the characteristics of influencers into Internet celebrities' charm, entertainment and professional knowledge, confirming that these three characteristics have a positive impact on consumers' purchase intention [31]. Zhao and Wang (2021) divided influencer features into interactivity, authenticity, professionalism and popularity, and verified that influencer = characteristics had a positive impact on consumers' perceived trust and perceived value from four dimensions, thus promoting their purchase intention [32]. Based on this, the following hypotheses are proposed:

Hypotheses 1 (H1). Anchor characteristics have a significant positive impact on consumers' perceived trust.

Hypotheses 2 (H2). Anchor characteristics have a significant positive impact on consumers' perceived value.

2.2.2. Hypothetical Research on Online Comment Factors

As the carrier of information dissemination, online comments have attracted increasing attention from companies and consumers. The function of online comments not only reduces consumers' uncertainty about product quality and experience attributes [33], but also significantly affects the company's operational decisions and helps reshape business growth [34]. Combined with the characteristics of live commerce, this study defines online comments as real-time comments sent by consumers during live watching. The comments are displayed on the screen in the form of subtitles to each consumer who watches the live commerce, which provides help for the online interactions between businesses and consumers. It provides real-time answers to relevant information about products and services and creates a shared shopping experience. Online product reviews are a powerful source of information for consumers. Personal reviews help consumers reduce product uncertainty and imperfect product page information [35]. Online reviews provide consumers with the possibility of real quality on e-commerce platforms and play an important role in customer purchase decisions [36]. Qi (2021) built a research model for consumer acceptance of online comments based on the TAM model, which proved that the quantity and quality of online comments has a significant positive impact on purchase behavior [37]. Shen and Chang (2021) showed that online comments and customer experience are important factors affecting consumers' online purchase intentions [38]. Customer value plays an intermediary role in the relationship between them. Accordingly, the following assumptions are proposed:

Hypotheses 3 (H3). Online comments have a significant positive impact on consumers' perceived trust.

Hypotheses 4 (H4). Online comments have a significant positive impact on consumers' perceived value.

2.2.3. Hypothesis Research on Logistics Service Quality

Logistics services play an important role in the development of live commerce. It is not only the key link and basic guarantee of online shopping [39], but also the last link to complete the online transaction. Consumers are the direct service objects of logistics services. They not only pay attention to product quality and price, but also have a variety of logistics needs. Businesses need to integrate and optimize the allocation of resources in the market and improve the consumption experience to meet the personalized and differentiated logistics service needs of customers. Meeting the logistics service needs of consumers will help to improve the relationship quality between customers and businesses and enhance consumer satisfaction. Similarly, online shopping can enhance consumers perceived value by improving the quality of logistics services. When consumers have a pleasant shopping experience, the degree of perceived value will increase. There are concessions before sales and guarantees after sales, and consumers are more likely to have impulsive buying behaviors. Liang and Zhang (2016) believe that logistics services play an important positive role in consumers' psychological contract, and one of the key elements of this psychological contract is consumers' trust [40]. Liu (2021) used a regression model analysis method to verify that personalized logistics services can effectively enhance consumers' willingness to make repeat purchases under the intermediary influence of relationship quality [41]. At the same time, personalized logistics services also affect consumer satisfaction. In this process, consumer trust plays an intermediary role. Yan et al., (2020) divided the refined logistics service elements into reliability, responsiveness, timeliness and the attitude of distribution personnel, and verified the positive impact of the four elements on customer perceived value [42]. Accordingly, the following assumptions are proposed:

Hypotheses 5 (H5). Logistics service quality has a significant positive impact on consumers' perceived trust.

Hypotheses 6 (H6). Logistics service quality has a significant positive impact on consumers' perceived value.

2.2.4. Hypothesis Research on Promotion Incentive Information

Promotional incentive information is used by e-commerce enterprises to publicize promotional activities and introduce product information before and during live commerce, especially to highlight the preferential strength of a price discount [43]. E-commerce merchants attract users by influencing consumers' sensitivity to price [44] so as to improve consumers' participation and purchasing power. Especially in the hunger marketing mode under the e-commerce platform, in order to induce consumers to buy specific goods immediately, the promotion incentive is often stronger than daily purchase activities, which directly affects the perceived trust, value and purchase intention of price sensitive consumers. The promotional incentive information can make consumers perceive the temptation of the promotion to save money, the richness of promotion commodity categories, quantities and styles, and the interest of participating in rush buying activities in the broadcast room, so as to encourage public participation and consumption enthusiasm. Through this shortterm stimulation tool, the broadcast room will form a shopping phenomenon of gathering shopping, which overcomes the wait-and-see mentality of consumers to a certain extent, strengthening consumers perceived trust with a rush buying atmosphere, thereby reducing perceived risks and eliminating pre-purchase doubts, and making impulse purchase decisions more likely. Prasilowati et al., (2021) believe that price promotion has a significant impact on customer satisfaction [45]. Chen (2021) pointed out that price promotion and gift promotion in promotion incentives can effectively improve consumers' perceived savings and reduce perceived risk, so as to enhance consumers' purchase intention [46]. Accordingly, the following assumptions are made:

Hypotheses 7 (H7). *Promotion incentive information has a significant positive impact on consumers' perceived trust.*

Hypotheses 8 (H8). *Promotion incentive information has a significant positive impact on consumers' perceived value.*

2.2.5. Hypothetical Research on Promotion Time Limit

As a marketing method, live commerce induces consumers to make immediate purchase decisions based on a certain amount of price discounts only available for a limited time. As one of the elements of the decision-making situation, time pressure plays a regulatory role in decision-making behavior [47]. In this short-term decision-making environment, under the atmosphere of exaggerating economic interests from businesses, consumers have a strong sense of time urgency, which affects consumers ability to obtain complete information and process the information in the process of evaluating the product. Instead, they pay more attention to the influencer's description of the items in question [48]. At the same time, shortened decision-making time leads to the impairment of consumers' cognitive functions, resulting in the perceived opportunity cost of "regret if you don't buy". According to the deadline effect, the closer to the end of the promotion, the stronger consumers' sense of urgency to buy. This "scarcity opportunity" urges consumers to convert potential purchase cognition into an obvious purchase opportunity, which not only ignores the purchase risk, but also emphasizes the acquisition of perceived benefits, resulting in positive perceived value.

Lu and Huang (2014) emphasized that the shorter the promotion duration, the stronger consumers' perception of opportunity cost and the higher their cognitive response to the perceived value of the promotion [49]. Hong (2021) pointed out that when online retailers promote products under the condition of a short time limit promotion and high time pressure, it increases the likelihood that consumers will make a decision to purchase [50]. Accordingly, the following assumptions are proposed:

Hypotheses 9 (H9). The promotion time limit has a significant positive impact on consumers' perceived value.

2.2.6. Research Hypothesis of Perceived Trust

Trust is an important variable in the study of consumer behavior. It is generally regarded as a psychological state with positive expectations for the behavior or intention of others [51]. As a prerequisite for online consumers' purchase decision [52], consumer trust is described as a basic element of e-commerce, which can reduce the uncertainty of a purchase and help improve consumers' loyalty and acceptance. Lack of trust will lead to consumer doubt and thereby and have a negative impact on their purchase intention [53]. Live commerce is a real-time interactive mode, which contributes to consumers having a sense of subjectivity and intimacy in the interaction. It can be seen that the richer the information content, the more diverse the forms, the higher the timeliness, and the higher the trust of consumers. At the same time, consumers have the pleasure of perceived value based on their trust in this hunger marketing model. Feng and Lu (2020) believe that based on consumers' strong sense of social presence, the webcast environment meets consumers' demand for complete information and promotes consumers' trust in promotion links, which plays an intermediary role in consumers' impulse purchase decisions [54]. Therefore, the following assumptions are put forward:

Hypotheses 10 (H10). *Perceived trust has a significant positive impact on consumers' impulse buying behavior.*

2.2.7. Research Hypothesis of Perceived Value

Perceived value is widely used to predict users' adoption behavior. Product quality, service quality and perceived value have an important impact on customer satisfaction. As a key mechanism to shape these relationships, customer perceived value has become an important consideration in the use and interaction with social media [55]. Zeeithaml (1988) first defined customer perception as "the overall evaluation of product effectiveness based on customers' perception of product acceptance and giving" [56]. Bettman (1998) defined customer perceived value as "the value perceived or experienced by customers through the use of a service" [57]. In the time-limited marketing link of the e-commerce platform, perceived value can be regarded as the overall evaluation of product performance by weighing perceived revenue and perceived purchase cost. When consumers feel that the comprehensive evaluation of the economic preference of goods, logistics, after-sales service level and relevant useful information in the broadcast room is higher than their original expectation, it will produce positive perceived value and make them more likely to purchase the item in question. Wang et al. (2021) believe that promotional purchase restrictions, on the one hand, enhance perceived value by highlighting the scarcity of restricted goods, while on the other hand stimulating consumer excitement to satisfy their impulse purchase intention [58]. Accordingly, the following assumptions are proposed:

Hypotheses 11 (H11). *Perceived value has a significant positive impact on consumer impulse buying behavior.*

The above hypotheses, supported by the relevant literature, is explained and incorporated in the research model in Figure 1.



Figure 1. Research model.

3. Questionnaire Design and Data Collection

3.1. Questionnaire Design

In order to verify the assumptions that the researchers have with regard to consumers' online behavior, the questionnaire mainly includes two parts: the first part includes the survey participants' sample demographic characteristics such as gender, age, education, occupation, monthly average income, monthly average online consumption and an empty field where the person can add content, which might include their opinions or suggestions. The second part is the variables and the scale, including "anchor characteristics such as (AC)", "online comment (OC)" and "logistics service quality (LSQ)", "promotion incentive information (PII)", "promotion time limit (PTL)" and "perceived trust (PT)", "perceived value (PV)", "and impulse buying behavior (IBB)". The questionnaire measurement mainly refers to the design of 7-level Likert scale, ranging from (1) completely disagree to (7) completely agree. The detailed description of the questionnaire items is shown in Appendix A.

3.2. Questionnaire Distribution and Data Collection

In order to further ensure the quality of the formal questionnaire samples, the preliminary survey of the questionnaire was carried out. This link will make the questionnaire available in online and offline channels. Online distribution is distributed on major social platforms, while offline distribution is targeted at college students and residents in surrounding communities. In the pre-survey, a total of 80 questionnaires were collected, and five invalid questionnaires were excluded from the survey. The recovery rate of valid questionnaires was 93.8%. Meanwhile, the data passed the skewness coefficient, kurtosis coefficient and reliability test.

The questionnaire was distributed mainly through online channels, primarily to consumers through various online social platforms. Data collection of the questionnaire lasted 75 days from 22 July to 4 October 2021, and a total of 593 users filled in the questionnaire. After screening, 533 valid questionnaires were screened out, with an effective recovery of 89.9%.

3.3. Sample Description

The first part of the questionnaire fully reflects the sample division of 533 valid questionnaires. In terms of gender, males and females accounted for 47.8% and 52.2%. Most of the respondents' ages are between 18–29 (52.9%). 61.2% of interviewees have a Bachelor's degree or junior college degree, and 18.9% of them have a Master's degree or higher, indicating that most of the interviewees have received a good education. By occupation, students were the major participants of this study (35.3%). For further demographic information, see Table 1.

NO.	Characteristics	Category	Frequency	%
1	C 1	Male	225	47.8
1	Gender	Female	278	52.2
		Less than 18 Years	10	1.9
		18–29 Years	282	52.9
2	Age	30–39 Years	114	21.4
		40–49 Years	48	9.0
		50 + Years	79	14.8
		Junior high school	15	2.8
3	Highest Education	High school/technical secondary school	91	17.1
5	Tignest Education	Undergraduate	326	61.2
		Postgraduate	101	18.9
		Students	188	35.3
		Civil servant	51	9.6
		Teachers, doctors, banker	56	10.5
4	Occupation	Staff	156	29.3
		Retirees	19	3.6
		Operators	31	5.8
		Others	32	6.0
		0–1000 Yuan	76	14.3
		1001–3000 Yuan	168	31.5
5	Income	3001–5000 Yuan	139	26.1
		5001–7000 Yuan	85	15.9
		7000 + Yuan	65	12.2
		Less than 500 Yuan	284	53.3
	Online Shopping	500–1500 Yuan	147	27.6
6	Consumption	1501–3000 Yuan	68	12.8
	Consumption	3001–5000 Yuan	23	4.3
		5000+ Yuan	11	2.1

Table 1. Demographic description.

4. Results and Discussion

4.1. Reliability Test

A reliability test was used to judge the internal stability of questionnaire data and the reliability of measurement results.

Table 2 shows the mean value, standard deviation, Cronbach's α value and CITC value (the corrected term and the total correlation coefficient) of each item. It can be seen from the calculation results that the mean values of all variables are distributed between 4.33 and 5.12. The standard deviation, Cronbach's α value and CITC value of other variables were consistent with their reference standard values except for the PV1 of variables. These data prove that the questionnaire as a whole has a strong internal consistency. Considering that the value distribution of PV1 items in the perceived value is uneven and the CITC value is lower than the reference value of 0.5, this study deletes the PV1 item to improve the reliability of the data.

4.2. Validity Test

A validity test, as an important means of testing, aims to verify the accuracy of the information reflected in the research content to the required measurement object. This validity test includes convergence validity and discriminant validity.

4.2.1. Convergence Validity

Convergence validity is used to check the consistency within the plane. The factor load of the standardized coefficient reflects the importance of the variable on the common factor, and the factor load of all variables ranges from 0.662 to 0.807, all exceeding the threshold value 0.5. Combined reliability (CR) refers to the reliability of a new variable composed of a combination of several other variables, reflecting whether each measurement item consistently explains the potential variable, and whether the CR value of each variable exceeds the recommended value of 0.7. The AVE value of the mean variance extraction

volume refers to the average variance of the interpretation level of potential variables to observed variables, and the AVE value of each variable that is distributed is between 0.5231 and 0.599, all of which are more than 0.5. It can be judged that the convergence validity of the sample data meets the standard.

Latent Variables	Number of Items	Measuring Item	Mean	Standard Deviation	CITC	Cronbach's α after Deleting the Item	Cronbach's α
Anchor Characteristics (AC)	5	AC1 AC2 AC3 AC4 AC5	4.48 4.47 4.34 4.76 4.72	1.552 1.532 1.471 1.487 1.529	0.696 0.700 0.713 0.709 0.680	0.847 0.846 0.843 0.844 0.851	0.873
Online Comment (OC)	5	OC1 OC2 OC3 OC4 OC5	4.81 4.87 4.62 4.63 5.04	1.482 1.470 1.575 1.513 1.387	0.597 0.711 0.601 0.673 0.697	0.823 0.792 0.823 0.813 0.798	0.842
Logistics Service Quality (LSQ)	5	LSQ1 LSQ2 LSQ3 LSQ4 LSQ5	4.85 4.90 4.86 4.90 5.05	1.482 1.519 1.445 1.455 1.432	0.665 0.737 0.727 0.725 0.725	0.868 0.851 0.853 0.854 0.854	0.881
Promotional Incentive Information (PII)	5	PII1 PII2 PII3 PII4 PII5	4.33 4.55 4.56 4.60 4.66	1.579 1.574 1.521 1.539 1.493	0.631 0.695 0.679 0.698 0.727	0.850 0.834 0.838 0.833 0.826	0.865
Promotional Time Limit (PTL)	4	PTL1 PTL2 PTL3 PTL4	4.70 4.64 4.58 4.64	1.387 1.571 1.482 1.540	0.631 0.623 0.648 0.637	0.770 0.773 0.760 0.765	0.815
Perceived Trust (PT)	4	PT1 PT2 PT3 PT4	4.56 4.54 4.71 4.69	1.383 1.374 1.377 1.342	0.683 0.722 0.677 0.714	0.824 0.807 0.826 0.811	0.856
Perceived Value (PV)	4	PV1 PV2 PV3 PV4	4.80 4.93 5.12 4.89	2.192 1.331 1.333 1.387	0.403 0.684 0.588 0.619	0.827 0.619 0.666 0.646	0.742
Impulse Buying Behavior (IBB)	5	IBB1 IBB2 IBB3 IBB4 IBB5	4.70 4.98 4.83 4.95 4.97	1.608 1.534 1.562 1.525 1.482	0.711 0.752 0.765 0.771 0.709	0.879 0.869 0.866 0.865 0.879	0.895

4.2.2. Discriminative Validity

Discriminant validity is used to test the degree of difference between a given facet and other facets, and the observed facets can be distinguished from each other. As shown in Table 3, the AVE arithmetic square root values of the diagonal lines are much higher than the values of the relevant factors in the same column, indicating that each potential variable can be distinguished from each of the others by high discriminant validity.

Table 3. Discriminant validity.

	AVE	PTL	PII	LSQ	OC	AC	РТ	PV	IBB
PTL	0.525	0.725							
PII	0.523	0	0.723						
LSQ	0.599	0	0	0.774					
OC	0.523	0	0	0	0.723				
AC	0.578	0	0	0	0	0.761			
PT	0.540	0	0.231	0.281	0.059	0.506	0.735		
PV	0.542	0.098	0.480	0.274	0.173	0.095	0.246	0.736	
IBB	0.592	0.039	0.265	0.196	0.088	0.192	0.404	0.48	0.769

4.3. Modification of Structural Equation Model

The structural equation model (SEM) is based on the covariance matrix and is designed to analyze the relationship between variables. The sample covariance matrix is equivalent to the model covariance matrix and determines whether the sample data and the hypothesis path analysis model match. In order to increase the comparability between variables, the model was modified in this study. It can be seen from the results that the standardized factor load of potential variables of each measurement model is distributed between 0.664 and 0.833, which is higher than the standard value of 0.6, and the SMC (square multiple correlation) value is distributed between 0.441 and 0.685, higher than the standard value of 0.36. According to the revision suggestions in the analysis report, it can be seen that the PT3 measurement has a great correlation with PT1, PT2 and PT4, which should be removed according to the revision suggestions. The modified normalized path diagram and the fitting index of the structural equation model are shown in Figure 2 and Table 4.

Table 4. Structural equation model fit index.

Fitness Index	Standard Values	Actual Value	Results
Chi-square value(X2)	-	1610.673	Supported
Degrees of freedom (DF)	-	594	Supported
Significance level(P)	< 0.05	0.000	Supported
X2/DF	< 3	2.934	Supported
Fitness index (GFI)	>0.9	0.832	Acceptable
Root mean square of approximate error (RMSEA)	< 0.08	0.060	Supported
Comparative fitness index (CFI)	>0.9	0.891	Acceptable
Standard fit index (NFI)	>0.9	0.844	Acceptable
Increasing adaptation index (IFI)	>0.9	0.891	Acceptable
Simplified benchmark adaptation indicators (PNFI)	>0.5	0.779	Supported
Reduced fit index (PGFI)	>0.5	0.725	Supported
Noncanonical fit index (TLI)	>0.9	0.882	Acceptable
Adjusted fit index (AGFI)	>0.8	0.808	Supported
HORLTER	>200	200	Supported

The nonstandardized path coefficient is used to verify the existence of the influence relationship between variables, and the standardized path coefficient is used to judge the strength of the influence relationship between variables. According to the path coefficient data displayed by AMOS. using the maximum likelihood estimation method, the standardized and non-standardized path coefficients are plotted into tabulated statistics, and the results are shown in Table 5.

Table 5. Path coefficient test results.

Hypothesis	Content	Normalized Path Coefficient (Std.)	Nonstandardized Path Coefficient (UStd.)	Standard Error (S.E.)	Critical Ratio (C.R.)	p	Results
H1	PT < -AC	0.497	0.429	0.050	8.632	***	Supported
H2	PV < -AC	0.092	0.076	0.044	1.740	0.082	Not Supported
H3	PT < -OC	0.036	0.037	0.058	0.636	0.525	Not Supported
H4	PV < -OC	0.173	0.169	0.058	2.923	***	Supported
H5	PT < -LSQ	0.229	0.284	0.051	5.530	***	Supported
H6	PV < -LSQ	0.275	0.249	0.050	4.942	***	Supported
H7	PT < -PII	0.224	0.208	0.052	3.977	***	Supported
H8	PV < -PII	0.480	0.426	0.057	7.469	***	Supported
H9	PV < -PTL	0.097	0.094	0.051	1.831	0.067	Not Supported
H10	IBB < -PT	0.301	0.332	0.058	5.721	***	Supported
H11	IBB < -PV	0.412	0.477	0.063	7.629	***	Supported

Note: *** *p* < 0.001; *p* > 0.05 (not significant).



Figure 2. Normalized path coefficient diagram.

5. Conclusions and Empirical Analysis

According to the path coefficient test results of the research hypothesis model in Table 5, the test results and empirical analysis of this study can be interpreted as follows:

- 1. The influencer characteristics have a significant positive impact on consumers' perceived trust ($\gamma = 0.497$, p < 0.001). Therefore, H1 was supported. This also confirms that in the hunger marketing mode, influencers provide comprehensive and professional explanations of product information and activity information with their rich professional knowledge and sales experience to bring consumers a sense of intimacy. They have established a bond of trust between consumers on platforms, products and services [59].
- 2. The effect of influencer characteristics on consumers' perceived value fails to reach a significant level ($\gamma = 0.092$, p = 0.082). Thus, H2 was untenable. This may be because before the influencers explain the preferential content, the product links on the platform have more or less introduced the product information and created preferential intensity or created initial impressions [60], which makes the influencers have certain psychological expectations of consumers in the process of explaining. Therefore, the influencer factor is not obvious in consumers' determination of perceived value.
- 3. Online reviews have no significant effect on consumers' perceived trust ($\gamma = 0.036$, p = 0.525). H3 was not assumed. There are two reasons for this: on the one hand, there are a large number of participants in the broadcast room and the scrolling speed of online comments is fast, making it difficult for consumers to capture useful information. On the other hand, due to the difficulty of supervision, chaos emerges in endless streams, and there are more malicious bad comments from competitive businesses in the comments, resulting in a mix of bad information. It is difficult for consumers to distinguish the quality with regard to real-time reviews, resulting in reduced trust in real-time review valence.

- 4. Online comments have a significant positive effect on consumers' perceived value ($\gamma = 0.173$, p < 0.001). Therefore, H1 was supported. Consumers are immersed in a strong sense of social presence by participating in real-time online reviews. Personal reviews also have a strong impact on consumers' perceived value and purchasing decisions. This effect is particularly pronounced when individual reviews help consumers resolve product uncertainties or contrast with aggregated information that is readily available on product pages [61].
- 5. Logistics service quality has a significant positive impact on consumers' perceived trust ($\gamma = 0.229$, p < 0.001) and perceived value ($\gamma = 0.275$, p < 0.001). H5 and H6 were supported. Faced with consumer demands for various logistics services, e-commerce enterprises can not only guarantee basic logistics services, but also launch other differentiated value-added services. This not only improves consumers' credit evaluation of online shopping, but also helps to reduce consumers' perceived risk and guarantee the promise of after-sales support [35].
- 6. Promotional incentive information has a significant positive impact on consumers' perceived trust ($\gamma = 0.224$, p < 0.001) and perceived value ($\gamma = 0.480$, p < 0.001). H7 and H8 were supported. Most of today's consumers are price-sensitive with utilitarian value appeals. Online promotions are favored by hedonistic consumers [62]. The promotion and incentive information of e-commerce can be fully interpreted in the warm-up process and marketing process, and a proper promotion strategy will win the trust of consumers. The activity intensity of goods makes consumers perceive the lure of promotional benefits, which not only deepens their sense of identity to the enterprise, but also effectively improves consumers' perceived value and increases the likelihood that they will make impulse purchase decisions.
- 7. The promotional time limit has no significant positive effect on consumers' perceived value ($\gamma = 0.097$, p = 0.067). H9 was not accepted. In the broadcast room, the hunger marketing strategy will have a time limit on preferential products and create opportunities for scarcity in order to incite consumers to purchase in large quantities. However, it is generally seen that the frequency of live commerce and the number of product events by merchants have increased significantly compared with the past, which has increased the opportunities for consumers to snap up quantity restricted products, and the promotional time limit is not the only opportunity for them to purchase these particular items, leading consumers to postpone purchases indefinitely. In addition, the fact that consumers are busy will weaken their tendency to wait for time-limited discount purchases [63], and the perceived value level of the hunger marketing model will weaken.
- 8. Consumers' perceived trust ($\gamma = 0.301$, p < 0.001) and perceived value ($\gamma = 0.412$, p < 0.001) both have a significant positive impact on impulse buying behavior. H10 and H11 were supported. Whether because the live commerce scene simulates the offline stores, or whether merchants and influencers comprehensive introduction of product and activity rule information [64] enhance trust, or whether it is because of the influence of other factors, it will have a certain impact on consumers' trust. It is because of consumers' trust in this kind of hungry marketing mode that consumers' complete demand for online shopping is met. At the same time, the rapid development of the e-commerce network shortens the space and time distance between products and consumers, and endows them with higher value. Compared the amount of time, energy and money, when consumers get caught up in the overall feelings referred to in the hunger marketing model, and particularly feel the urge to make a purchase because of time restrictions under the constraints of the perception of an opportunity cost, this generate a positive perception of an item's value, and the feasibility of the impulse buying behavior will be stronger.

6. Research Suggestions and Limitations

6.1. Research Suggestions

From all the collected questionnaire information and the analysis of the research results, we can find that the consumer acceptance of the hunger marketing in live commerce is relatively high. However, some problems existing in some links in the purchase process will also affect consumers' continuous repurchase. In order to help the sustainable development of live commerce, we put forward the following suggestions for the shortcomings of the hunger marketing mode adopted by live commerce.

6.1.1. Anchor

The behavior of e-commerce anchors most directly affects consumers' awareness of product information and activity content. In recent years, e-commerce anchor has become a popular career which has a low threshold but high income, resulting in the uneven quality of live commerce anchors. Therefore, e-commerce platforms should hire anchors with a certain level of popularity, professional knowledge, and a good image and temperament. Meanwhile, in order to improve customer satisfaction, anchors should not only introduce product information and in detail on the live broadcast, but also possess other qualities that enable them to reply to customer questions accurately and in a timely manner, with emphasis on their positive interaction with consumers. If necessary, the mode of "professional anchor + popular flow" can be adopted to ensure the popularity of live commerce.

6.1.2. Online Comments

Online real-time comments are an important factor for consumers to integrate into the hunger marketing model. E-commerce merchants should attach importance to the interactions in the comment sections and actively respond to the questions found there. In order to attract more consumers to participate in live broadcasts, e-retailers can use the review section as a platform to enrich the form of activities add interactive links, so as to increase the purchase rate. In addition, businesses should also be vigilant and strengthen measures to reduce the occurrence of malicious fake reviews.

6.1.3. After-Sales Service

After-sales service quality has a significant impact on the performance of online retailers [65]. Ensuring after-sale service is a necessary measure to eliminate consumers' misgivings. Therefore, merchants should make wise measures, such as improving after-sales quality assurance security, adding seven-day return service without conditions attached, and adding complimentary freight insurance in order to enhance consumer confidence that they are having a worry-free purchase experience.

6.1.4. Promotional Incentive Information

The lower price and relevant preferential activities are often the things that consumers notice the most. However, due to the improper publicity of the enterprise in the early stage, many potential consumers do not receive the activity information, which in the loss of customers for the business. Therefore, businesses should promote through multiple channels that are widely spread through various ways to let consumers know the facts, improve the weight of livestreaming numbers, and create promotional policies. At the same time, high-quality live commerce incentives can help improve consumers' experience and increase their desire to buy. In the selection of products, businesses should be more strictly with the supply to ensure that product quality remains as the core priority. At the same time, they should ensure that the preferential policies are understandable and true, and make the price comparisons between live commerce platforms transparent. In the process of live commerce, they should try to increase the distribution of discounts and coupons for goods, and adjust the frequency of time-limited promotions according to the attributes of the goods [66]. When necessary, businesses can meet consumers' demand for "price

protection", establish a good brand image, improve pre-sale standards, improve after-sales mechanisms, and protect the interests of the platform, of the sellers, and of the consumer.

6.1.5. Promotion Time Limit

The setting of time limits for promotions is the most controversial aspect of e-commerce platforms. The length of promotion time will directly affect consumers' buying mentality. Setting time limits is the hard part of buying campaigns. If it lasts too long, consumers will think that waiting is a waste of their time. Too short a buying time will give people a sense of urgency. In addition, we found that the frequency settings of live commerce events also need to be optimized. The excessive frequency of live commerce of restricted goods will reduce consumers' perceived value of the item in question. However, if the frequency is low, consumers will be irritated because they cannot purchase the goods in question. Accordingly, the improvement of the promotional time limit can be carried out in the following manner. First, merchants should set the expected sales volume according to the expected response. Second, a fixed start time should be set, and a timely notice of the time of buying should be given to the consumer in order to avoid the loss of customers. Third, a promotional buying time should be set so that consumers can receive information and consider the potential purchase rationally on the basis of the same acquisition process of perceived pleasure.

6.1.6. Promotional Program Setting

In order to enhance their popularity and gain more exposure, businesses will ask consumers to complete some tasks during the live commerce event in order to qualify to purchase of the item being advertised. Too many routines and tedious steps in this process will reduce consumers' purchasing motivation. It is suggested that direct broadcast promotion procedures be simplified. Businesses should establish the connection between consumers and brands by designing reasonable promotional programs to ensure the consumers' purchasing enthusiasm [67], so as to increase the likelihood that the customer will return to make future purchases, while at the same time addressing the concerns of consumer groups.

6.2. Research Limitations

Despite our new findings and recommendations for the sustainability of live commerce, this article has some limitations. We used online questionnaires, and the arbitrary attitudes of some respondents may have a certain influence on the research results. In the future, we might consider adding offline filling channels in order to help researchers understand consumers' intuitive feelings about e-commerce live broadcasts in person and thereby draw more comprehensive research conclusions. In addition, there may be limitations in the selection of external stimulus influencing factors. In the future, we will try to consider more factors.

Author Contributions: Z.Z. presented the idea. N.Z. wrote the first draft of this paper. J.W. revised the first draft. Z.Z. was involved in the literature review section and data collection along with N.Z. and J.W. Data analysis was done by the joint effort of N.Z. and J.W. All authors have read and agreed to the published version of the manuscript.

Funding: This research was funded by the Program for the Philosophy and Social Sciences Research of Higher Learning Institutions of Shanxi, grant number 2020W013 and the educational reform Innovation Programs of Higher Education Institutions in Shanxi, grant number J2021041.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Not applicable.

Conflicts of Interest: The authors declare that they have no conflict of interest.

Appendix A

Table A1. Questionnaire item.

Latent Variables	Serial Number	Measuring Item
	AC1	When watching the live commerce, I think the anchors will respond to my questions in a timely manner.
_	AC2	The communication and interaction of the anchors made me feel that the live commerce marketing activity was very valuable.
Anchor Characteristics — (AC) [68,69]	AC3	I think the product information recommended by anchors for me is authentic and credible.
	AC4	Through the direct experience of the anchor to the product, I deepened my understanding of the product.
	AC5	I think anchors are familiar with the product and have professional knowledge, so they can explain clear and accurate activity information.
	OC1	I take the initiative to read the feedback of the online comments in the studio.
Online	OC2	I choose to buy items with a high number of real-time reviews.
(OC)	OC3	I tend to be drawn to extreme (positive or negative) real-time comments.
[70–72]	OC4	I can share my shopping tips with other customers through real-time online reviews.
	OC5	The content of real-time product reviews has an impact on my purchase decision.
	LSQ1	The timely and accurate processing of orders on e-commerce platforms encourages me to snap up goods.
	LSQ2	Merchants offer value-added services such as "seven-day no-excuses returns" and "instant refunds" to encourage me to snap up goods.
Logistics Service Quality (LSQ) [73]	LSQ3	If the logistics company cooperating with the merchant can promise to deliver the goods on the scheduled time, I will consider snapping up the goods.
_	LSQ4	When the distribution range of logistics outlets meets my needs, I will consider buying goods.
	LSQ5	When the e-commerce platform can timely deal with my return or replacement requirements, I will consider buying goods.
	PII1	I will always pay attention to the promotional prices and coupon information released in the live commerce.
Promotional; Incentive Information	PII2	When buying goods, I like to buy promotional goods in the broadcast room.
(PII)	PII3	Livestreaming promotions allow me to get a more reasonable consumer price.
[±0,/±]	PII4	The price discount in the broadcast room makes me feel very generous.
	PII5	The buying in the studio makes me feel more affordable.
	PTL1	I feel that the sale time of the broadcast room is usually relatively short.
Promotion Time Limit	PTL2	I feel like I have less time to decide if I want to buy something that's on sale live.
(PTL)	PTL3	I feel like I'm snapping up sales right before the deadline.
[/5]	PTL4	The closer I get to the end of a limited-time sale, the time pressure pushes me to buy as quickly as possible.
	PT1	I think e-commerce live commerce platforms are trustworthy.
	PT2	I think the promises made by e-commerce platforms are reliable.
(PT) [76,77]	PT3	If there is a problem with the product, I believe the merchant will try his best to provide me with a solution.
	PT4	I think e-commerce platforms can provide enough commodity information and quality services.
	PV1	I think the products recommended in the e-commerce live commerce just meet my needs.
Perceived Value	PV2	The sales promotion makes me feel economical.
(PV) [78]	PV3	When I participate in a sale, I feel good about getting a good deal.
	PV4	I feel it's very cost-effective to snap up sales.
	IBB1	When I watch the livestream of e-commerce, I often buy something that I didn't intend to buy.
Impulse Buying Behavior (IBB)	IBB2	When I watch the live commerce, I often find some goods I want to buy that are not in my plan.
[79] —	IBB3	In the flash sale, I will have a strong desire to buy goods.
	IBB4	I have a sudden urge to buy something during a flash sale.
-	IBB5	After reading the preferential content of the product, I have a great possibility to buy.

References

- 1. Gong, X.X.; Ye, Z.L.; Liu, K.; Wu, N. The effects of live platform exterior design on sustainable impulse buying: Exploring the mechanisms of self-efficacy and psychological ownership. *Sustainability* **2020**, *12*, 2406. [CrossRef]
- Geng, X.R.; Hao, Z.R. E-commerce direct marketing system architecture from the perspective of retail digital transformation. J. Commer. Econ. 2021, 13, 79–82.
- Xia, S. Research on problems and countermeasures of e-commerce direct broadcast with goods marketing model. Log. Eng. Manag. 2021, 43, 76–79.
- Huang, M.X.; Ye, Y.X.; Wang, W. The interaction effect of broadcaster and product type on consumers' purchase intention and behaviors in livestreaming shopping. *Nankai Bus. Rev.* 2021. Available online: https://kns.cnki.net/kcms/detail/12.1288.F.2021 0915.0954.002.html (accessed on 5 January 2022).
- Yan, X.X.; Dong, Y.H.; Zhang, M.M.; Qiao, J. Research on the impact of live broadcasting on consumers' buying behavior intermediate by perceived value. *Price Theor. Pract. Mag. House* 2021, 6, 137–140.
- Liu, Y.; Hong, X.L.; Feng, H. Website attributes in urging online impulse purchase: An empirical investigation on consumer perceptions. *Decis. Support. Syst.* 2013, 55, 829–837. [CrossRef]
- Zhen, Q.D. Research on the influence of live streaming goods interactivity on consumers' cross-purchase intention. *Pop. Stan.* 2021, 11, 140–142.
- Zhang, Y.J. Defects of traditional marketing model and new consumer experience requirements in the context of "new retail". J. Financ. Res. 2018, 2, 82–86. [CrossRef]
- 9. Sun, Y. Analysis of impulsive buying behavior in live broadcast scenarios. Educ. Reform. Dev. 2020, 2, 147–152. [CrossRef]
- Guan, P.C.; Xiang, Y. On the guiding effect of hunger marketing strategy on consumer behavior. *J. Commer. Econ.* 2016, *4*, 53–55.
 Feng, H.; Fu, Q.; Zhang, L. How to launch a new durable good: A signaling rationale for hunger marketing. *Int. J. Ind. Organ.* 2020, *70*, 102621. [CrossRef]
- 12. Zhang, Z.H.; Qiu, Y.S. Analysis on the marketing model of Douyin Live Broadcast with goods hungry. Mark. Mod. 2021, 3, 44-46.
- 13. Yu, L.I.; Zhang, J.Z. A two-period pricing model with hunger marketing strategy. J. Model. Manag. 2018, 13, 81-100. [CrossRef]
- Tolety, R.K. Hunger Marketing, Flash sale and cost optimization strategy of Xiaomi inc.—Setting a new trend through etailing. Int. J. Res. Financ. Mark. 2017, 7, 182–185.
- 15. Chen, Q.Y. The Analysis of application of huger marketing in China's mobile phone industry. Sci. Innov. 2016, 4, 117–121. [CrossRef]
- Liu, Y.; Li, Q.; Yin, M. Research on the influence of webcast shopping features on consumer buying behavior. Soft. Sci. 2020, 34, 108–114.
- 17. Rook, D.W.; Fisher, R.J. Normative influences on impulsive buying behavior. J. Consum. Res. 1995, 22, 305–313. [CrossRef]
- Wells, J.D.; Parboteeah, V.; Valacich, J.S. Online Impulse Buying: Understanding the interplay between consumer impulsiveness and website quality. J. Assoc. Inf. Syst. 2011, 12, 32–56. [CrossRef]
- Umair, A.; Peng, H.; Muhammad, K.K.; Chen, Y.; Zubair, A. Factors affecting online impulse buying: Evidence from Chinese social commerce environment. *Sustainability* 2018, 10, 352.
- 20. Lu, X.M.; Xue, Y.J. Analysis on the influencing factors of impulsive buying behavior. J. Bus. Res. 2011, 7, 50-52.
- Jiang, C.; Zhao, H.X. B2C online store image, consumer perception and purchasing behavior. *Res. Financ. Econ. Issues* 2013, 10, 116–122.
- Naratama, I.M.; Rahyuda, I.K. Study of influence of store environment on impulse buying tendency. Russ. J. Agric. Socio-Econ. Sci. 2018, 82, 161–170. [CrossRef]
- Carunia, M.F.; Maria, F. The effects of sales promotion, attractiveness of internet advertising, and website quality on impulse buying of consumers of Tokopedia in Indonesia. Int. J. Manag. Ent. Dev. 2021, 20, 34–48.
- Li, C.X.; Wang, Y.T.; Lv, X.Y.; Li, H. To buy or not to buy? The effect of time scarcity and travel experience on tourists' impulse buying. Ann. Tour. Res. 2021, 86, 103083. [CrossRef]
- Zhao, J.D.; Huang, J.S.; Su, S. The effects of trust on consumers' continuous purchase intentions in C2C social commerce: A trust transfer perspective. J. Retail. Consum. Serv. 2019, 50, 42–49. [CrossRef]
- Jacoby, J. Stimulus-Organism-Response reconsidered: An evolutionary step in modeling (consumer) behavior. J. Consum. Psychol. 2002, 12, 51–57. [CrossRef]
- 27. Jing, W.; Sung, K.Y. A Study on the effect of service recovery through mediating customer forgiveness on customer behavior intention of online shopping mall—Based on the SOR model. J. Korean Soc. Qlty Manag. 2019, 47, 615–630.
- Hu, H.Q.; Yan, J.Y.; Xu, L. Information richness, purchasing cost, channel mode and consumer purchasing behavior. *Manag. Rev.* 2012, 24, 80–88.
- 29. Lin, Z.H. The Influence of network broadcast information characteristics on impulsive buying: Based on the mediating role of flow experience. J. Commer. Econ. 2021, 14, 75–78.
- 30. Zhao, D.W.; Feng, J.X. The impact of characteristics of e-ommerce anchor as a key opinion leader on consumers' purchase intention. *Commer. Res.* **2021**, *4*, 1–9.
- Zhang, Z.; Zhang, G.Q. Research on the Influences of anchors on webcast sales—Empirical test based on consumer questionnaire. Logist. Sci.-Tech. 2021, 44, 53–56.
- 32. Zhao, B.G.; Wang, Y.F. The influence of e-commerce anchor characteristics on consumers' purchase intention. Comm. Res. 2021, 1, 1–6.

- Yan, X.; Han, X.H. Optimal pricing and remanufacturing entry strategies of manufacturers in the presence of online reviews. Ann. Oper. Res. 2021, 1, 1–34. [CrossRef]
- 34. Jain, P.K.; Pamula, R.; Srivastava, G. A systematic literature review on machine learning applications for consumer sentiment analysis using online reviews. *Comput. Sci. Rev.* 2021, 41, 100413. [CrossRef]
- 35. Prasad, V.; Anja, L. The effect of individual online reviews on purchase likelihood. Market. Sci. 2021, 40, 708–730. [CrossRef]
- Wu, X.I.; Liao, H.C. Learning judgment benchmarks of customers from online reviews. Or Spectr. 2021, 43, 1125–1157. [CrossRef]
 Qi, Z.C. Analysis of the effect of online reviews on consumer intention. Front. Econ. Manag. 2021, 2, 90–98. [CrossRef]
- Shen, C.C.; Chang, Y.R. Exploring the relationship among online review, perceived barriers, customer experience, and purchase intention of online booking consumers customer value as a mediator. J. Tour. Hosp. 2021, 10, 1–14.
- 39. Liu, Z.Y.; Yin, L.J.; Yuan, L.N. Research on the impact of logistics service factors on consumers' online shopping intention based on structural equation modeling. J. Math. Pract. Theory 2019, 49, 34–42.
- Liang, W.; Zhang, W. Study on influence of logistics service involvement and perceived service quality on consumer psychological contract. J. U. Financ. Econ. 2016, 29, 68–74.
- Liu, L. Analysis of the impact of personalized logistics services on consumers' repeated consumption intentions under the B2C model. J. Comm. Econ. 2021, 1, 115–118.
- Yan, X.X.; Qiao, J.; Liu, X.M. Research on refined logistics service of fresh products under the background of new retail—Based on customer online repurchase willingness analysis. *Price Theory Pract.* 2020, 12, 111–114.
- Yang, X.C.; Zhang, L. Research on the long-term effects of e-commerce promotions—An empirical study of t-mall consumers. J. Technol. Econ. Mana. 2020, 6, 36–41.
- Zhang, H.; Li, L.; Zhu, X.Z.; He, X. Online shopping platform optimal price promotion strategy: Price discount or cash coupon? Chinese J. Mana. Sci. 2021. [CrossRef]
- Prasilowati, S.L.; Suyanto, S.; Safitri, J.; Wardani, M.K. The impact of service quality on customer satisfaction: The role of price. J. Asian Financ. Econ. Bus. 2021, 8, 451–455.
- 46. Chen, K.W. Research on influence of online promotion type on consumer purchase intention. Mark. Wkly. 2021, 34, 82–85.
- Cui, Y.Y.; Kim, S.S.; Kim, J.K. Impact of preciseness of price presentation on the magnitude of compromise and decoy effects. J. Bus. Res. 2021, 132, 641–652.
- Ye, J.; Hu, C.L. The impact of live streaming time-limited promotion and anchor trust on consumers' clothing purchasing behavior. J. Silk 2021, 58, 57–67.
- Lu, C.B.; Huang, C.F. The influence mechanism of Time pressure with need for cognitive closure on undoing regrets arising in decision-making of sales promotion. J. Bus. Manag. 2014, 36, 145–158.
- 50. Hong, R. Online flash sales product pricing strategy under time pressure. J. Wld. Sci. Res. 2021, 7, 376–382.
- Rousseau, D.M.; Sitkin, S.B.; Burt, R.S.; Camerer, C. Introduction to special topic forum: Not so different after all: A cross-discipline view of trust. Acad. Manag. Rev. 1998, 23, 393–404. [CrossRef]
- Anas, H.; Tony, W.; Asmai, I.; Putra, E.C. Consumer trust as the antecedent of online consumer purchase decision. *China Inform.* 2021, 12, 145.
- Fan, M.; Ammah, V.; Dakhan, S.A.; Liu, R.; Mingle, M.N.; Pu, Z.J. Critical factors of reacquainting consumer trust in e-commerce. J. Asia Fbt. Econ. Bus. 2021, 8, 561–573.
- 54. Feng, J.; Lu, M. The empirical research on impulse buying intention of live marketing in mobile internet. *Era. Soft Sci.* 2020, 34, 128–133.
- Hossain, U.M.U.; Ishraq, J.; Hussam, A.H.; Abdul, H.A.B.; Abdul, L.A.S. Does quality stimulate customer satisfaction where perceived value mediates and the usage of social media moderates? *Heliyon* 2020, *6*, E05710.
- Zeithaml, V.A. Consumer perceptions of price, quality, and value: A means-end model and synthesis of evidence. J. Mark. 1988, 52, 2–22. [CrossRef]
- 57. Bettman, J.R.; Luce, M.F.; Payne, J.W. Constructive consumer choice processes. J. Consum. Res. 1998, 25, 187–217. [CrossRef]
- Wang, G.C.; Liu, W.J.; Wang, X.F. A situational study on the role of promotion purchase restriction under different promotion types: From the perspective of framing effect theory. *Nankai. Bus. Rev. Int.* 2021. Available online: http://kns.cnki.net/kcms/ detail/12.1288.F.20210629.1600.002.html (accessed on 5 January 2022).
- 59. Han, X.Y.; Xu, Z.L. Impacts of e-commerce anchor attributes on consumers' willingness to buy online: Research based on the grounded theory. *Fgn. Econ. Manag.* **2020**, *42*, 62–75.
- Meng, Y.; Xie, H.T. Research on Preheating marketing of double—11 e-commerce platforms under entertainment fatigue. Log. Eng. Manag. 2020, 42, 98–100.
- Kao, C.K.; Chen, Z.N.; Lin, P.J. Effect of online shopping credit evaluation on customer purchase risk: Logistics service as a mediator. Am. J. Theor. Appl. Bus. 2021, 7, 16–29. [CrossRef]
- Narayanaswamy, R.; Heiens, R.A. Online sales promotion in hedonic versus utilitarian product categories. Int. J. Electron. Mark. Retail. 2018, 9, 77–88. [CrossRef]
- Mao, Z.F.; Yu, S.Q.; Li, H.; Yuan, Y.Y. The time-limited discount pricing decision considering consumer demand transfer. Syst. Eng. Theory Pract. 2021, 41, 3232–3248.
- 64. Yang, N. The influence mechanism of web celebrity live broadcast on consumers' brand attitude. J. Cent. Univ. Financ. Econ. 2021, 2, 118–128.

- Liu, Y.; Gan, W.X.; Zhang, Q. Decision-making mechanism of online retailer based on additional online comments of consumers. J. Retail. Consum. Serv. 2020, 59, 102389. [CrossRef]
- Jaiswal, B.; Saba, N.U. An Analytical study of behavioural response to sales promotion offers: With special reference to apparels. Int. J. Manag. IT Eng. 2019, 9, 79–88.
- 67. Jones, J.M. Non-Price sales promotion impact and moderator role. J. Promot. Manag. 2019, 25, 861–889. [CrossRef]
- Harikesh, S.N.; Puneet, M.; Tulikaa, B. Asymmetric social interactions in physician prescription behavior: The Role of opinion leaders. J. Mark. Res. 2010, 47, 883–895.
- Echchakoui, S. An analytical model that links customer-perceived value and competitive strategies. J. Mark. Anal. 2018, 6, 138–149. [CrossRef]
- 70. Chiou, C. Should a company have message boards on its websites. J. Interact. Mark. 2003, 17, 50-61. [CrossRef]
- Park, D.H.; Lee, J.; Han, I. The effect of on-line consumer reviews on consumer purchasing intention: The moderating role of involvement. Int. J. Electron. Comm. 2007, 11, 125–148. [CrossRef]
- Zhao, H.X.; Cai, Z.H.; He, S. The relationship between online merchandise displaying, online interaction and impulsive buying based on virtual tactility. *Chin. J. Manag.* 2014, 11, 133–141.
- 73. Mentzer, J.T.; Flint, D.J.; Hult, G.T.M. Logistics service quality as a segment-customized process. J. Mark. 2001, 65, 82–104. [CrossRef]
- 74. Payne, J.; Bettman, J.; Luce, M.F. When time is money decision behavior under opportunity cost time pressure. *Organ. Behav. Hum. Decis. Processes* **1996**, *66*, 131–152. [CrossRef]
- Lu, C.B.; Qin, Q.X.; Lin, Y.Y. Cognitive mechanism in consumer decision-making under conditions of deceptive promotion: An Empirical research based on time pressure and overconfidence. *Nankai Bus. Rev.* 2013, 16, 92–103.
- Alalwan, A.A.; Dwivedi, Y.K.; Rana, N.P. Factors influencing adoption of mobile banking by Jordanian bank customers: Extending UTAUT2 with trust. Int. J. Inform. Manag. 2017, 37, 99–110. [CrossRef]
- Huang, S.H.; Xiao, J.C.; Jin, Y.N. Study on the influencing factors of consumers' continuous buying intention on social e-commerce platform based on S-O-R theory. Soft Sci. 2020, 34, 115–121.
- Liu, J.; Zou, Y.J.; Liu, Z.X. Analysis of influencing Factors of Consumers' purchase intention in E-commerce live Broadcast based on SEM Model. *Stat. Decis.* 2021, 37, 94–97.
- Adelaar, T.; Chang, S.; Lancendorfer, K.M.; Lee, B.; Morimoto, M. Effects of media formats on emotions and impulse buying intent. J. Int. Technol. 2003, 18, 247–266. [CrossRef]





Article Challenges and Perspectives in Innovative Projects Focused on Sustainable Industry 4.0—A Case Study on Polish Project Teams

Marzena Podgórska

Faculty of Organization and Management, Department of Management, Silesian University of Technology, Akademicka 2A, 44-100 Gliwice, Poland; marzena.podgorska@polsl.pl

Abstract: Contemporary project teams are increasingly used to solve problems that are at the crossroads of many disciplines and areas dedicated to Industry 4.0, which is a watershed in the implementation of Sustainable Development Goals (SDGs). Industry 4.0 can serve as a platform for the alignment of SDGs with the ongoing digital transformation. This involves specific challenges for teams, but also allows perspectives that may create innovative and high-quality results. In order to meet these challenges while taking advantage of the opportunities offered by interdisciplinary cooperation, project teams, including the team leader, should have specific competencies. With this in mind, the aim of this article is to identify the challenges and perspectives related to working in interdisciplinary Sustainable Industry 4.0 project teams and to define the competencies necessary to act as a member and leader of these teams. Implementation of this aim will be possible by answering two research questions: (1) What requirements and opportunities are involved with interdisciplinary work amongst members of Sustainable Industry 4.0 project teams; and (2) What are the competencies necessary of members and leaders of such teams to meet these requirements and take advantage of the opportunities for such cooperation? An exploratory case study was conducted among members of interdisciplinary project teams at one of the leading technical universities in Poland. Qualitative data were obtained from many sources: interviews, internal documentation of analyzed projects and managerial notes. The obtained results allow us to state that the most important challenges and perspectives related to the work of interdisciplinary Sustainable Industry 4.0 teams include coordination of individual parts of the project, integrative leadership, establishing a common language, broad views on the issues raised and building a team consisting of specialists with the required competencies. The competencies of the project team that are important for working in the analyzed environment include strategic perspective, communication skills and persuasion, while for leaders, competencies must include the ability to coordinate work, resource management, empowering and motivation.

Keywords: Sustainable Industry 4.0; innovative projects; interdisciplinarity; project team competencies; leadership

1. Introduction

Over the past centuries, as a result of industrial revolutions, society has experienced significant changes and technological progress [1]. Industry 1.0 was characterized by introducing machines to industrial processes [2]. Industry 2.0 was related to the concept of multi-scale production, reduction of costs and lead time, and the introduction of technical and organizational innovations [3]. The next age, Industry 3.0, was associated with the development of technological information systems [4]. Currently, the fourth stage of industrialization (Industry 4.0) affects the creation of industrial value in the most-developed countries. Industry 4.0 is treated as the beginning of a new era in which industry is becoming more and more "intelligent" thanks to the use of Internet of Things technology, predictive analysis and intensive data exchange [5,6]. Industry 4.0, as many researchers (e.g., ref. [7]) emphasize, also changes the way of performing and controlling tasks related to project

Citation: Podgórska, M. Challenges and Perspectives in Innovative Projects Focused on Sustainable Industry 4.0—A Case Study on Polish Project Teams. *Sustainability* **2022**, *14*, 5334. https://doi.org/10.3390/ su14095334

Academic Editor: Fabrizio D'Ascenzo

Received: 2 April 2022 Accepted: 27 April 2022 Published: 28 April 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the author. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). management as projects become more complex, innovative, and implemented at the crossroads of many areas of the economy. Industry 4.0 is also a sociotechnical concept that can support the entire society and industry in achieving sustainable development [8–11]; hence, this stage of the revolution can be called Sustainable Industry 4.0. Considering the above, it becomes necessary to more deeply understand the challenges and perspectives that project team management brings with it in an environment that is strongly connected to Sustainable Industry 4.0 and to undertake an analysis of the required competencies of members and leaders of such project teams.

Sustainable Industry 4.0 is the new path of organization and control for systems with full added value, and its overriding goal is to satisfy the individual needs of customers with the costs of mass production. On this account, work in an economy based on Sustainable Industry 4.0 means behaviors and skills [12] required from employees, including project team members, must enable them to meet the challenges of Sustainable Industry 4.0. The need for a new approach to human resources management in the era of Industry 4.0 is emphasized by many researchers (e.g., ref. [13]), who point out that Sustainable Industry 4.0 may create a need for a management paradigm shift. Nevertheless, as researchers (e.g., ref. [14,15]) emphasize, analyses in the context of Sustainable Industry 4.0 mostly refer to the new technologies driving this revolution, and the literature on the human factor in the area of Industry 4.0 is still narrow and rare, which necessitates more research in this area.

Sustainable Industry 4.0 is an excellent opportunity to take a real step towards better performance and strategic value creation by organizations [16]. For this reason, more and more references related to Sustainable Industry 4.0 can be found in various citation databases today [17]. These considerations focus on the concept of Sustainable Industry 4.0 in the context of project management. Ref. [18] points out, the role of project management is crucial for the success of Sustainable Industry 4.0 and vice versa. Therefore, many researchers emphasize the need for further work in this area to provide a more in-depth analysis of the use of project management during the 4th Industrial Revolution. Ref. [19], on the basis of their analysis of the main research areas in project management and activities covered by the 4th Industrial Revolution, identified human resource management as one of the common themes for future analysis in this area. The relevance of this topic from the point of view of the discussed issues is also indicated by other researchers. For example, ref. [20,21] emphasize that Sustainable Industry 4.0 influences project management and makes it necessary to change the way projects are implemented and the interaction between team members. It also forces project teams to develop new competencies necessary for activities within the 4th Industrial Revolution. Similarly, ref. [22] points out that project managers and teams will require higher soft and hard skills than needed in the past, and more autonomy in handling Sustainable Industry 4.0 projects. Ref. [23] adds that Sustainable Industry 4.0 challenges the project managers in different ways.

Moreover, it should be noted that 4.0 projects are characterized by higher complexity and uncertainty, which requires a flexible approach to solved problems and interdisciplinary work. Ref. [7] draws attention to that, also pointing out that the two main theoretical attributes of project management—uncertainty and complexity—also play a significant role in Sustainable Industry 4.0 projects because they shape and define the necessary features to support the workload and are associated with increasing project complexity. Therefore, as [24] indicates, project teams in Sustainable Industry 4.0 must be self-organizing, interdisciplinary, real-time and autonomously optimized. This is also emphasized by the authors of [25], who state that project management in Sustainable Industry 4.0 is characterized by digitization, virtualization, transnationality, professionalization, the transition from Waterfall to Agile, focus on the project-organization relationship and organization maturity in project management. The complex problems that characterize Sustainable Industry 4.0 projects, as many authors emphasize (e.g., ref. [24]) and is also emphasized above, require interdisciplinary cooperation. However, based on analysis of the literature, it can be concluded that analyses related to interdisciplinary cooperation of teams operating in the 4.0 environment are rare.

The motivation for this article was to identify ways in which the changes accompanying the 4th Industrial Revolution affect the new requirements for workers from many industries and sectors, and the aim of this article is to develop a theoretical and empirical understanding of how the changes caused by the 4th Industrial Revolution affect the competencies of project teams. Although research and practice show that in order to achieve the sustainable implementation of technologies supporting Sustainable Industry 4.0, a new approach to human resource management will be necessary, including, in particular, the acquisition and improvement of appropriate competencies by employees, including members of project teams. However, streams of literature connecting management projects and human resource management have not been largely integrated into Sustainable Industry 4.0 literature. The absence should be seen as a problem, because research of various disciplines suggests (e.g., ref. [2]), that, in the face of future trends of the 4th Industrial Revolution, it is necessary to develop a new management paradigm, which will be largely based on the interdisciplinary cooperation of competent teams, and their lack is seen as the main threat to economic growth and should be a top priority in the agendas of enterprises and policy makers. Further, management literature indicates that the development of employee competencies is necessary in rapidly changing industries and sectors, as is the case in industries and sectors of Sustainable Industry 4.0, so that companies can adapt to changing conditions [26]. Taking into account the above, a case study of project team members at one of the leading technical universities in Poland between February 2019 and September 2020 was conducted in this article. The obtained results fill the research gap concerning the requirements of Sustainable Industry 4.0 for members of project teams, and are some of the first to indicate the necessary competencies in this area. What is more, the inclusion of interdisciplinary project teams in the analysis of the interdisciplinary context makes these considerations original, as they are overlooked or considered in a very narrow scope and can provide a basis for shaping and improving the selection process of interdisciplinary teams in the area of Sustainable Industry 4.0.

2. Literature Review

2.1. Sustainable Industry 4.0

Industry 4.0 is treated as the 4th Industrial Revolution [27]. It includes a number of new technologies influencing new digital value chain creation as well as the digitization and automation of the production environment. The 4th Industrial Revolution is based on development of intelligent systems that enable system monitoring and decision making in real time [28]. A new paradigm appears here, focused on modern enterprise management based on new technologies enabling integrated optimization of processes [7]. This paradigm results from the combination of production systems, which on the one hand are vertically connected to the company's business processes, and, on the other hand, horizontally connected to other points in the value chain in order to increase efficiency and response time to each request, with the customer being the main target [29–31]. It can therefore be concluded that the transition to Sustainable Industry 4.0 and the development of cyber–physical systems introduces technical, organizational and human changes in various organizational layers of enterprises [32,33].

Although the 3rd Industrial Revolution was marked by simple digitization, the 4th Industrial Revolution has an inexorable shift in this simple digitization, where the rush of innovation comes from many combinations of technology in new forms. The 4th Industrial Revolution can be distinguished from others for three reasons: (1) Speed—the interconnectedness of the world increases at an exponential rate, and the emergence of new technologies generates further innovation. (2) Breadth and Depth—taking the 3rd Industrial Revolution as a reference, the 4th Industrial Revolutions integrates and expands various technologies. (3) Systemic Impact—entire systems are transformed within countries, companies, value creation networks or in society in general [24]. Therefore, despite the many advantages of Sustainable Industry 4.0, this new era of industry is also associated with a number of challenges for many organizations. Examples of such challenges are technology integration,
data security, organizational transformation [34], acquisition and implementation of appropriate information technologies, distribution of decisions, encouragement to experiment with new ideas in the workplace [35] and greater complexity of projects [36]. Consequently, these challenges are reflected in new requirements for employees, changing professional profiles of employees and employment structures. The change of professional profiles especially is characterized by a change in the required competencies of employees. Examples of such desirable competencies include abstract and analytical skills, openness to changes and knowledge in the field of data processing [37]). Much emphasis is also placed on active individual development.

In their research, ref. [38] divided professional profiles into two groups: 4.0 and non-4.0 profiles, and then assessed the differences between the skills of these profiles, dividing the skills into three groups: everyday execution, operational abilities and functional abilities. The operational skills consisted of hard and transversal skills. The everyday skills cluster included eight abilities (e.g., business acumen, project management, change management and problem solving). All of them were of a cross-sectional character, and many of them also had a methodological character (e.g., project management and change management). On the other hand, the functional cluster included 28 skills, both hard and cross-sectional. The results of these studies show that the average level of vertical skills, such as operational and functional, turn out to be the same for both 4.0 and non-4.0 profiles. Transversal skills, in turn, prove to be much more appropriate for 4.0 profiles. This indicates that there is an important soft component to manage and face the introduction of new technologies [39]. The importance of soft skills in the digital age is also emphasized in many other studies, e.g., ref. [40–42], where they are indicated as resistant to change and a vital new component of the workforce of the future, not just the key to differentiate people from machines.

Ref. [43] pointed out that the main skills that can best influence the context of Sustainable Industry 4.0 are based on three skill groups: cognitive skills, interpersonal skills and strategic skills. In the interpersonal leadership group, the most outstanding competencies are negotiation, persuasion and social perception. In the case of the strategic leadership group, the most significant competencies are vision, identification of key requirements and system and solution assessment. For the cognitive group, the important competencies are communicative skills, active listening, learning and critical thinking. Ref. [44], in turn, identified two sets of leadership skills necessary in Sustainable Industry 4.0: one related to human relationships and the other related to knowledge about the use of technology. The main leadership skills were listening, teamwork, stakeholder relations, team member relations, use of digital tools and ability to deal with change.

Moreover, ref. [38] proved that significantly important competencies for 4.0 professions are abilities such as business acumen, project management and change management, in which the difference between 4.0 and non-4.0 profiles is substantial, and which play a key role in the context of renewal.

2.2. Project Management in Sustainable Industry 4.0

In Sustainable Industry 4.0, a faster and more dynamic flow of information is expected. Working groups will cover very different areas of knowledge and will be varied. There will be development and use of more predictive and faster management tools. The use of Big Data tools will enable rapid flow of huge amounts of information with a wide spectrum Robotization of the industry will increase its efficiency, but under certain circumstances it can also remove the human factor [45]. All these factors will cause significant changes in the form of management, and thus also in project management [7].

The contribution of the human factor in project activities related to Industry 3.0 is visible and observed in all processes that make up project management. The tasks carried out within the project are defined here, timed and used as a project management and control instrument covering all activities ranging from initiation to completion of the project. Initiating, planning, implementing, monitoring and controlling the project, as well as the

processes related to closing the project, are subprocesses of the classic approach to project management [45] and are implemented in Industry 3.0 projects. In contrast, Sustainable Industry 4.0 projects are individual-customer driven. For this reason, order management applies to all areas, such as research and development, production, commissioning, delivery and recycling of manufactured products. It is also particularly noticeable in the production sector, which today is pioneering Sustainable Industry 4.0, and in which the human factor is gradually replaced by machines, and the subprocesses of applying, monitoring and controlling are fully automated. In this case, initiation and planning processes take on more importance than in the classic approach to project management, and identification of the human factor involved in project initiation and planning requires restructuring to include its functions during the implementation, monitoring and control of the project [25]. This is also emphasized by [46], who indicate that, unlike "classic" projects, sustainable Industry 4.0 projects are characterized by an unclear purpose and/or unclear solution, therefore an important aspect here is the phase of gathering and identifying information that is missing in the implementation of "classic" projects.

In addition, in the era of Sustainable Industry 4.0, new opportunities are the digitization of production using base, cyber–physical production systems This is why resources, such as all employees, products, resources and systems, must additionally be integrated as intelligent, self-organizing, interdisciplinary, real-time and autonomously optimized. This is also pointed out by [46], who emphasize that 4.0 projects are increasingly becoming interdisciplinary projects that combine various project grounds that can no longer be clearly separated from each other. Moreover, as the authors point out, management of 4.0 projects encounters barriers related to such issues as: insufficient change management, corporate culture inhibiting innovation, insufficient support and quality of decision making by managers, or the lack or inadequately planned project characteristics.

Project management in Sustainable Industry 4.0, through manifestation of its characteristic elements, will have a significant impact on the planning and implementation of projects in various areas of activity. For example, the new function of project management in Sustainable Industry 4.0 projects will be as follows: In the area of time management—realtime monitoring of project execution and eliminating gaps in progress reports; In the area of team management-generalize the use of virtual teams and collective intelligence; In the area of communication management-accelerate communication processes within projects, remove physical communication support and increase connectivity [7]. These changes can also be emphasized in the soft and hard skills of project managers and their team members. In Sustainable Industry 4.0, the soft skills of project managers will undergo a significant transformation mainly related to new ways of interacting with project stakeholders. There will be important soft skills such as [22]: communication skills, power, team management, management of unforeseen events and negotiating skills. Therefore, the project manager should build their authority on the basis of 360-degree knowledge of the project and the related field. In order for the team to operate effectively, communication should be carried out in a transparent and responsible manner.

2.3. Interdisciplinary Project Teams

Working in interdisciplinary project teams is becoming increasingly popular [47–50] due to their increasing use in the development of the new processes and products [51] required by Sustainable Industry 4.0. This also affects the composition of project teams, as team members should be highly qualified and have diverse competencies. Therefore, a very important issue, already at the stage of defining the resources necessary for project implementation, is the proper selection of employees for the project team [52].

The work of project teams in an environment such as Sustainable Industry 4.0—which, therefore, is subject to constant innovation—requires team members not only to develop technical knowledge and hard skills, but also necessitates changes in behavior and development of soft skills, among which interdisciplinarity is emphasized [53]. Ref. [54], analyzing the new skills of teams necessary for Sustainable Industry 4.0, indicates that the most

frequently mentioned skills include interdisciplinary thinking, problem solving, flexibility and creativity, and adds that the need for interdisciplinary thinking is a consequence of the high complexity of knowledge (tacit and explicit) in companies. In this context, ref. [55] also emphasizes education of team members must be adapted to new teaching and learning models that focus on developing interdisciplinary competencies and increasing employees' ability to solve problems and face the challenges resulting from the 4th Industrial Revolution. This is also emphasized by [56], pointing out that it is important for organizations to remodel the learning process and develop interdisciplinary competencies that will increase the capacity of team members in problem solving, creativity, and innovation.

Moreover, in the case of 4.0 projects that are realized in a dynamic environment that is constantly changing and for which the goals cannot be clearly defined at the beginning, agile approaches to project management become important [57]. Agile project management is characterized by an iterative and adaptive approach, consisting of short, customer-oriented feedback loops, formal and informal communication, and enabling self-organization in interdisciplinary teams [58]. In addition, the agile approach allows project teams to respond to emerging needs in a timely manner [59–61], which is extremely important in the context of the 4th Industrial Revolution, which is aimed at meeting rapidly changing preferences, customers and available technologies.

The interdisciplinary teams behind [62] are those in which the team uses the information, data, techniques, tools, perspectives, concepts and/or theories available within different disciplines or groups of expertise to solve an issue that goes beyond the scope of one discipline or area of research. Of course, the project team itself has, by definition, distributed roles and expertise [63], but in the case of interdisciplinary project teams, we deal with a different type of skills and highly specialized knowledge from several disciplines, professions, organizations or nations. Moreover, it should be noted that degrees of distributed knowledge and expertise likely vary according to project team type; therefore, when considering an interdisciplinary project team, we are dealing with integrated project teams that have multiple varied goals that must be clarified across their heterogeneous membership and multiple work cycles. Furthermore, such a team must integrate extremely large amounts of information, as processes and routines must be adapted or created for managing and executing projects [63].

The main challenge in interdisciplinary team cooperation is overcoming the disciplinary gap, but often, also, the form of cooperation itself. Interdisciplinary project teams vary in configuration in terms of how they perform their work. An interdisciplinary project team may include members who each have expertise in a different area—i.e., the diversity of disciplines is "interpersonal" at the team level [64]—or include members who each have experience in all the different aspects present in the team—i.e., the interdisciplinarity is "interpersonal". In addition, interdisciplinary teams may be dispersed or partially dispersed, come from one or more organizations, or cooperate within one organization or with many external institutions [65].

2.4. Challenges of Interdisciplinary Work

Interdisciplinary teams try to integrate their diverse knowledge so that their collective knowledge is greater than the sum of the knowledge of individual members [66,67]. This provides better-quality problem solving and a broader view of the analyzed problem [68]. On the other hand, the researchers in [69–71] emphasize that the integration of knowledge of team members, especially in situations novel to Industry 4.0 projects, is a challenge. In this context, ref. [72] believes that interdisciplinary projects can be more complex and require more bureaucracy or other efforts to organize and coordinate individual contributions. Others researchers, e.g., ref. [72,73], emphasize members of interdisciplinary teams speak different "languages", which influences the direct cooperation in these teams, creates reasons for conflict and makes them tend to share tasks more strongly. In addition, the literature, e.g., ref. [51,74], points out the difficulty of integrating the various knowledge bases of interdisciplinary teams, emphasizing that closer interaction between team

members is necessary. Ref. [75] stresses that "interdisciplinarity requires more than just complementarity", it requires "new types of empirical approaches" as well as "integrated analyses". Similarly, ref. [76] points to the importance of close communication and cooperation between team members in interdisciplinary projects. They argue that frequent direct interactions facilitate knowledge integration but also contribute to building trust and common understanding of the research problem, which improves the quality of results and reduces coordination costs over time. In this context, ref. [77] identified five practices that project teams can use to integrate their knowledge to cocreate a solution. Moreover, the researchers confirm the key role of active communication in the process of knowledge transformation and integration, especially when the tasks are of innovative nature, i.e., as it is the case with projects implemented at the intersection of different disciplines and in the area of Sustainable Industry 4.0.

Refs. [78,79] emphasize that the more diverse the disciplines are, as presented by team members with little common experience, the greater the differences between their tasks may be. Each team member, who represents a different specialization of knowledge, comes to the team with a different "world of thought" and understands problems, critical elements and problem-solving steps differently than other team members [77,80]. Therefore, it is important that each interdisciplinary team includes leaders who, through their integration skills, will support interaction and discussion between team members to generate the trust and common understanding necessary for knowledge integration [81].

Ref. [82] treats team-leader behavior as the key input that shapes the knowledge development process, and suggests that this behavior is key to ensuring that the unique knowledge of team members can be used by the entire interdisciplinary team. Ref. [83] points out that coordination in interdisciplinary teams can be improved if the team uses members' individual leadership strengths. This is also emphasized by [84], emphasizing that interdisciplinary projects can increase the probability of scientific breakthroughs by bringing together ideas, tools and specialists from different research and practice areas. Moreover, the importance of leadership is also emphasized for projects in Sustainable Industry 4.0 and referred to as Leadership 4.0 [85,86]. Ref. [87] emphasizes that implementing Leadership 4.0 requires investment and openness to cultural change. Leadership 4.0 styles must be open, leading to a learning- and innovation-oriented culture focused on improving and integrating knowledge and thinking outside the box [88]. Ref. [89] indicates that in order to improve coordination in interdisciplinary teams, it is appropriate to adopt a common leadership structure focused on the team. Ref. [84] indicates that an integrative approach to team leadership is necessary for successful interdisciplinary projects. The goal of integrative leadership is to achieve team creativity through a process of integrating the experiences, perspectives and ideas of both team leaders and their members [90,91]. Leadership is, therefore, another element that influences the effective cooperation of an interdisciplinary Sustainable Industry 4.0 project team.

According to the analysis, the challenges related to the work of interdisciplinary Sustainable Industry 4.0 project teams involve many aspects. The most frequently stressed is the need to agree on a "common language", close coordination of individual parts of the project, and communication and cooperation between team members. Moreover, leadership and selection of a team with specific competencies is indicated as an important factor for the success of such projects. However, from the results of the conducted analyses and what has already been emphasized, it is difficult to find in the literature research that would treat the issue of interdisciplinary competencies of project teams solving problems for Sustainable Industry 4.0 as a whole, and to indicate the competencies of members and leaders of such teams. Therefore, the research questions referred to in this article are: (1) What requirements and opportunities are set before members of project teams undertaking interdisciplinary Sustainable Industry 4.0 work? and, therefore (2) What are the competencies of members of such teams and their leaders to meet these requirements and take advantage of the opportunities for such cooperation?

3. Materials and Methods

Due to the lack of previous systematic research on the competencies of members of project teams carrying out interdisciplinary projects in Sustainable Industry 4.0, we decided to conduct an exploratory case study among members of interdisciplinary teams at one of the leading technical universities in Poland. The case study is a valuable research project in exploratory research because it is possible to obtain extensive data that allows researchers to investigate specific managerial problems in existing application areas, which extends the existing state of research [92,93]. In addition, case studies have been successfully used elsewhere to identify necessary competencies [94], including studies dedicated to Sustainable Industry 4.0 [95].

Qualitative data came from many sources: interviews, internal documentation on analyzed projects and reflective notes of project managers. In order to establish the reliability of the analysis, the sources used were triangulated [96,97].

The study started with finding a representative sample [92,93]. Subsequently, the interviews were conducted, which were partially structured and consisted of guidelines representing the research questions posed in the article. The interviews were conducted in person. Each of the interviews lasted between 30 and 60 min, depending on the respondent's statements. With the consent of the interviewees, the interviews were recorded and then transcribed from audio files to text. In addition, internal documentation of the analyzed projects provided by the technical university was used as a secondary source for triangulation [92,98,99]. All the collected data were arranged in a spreadsheet, and qualitative analysis of their content was performed [100]. During qualitative content analysis, requirements and perspectives were defined in an inductive way according to the description of [101,102] and adjusted to existing research, enabling the emergence of new knowledge [103]. The coding process started by "plunging" into the data by repeatedly reading interview transcripts to achieve a high level of data knowledge. Then, an initial code list related to the perspectives and challenges of 4.0 teams was generated and organized. In the next step, these codes were used to search for emerging categories by frequency analysis [104]. Subsequently, categories were refined by further searching for data that supported or rejected the proposed categories and the links between overlapping categories. The categories were then checked for consistency and compared to obtain reliability between encoders [104]. In compliance with high-quality research practice, the analysis was performed by a research team that included an "internal" researcher who collected the data and an "external" researcher who remained independent of the project. This approach is used extensively in psychology and social science research to provide different perspectives of data. The data were analyzed by both researchers, and disputes were resolved through discussion, ensuring a high level of agreement between the evaluators and the collected data. In the last stage, competencies corresponding to the previously identified requirements and perspectives were defined. In this process, competencies corresponding to the challenges and perspectives identified in the first stage of coding were systematically identified and improved. Then, the competencies were divided into two categories: team members and the team leader. Then, the relationships between the identified competencies and the previously identified challenges and perspectives were analyzed; in the next step, the main competencies were summarized. In order to triangulate, additional notes from project managers were used, and all 50 respondents and a focus group were interviewed to reach agreement on the identified competencies.

For the case study, one of the leading technical universities in Poland—which carries out large-scale interdisciplinary projects corresponding to Sustainable Industry 4.0 needs was selected. The analyzed projects were undertaken as part of the "Silesian University of Technology as a Center for Modern Education based on research and innovation" project, co-financed by the European Union. The implemented projects corresponded to topics reported by employers, in particular, regional enterprises and socio–economic institutions leading in the field of Sustainable Industry 4.0 and were of significant importance for the development of these enterprises and/or the development of the region itself. The projects were of research and development nature and included interdisciplinary issues from various disciplines (materials engineering, environmental engineering, biomedical engineering, computer science, construction, architecture, medicine, automation, mechanics and machine construction). For the implementation of each project, interdisciplinary teams were established, the work of which was managed by one main tutor and two auxiliary tutors. In addition, each project was supported by industry experts with competencies relevant to the project or experts from the University with documented scientific achievements in a given field. The projects were supported by industry experts, and the projects were derived from real problems reported by Sustainable Industry 4.0 organizations, which validates the teams as participating in Sustainable Industry 4.0 projects and makes the results applicable to other Sustainable Industry 4.0 challenges. Fifty respondents (17 women and 33 men) took part in the study. Each of the respondents came from a different project with varied interdisciplinarity. The analyzed projects included both (1) teams in which members each had expertise in a given area, and the project dealt with problems at the crossroads of at least two areas, and (2) teams where only a single member had expertise in a given area, and the project dealt with issues from several areas.

Respondents were randomly selected to participate in the study in order to generate generalizable results with the least possible burden. Freedom of speech and anonymity were ensured during the research in order to obtain the most reliable answers. Thanks to the advanced implementation of Sustainable Industry 4.0 in the analyzed entity, the obtained results may provide good practices for building teams executing interdisciplinary Sustainable Industry 4.0 projects.

4. Results

When asked about the challenges and perspectives of working in interdisciplinary project teams implementing Sustainable Industry 4.0 solutions, the respondents pointed out numerous problems they faced, but they also highlighted opportunities this work entailed. Table 1 presents perspectives and challenges defined by the 50 respondents, along with the frequency of individual indications.

As seen in Table 1, the biggest challenges for the respondents were: coordination of the project and its individual parts (32 out of 50 respondents), integrative team leadership (27 out of 50 respondents), establishing a common language among specialists from different disciplines (27 out of 50 respondents) and building a team consisting of specialists with competencies required for the project (21 out of 50 respondents). Regarding perspectives of working in an interdisciplinary Sustainable Industry 4.0 team, the respondents particularly pointed out the possibility of obtaining better results (26 out of 50 respondents) thanks to synergy, a wide view of the project (28 out of 50 respondents), proper division of competencies, which allowed individual team members to focus onn the areas of the project that were best known to them (25 out of 50 respondents) and mutual learning in the team (23 out of 50 respondents). For the next part of the study, the results of repeated interviews with the 50 respondents were used to define competence requirements for members of interdisciplinary Sustainable Industry 4.0 teams. Due to the fact that the interviews clearly emphasized the role of the leader(s) in project management, the research results (Figure 1) defined common competencies for the project team and additional competencies for the leader(s).

The description of these competencies is presented in Table 2.

As seen in Table 2 and Figure 1, the study identified a total of 15 competencies necessary to implement interdisciplinary Industry 4.0 projects. Among these 15 competencies, 9 (strategic perspective, communication skills, readiness to compromise, creativity, digital skills, ability to learn actively, ability to listen actively, conscientiousness and persuasion) are dedicated to the entire project team, while 6 (ability to coordinate work, conflict resolution, decision making, motivation, empowering and resource management) are assigned only to the leader(s). Table 3 shows the links between the defined competencies (Table 2) and the requirements and perspectives identified at an earlier stage (Table 1).

ID	Challenges	Number of Indications	ID	Perspectives	Number of Indications
W1	coordination of the project as a whole and the need to coordinate different parts of the project	32	P1	a comprehensive view of the project problem/product	28
W2	establishing a leader to integrate the project and share leadership within the team	27	P2	higher quality of the solution received	26
W3	different "language" presented by specialists from different disciplines	27	Р3	effective use of the competencies held by individual team members—everyone is competent in what they do	25
W4	building a team with all the competencies required in the project, including digital competencies	21	P4	mutual learning in a team, drawing on the experience and knowledge of other specialists	23
W5	more time for the right communication flow	18	Р5	the ability to solve more complex problems	22
W6	lack of possibility of mutual assistance in issues known only by individual team members	13	P6	the ability to solve innovative problems	20
W7	more time for the team to adjust to each other	12	P7	greater creativity	19
W8	rigid time frame	11	P8	the general knowledge of related disciplines acquired by the team members	16
W9	conflicts arising from a different perceptions of the problem	11			
W10	frequent changes in projects caused by unpredictable events related to the requirements of given disciplines and the emergence of new technological solutions	9	Р9	shared responsibility for the outcome of the project, which contributes to greater involvement of the entire team	14

 Table 1. Challenges and perspectives for interdisciplinary project teams implementing Sustainable

 Industry 4.0 solutions.

Source: own study based on the conducted research.

As can be seen in Table 3, all of the defined competencies are related to the challenges and perspectives of working in Sustainable Industry 4.0 interdisciplinary teams. Competencies dedicated to project team members slightly favor perspectives related to the implementation of analyzed projects than to cope with challenges, while the additional competencies identified for the project leader are slightly more focused on diagnosing challenges than understanding perspectives. Moreover, as shown in Table 3, most of the links between perspectives and requirements for project team competencies can be seen in relation to strategic perspective, communication skills and persuasion, while for leaders, competencies relate to work coordination, resource management, empowering and motivation.

The obtained results, as one of the first studies, present the competencies of team members and their leaders, combining the competencies necessary for Sustainable Industry 4.0 and their interdisciplinary context, thus identifying the interdisciplinary competencies of project teams and their leaders necessary for Sustainable Industry 4.0. Earlier studies in this area referred to the analyzed areas separately, not allowing for a holistic approach to this issue. What is more, unlike previous studies, this study highlights important

competencies in the analyzed areas, not only regarding the project manager himself, but also in relation to team members, which makes the results useful for HR managers who are searching for Sustainable Industry 4.0 employees as well as for project managers.



Figure 1. Competencies of interdisciplinary Sustainable Industry 4.0 teams.

Table 2. Description of competencies of interdisciplinary Sustainable Industry 4.0 teams.

ID	Competencies	Characteristics
SP	Strategic perspective	Provides a comprehensive view of the problems solved and considers the consequences of solving them in the short and long term.
СС	Communication capabilities	Active communication between all people involved, tailored to the requirements of each member. It also means being accessible to other team members and being able to communicate in a digital environment.
RC	Readiness to compromise	Enables working on the principles of cooperation and collaboration. It allows accepting different ideas, searching for a common solution to the problem and understanding interpersonal relations.
С	Creativity	Enables the use of different skills and knowledge, from different perspectives and worldviews of project team members to achieve a common solution. It allows going beyond one's own thinking patterns and to operate efficiently at the crossroads of different fields and disciplines.
DS	Digital skills	The ability to communicate and use information and data and to use digital technologies to solve problems and think critically. Building a learning environment in which new technologies are used in proposed solutions and for teamwork.

Competencies	Characteristics						
Ability to learn actively	A willingness to experiment in order to learn and to innovate using knowledge from different disciplines and different technologies, guided by the digital way of thinking.						
Ability to listen actively	Focus on cooperation in a culture of feedback, and openness to suggestions and ideas from team members outside their discipline						
Conscientiousness	A clear commitment to the direction one is working in, and to encourage other team members to support the direction chosen. It supports timely completion of tasks and cooperation at every stage of the project.						
Persuasion	Building a learning and innovation environment that enables presentation of an approach to other team members to encourage them to change their point of view, while understanding the position of other team members, recognizing the need to listen to that position and indicating the rationale for a change in approach						
Team leader							
Work coordinating skills	Enables coordination of the different parts of the project as a whole and directing the work of the project team members in such a way that together they achieve the desired result						
Conflict resolution skills	Eliminates, reduces and weakens conflict among team members, or gives it a certain direction in order to use its positive and creative features and thus achieve a better-quality solution						
Ability to make decisions	The ability to gather relevant information from a wide range of sources (including consideration of the challenges of using new, beneficial technologies) and, on this basis, make the right decisions						
Motivating others to act	The ability to influence (without using power) team members to stimulate them to achieve their goals, and to support them in their efforts						
Empowering	Involving team members in the decision-making process and engaging them in digital environments. Enables the creation of a participatory culture based on trust and cooperation, a sense of belonging and active participation.						
Resource management	Ability to organize work in a team and assign responsibilities in accordance with the competencies and personal characteristics of team members, and to monitor the effectiveness of teamwork						
	Competencies Ability to learn actively Ability to listen actively Conscientiousness Persuasion Work coordinating skills Conflict resolution skills Ability to make decisions Motivating others to act Empowering Resource management						

Table 2. Cont.

Source: own study based on the conducted research.

Table 3. Linking the defined competencies to the identified challenges and perspectives related to working in interdisciplinary Sustainable Industry 4.0 teams.

Competencies/ Challenges and Perspectives	Strategic Perspective	Communications Skills	Readiness to Compromise	Creativity	Digital Skills	Active Learning	Active Listening	Conscientiousness	Persuasion	Work Coordinating Skills	Conflict Resolution Skills	Decision Making	Motivation	Empowering	Resource Management
coordination of the project		+								+					
appointing a leader										+	+	+	+	+	+

Table 3. Cont.

Competencies/ Challenges and Perspectives	Strategic Perspective	Communications Skills	Readiness to Compromise	Creativity	Digital Skills	Active Learning	Active Listening	Conscientiousness	Persuasion	Work Coordinating Skills	Conflict Resolution Skills	Decision Making	Motivation	Empowering	Resource Management
different language	+	+	+				+		+		+				
required competencies of a team										+			+		+
more time for communication		+	+							+			+	+	
lack of possibility of mutual assistance								+	+						
more time to adjust to each other		+	+										+	+	
rigid time frame								+							
conflicts		+							+		+	+			
frequent changes	+				+							+			
a comprehensive view of the problem	+	+													
higher-quality results				+	+	+	+			+					+
effective use of individual competencies		+							+				+		+
mutual learning in a team		+				+								+	
ability to solve complex problems	+			+	+						+				+
ability to solve innovative problems	+	+		+	+	+	+								+
greater creativity	+			+		+	+			+					
general knowledge of the related disciplines	+					+	+								
shared responsibility for the results		+						+					+		

Source: own study based on the conducted research.

5. Discussion

The article poses two research questions. The first concerned requirements and opportunities connected with interdisciplinary Sustainable Industry 4.0 work, and the second concerned competencies of such team members and their leaders.

Answering the questions related to requirements and opportunities of such work, it was found that, in the studied projects, the biggest challenges were coordination of the project and combining its individual parts, integrative team leadership, establishing a common language among specialists from different disciplines and building a team, while perspectives connected with such work included the possibility of obtaining better results thanks to synergy, a wide view of the project, proper division of competencies and mutual learning in the team.

Answering the second research questions, related to competencies of interdisciplinary Sustainable Industry 4.0 teams, it was found that 15 competencies are necessary, and, among these 15 competencies, 9 are dedicated to the entire project team (strategic perspective, communication skills, readiness to compromise, creativity, digital skills, ability to learn actively, ability to listen actively, conscientiousness and persuasion), while 6 are assigned only to the leader(s) (ability to coordinate work, conflict resolution, decision making, motivation, empowering and resource management).

The considerations presented in the article allow a deeper understanding of the specificity of work in Sustainable Industry 4.0 teams. When analyzing the characteristics of Sustainable Industry 4.0 work and the requirements set here for various groups of professions, it can be concluded that the 4th Industrial Revolution has unequivocally transformed the business sector around the world. This is shown in various studies [105,106], which emphasize that, in order to ensure a successful transition to Sustainable Industry 4.0, it will be necessary to change the management paradigm and apply a new approach to teaching and improving the individual competencies of employees. Although research in the field of Sustainable Industry 4.0 uses different levels of analysis to explain the changes brought about by the new fourth era of digitization and their impact on the development of the labor market and the technologies themselves, as many authors point out [38], the literature lacks an analysis of the impact of the sustainable 4.0 paradigm on employees. Some of the few publications that pay attention to this aspect are the studies by [53,107,108], in which the authors emphasize that Sustainable Industry 4.0 poses challenges in the social sphere, requiring companies to develop the highest level of competence of their specialists and attract new talent able to deal with the increased complexity of new technologies.

As the analysis shows, the changes caused by the new 4.0 paradigm are also reflected in the higher requirements for project team members, which are indicated as an important element of Sustainable Industry 4.0. This is also associated with a change in the approach to 4.0 project management itself, which, in this case, is more complex, innovative, burdened with a higher level of risk and requires the cooperation of highly specialized employees from various disciplines, often supported by employees from their economic–social environment. This is consistent with previous analyses in this area [22,109] that emphasize that Sustainable Industry 4.0 will bring radical changes in relation to project management. For example, refs. [22] points out that traditional project management styles must be changed in order to adapt to the 4th Industrialization Revolution with many variations, and that there are many challenges connected with project managers in Sustainable Industry 4.0, as it is certain that a project manager must have improved soft and hard skills to implement complex, autonomous and sustainable Industry 4.0 projects.

One of the key aspects of cooperation in 4.0 teams is their interdisciplinary nature due to the fact that interdisciplinarity is strongly related to innovation [110]—which is an inseparable element of Sustainable Industry 4.0-and due to the increasing complexity of projects and tasks faced by modern employees [111]. The requirement of interdisciplinarity of projects implemented for Sustainable Industry 4.0 is also emphasized in publications by other researchers. Refs. [112,113] indicate that complex problems cannot be solved within one discipline, and employees, including team members, require appropriate personal and social skills, defined as interdisciplinary competencies, in the era of the modern 4.0 economy. Among these competencies, the authors mention: solving complex problems in various disciplines, communication between disciplines, handling interdisciplinary cooperation and teamwork, and using integration potential to create innovation. Similar conclusions were obtained in this study, which shows that the most important challenges and perspectives related to the work of such teams includes coordination of individual parts of the project, communication between members from different disciplines, and a broad view of the problems raised in the project. This is also in line with other research, where attention was paid to difficulties in integrating the different parts of the project e.g., ref. [50] or problems in going beyond their knowledge disciplines, e.g., ref. [77].

Moreover, leadership is extremely important in the context of the analyzed subject matter, as it enables effective teamwork. Contemporary research in this area highlights the role of inclusive leadership, which is linked to the concept of shared leadership, as also noted in this research. As [114,115] point out, the project team also has an inclusive role,

which is to fill knowledge gaps by providing people with the right competencies to the team at different stages of the project. That is why it is so important that team members are entitled to use their competencies effectively in developing a project solution and to take on leadership roles if necessary [116]. This is consistent with research, e.g., ref. [88] conducted in the area of Sustainable Industry 4.0, which stresses that if leadership is more collaborative, participatory and decentralized, people feel more encouraged to make decisions and experiment to create an agile and responsive network culture.

According to the results of the research, the members of teams should mainly have competencies such as: strategic perspective, communication skills, willingness to compromise, creativity, digital skills, ability to learn actively, ability to listen actively, conscientiousness and persuasion. This is confirmed by previous reports that indicate that there should be an important soft component to manage (and face) the introduction of new technologies [39], and so-called soft skills are increasingly important in the current digital age, and can constitute its key element as competences resistant to changes [38,41].

In light of the conducted research, it was also revealed that the project manager plays a key role in the digitization process. Due to changes in work methods and the challenges facing Sustainable Industry 4.0, changes and challenges are also faced by project managers. Key competencies identified for the 4.0 project leadership role were highlighted: ability to coordinate work, conflict resolution, decision making, motivation, empowering and resource management. This is consistent with other research on 4.0 project leader competencies. This is also confirmed by [7], which indicates that the most significant changes related to the requirements of Sustainable Industry 4.0 project managers are communication style, interaction, speed and ability to work, and basic knowledge. They also add that the project manager must have improved soft and hard skills to implement complex and autonomous Industry 4.0 projects, and that the competencies related to Sustainable Industry 4.0 go beyond those defined by PMI or IPMA for the role of project manager.

The main limitation of this study is, by definition, a limited research method. Therefore, it is necessary to repeat this study in different contexts, in organizations of different sizes and types of projects, and in different countries. While our qualitative method allowed for an in-depth examination of the context, it also, to some extent, limits the reliability of the findings. By incorporating quantitative methods and replicating this study with either a quantitative or blended method, future research can further enhance the reliability of these findings. For example, quantitative research can be conducted to analyze the relationship between the identified competencies of interdisciplinary Industrial 4.0 project teams and the success of these projects. Moreover, some of the identified challenges and opportunities related to interdisciplinary Sustainable Industry 4.0 work may also apply to other areas. Therefore, future research could shed more light on the clear challenges and perspectives related to interdisciplinary Sustainable Industry 4.0 work, as some of them may relate to overall organizational transformation. Moreover, it is also worth considering how the competencies of such teams in other countries and cultures are developed, as the project environment is expected to have a significant impact on the competence requirements of the analyzed teams. A comparative analysis of competencies of teams implementing singlearea Sustainable Industry 4.0 projects with the results obtained would also be cognitively interesting. This would indicate whether team cooperation influences the competencies required to a large extent. One limitation is also that 50 individuals spoke from the perspective of 50 teams, so individual's findings were confounded by their teams. Therefore, it is not possible to conclusively state whether the obtained results represent individual idiosyncrasies or team differences. Therefore, in future research, the target group can be expanded to include multiple members of a given team to get a more complete picture of the differences between teams.

6. Conclusions

The obtained results, as one of the first studies on this topic, present the competencies of team members and their leaders, combining the competencies necessary for Sustainable

Industry 4.0 and their interdisciplinary projects, thus identifying the interdisciplinary competencies of project teams and their leaders necessary for Sustainable Industry 4.0. Earlier studies of this topic referred to the analyzed areas separately, not allowing a holistic approach to this issue. What is more, unlike previous studies, this study highlights important competencies in the analyzed areas, not only in relation to the project manager himself, but also in relation to team members, which makes the obtained results useful for HR managers who are searching for Sustainable Industry 4.0 employees as well as for project managers.

This article is a contribution to the literature on management in the area of team competency management and from the perspective of organizations implementing sustainable 4.0 projects, thus extending the theory of human resource management to Sustainable Industry 4.0 and developing the argument for a greater focus on improving the soft skills of employees, in particular on developing their interdisciplinary skills. In practice, the article responds to the needs of companies and educational institutions (e.g., universities and colleges) to reflect on the desired competencies of employees—including, in particular, members of project teams implementing Sustainable Industry 4.0. projects—to cope with upcoming industry and sectoral changes. Hence, the obtained results are the basis for identifying the following recommendations for future teams implementing similar projects:

- Awareness by the organization that Sustainable Industry 4.0 projects require interdisciplinary project teams and often also additional external support. Project team members can come from areas directly and indirectly related to the project topic. In this context, it is important to create organizational opportunities for cooperation within such teams.
- Making the organization's management aware of important competencies necessary to implement interdisciplinary Sustainable Industry 4.0 projects.
- Taking into account that project management 4.0 very often requires a transition from a classic to an agile approach to project management, which is the basis for innovation and customer satisfaction.
- The need for an integrative approach to team leadership by the project manager, aimed at supporting an agile environment,
- Taking into account whether the proposed project management frameworks allow for the additional time and communication costs required for the successful implementation of interdisciplinary 4.0 projects.

The study also contributes to the knowledge base concerning project management in the area of individual competence development of project team members and the management of interdisciplinary and sustainable 4.0 projects, indicating that the modern 4th Industrial Revolution requires a new, more flexible and agile approach to project management and new competencies of team members, including, in particular, the project manager. Thus, the obtained results will allow for more thoughtful team selection and, consequently, for greater effectiveness of the implemented projects.

Funding: This research was funded by Silesian University of Technology, grant number 13/040/BKM22/0112.

Institutional Review Board Statement: The study was conducted in accordance with the ethics regulations of the Silesian University of Technology.

Informed Consent Statement: Not applicable.

Conflicts of Interest: The author declares no conflict of interest.

References

- 1. Svejvig, P.; Andersen, P. Rethinking project management: A structured literature review with a critical look at the brave new world. *Int. J. Proj. Manag.* 2015, 33, 278–290. [CrossRef]
- Möller, D.P.F. Digital Manufacturing/Industry 4.0. In Guide to Computing Fundamentals in Cyber-Physical Systems (Computer Communications and Networks); Springer: Berlin/Heidelberg, Germany, 2016; pp. 307–375. ISBN 978-3-319-25176-9.
- Dalenogare, L.S.; Benitez, G.B.; Ayala, N.F.; Frank, A.G. The expected contribution of Industry 4.0 technologies for industrial performance. Int. J. Prod. Econ. 2018, 204, 383–394. [CrossRef]

- Cobo, M.J.; Jürgens, B.; Herrero-Solana, V.; Martínez, M.A.; Herrera-Viedma, E. Industry 4.0: A perspective based on bibliometric analysis. Procedia Comput. Sci. 2018, 139, 364–371. [CrossRef]
- Lenka, S.; Parida, V.; Wincent, J. Digitalization Capabilities as Enablers of Value Co-Creation in Servitizing Firms: Digitalization Capabilities. Psychol. Mark. 2017, 34, 92–100. [CrossRef]
- 6. Porter, M.E.; Heppelmann, J.E. How smart, connected products are transforming companies. Harv. Bus. Rev. 2015, 93, 96–114.
- Ribeiro, A.; Amaral, A.; Barros, T. Project Manager Competencies in the context of the Industry 4.0. Procedia Comput. Sci. 2021, 181, 803–810. [CrossRef]
- Caiado, R.G.G.; Scavarda, L.F.; Azevedo, B.D.; de Mattos Nascimento, D.L.; Quelhas, O.L.G. Challenges and Benefits of Sustainable Industry 4.0 for Operations and Supply Chain Management—A Framework Headed toward the 2030 Agenda. Sustainability 2022, 14, 830. [CrossRef]
- Beier, G.; Ullrich, A.; Niehoff, S.; Reißig, M.; Habich, M. Industry 4.0: How it is defined from a sociotechnical perspective and how much sustainability it includes—A literature review. J. Clean. Prod. 2020, 259, 120856. [CrossRef]
- Sony, M.; Naik, S. Industry 4.0 integration with socio-technical systems theory: A systematic review and proposed theoretical model. *Technol. Soc.* 2020, 61, 101248. [CrossRef]
- De Sousa Jabbour, A.B.L.; Jabbour, C.J.C.; Foropon, C.; Godinho Filho, M. When titans meet—Can industry 4.0 revolutionise the environmentally-sustainable manufacturing wave? The role of critical success factors. *Technol. Forecast. Soc. Change* 2018, 132, 18–25. [CrossRef]
- 12. Bushuyev, S.; Bushuiev, D.; Bushuieva, V. Project Management during Infodemic of the COVID-19 Pandemic. *Innov. Technol. Sci. Solut. Ind.* 2020, 2, 13–21. [CrossRef]
- Maskuriy, R.; Selamat, A.; Maresova, P.; Krejcar, O.; David, O.O. Industry 4.0 for the Construction Industry: Review of Management Perspective. *Economies* 2019, 7, 68. [CrossRef]
- Badri, A.; Boudreau-Trudel, B.; Souissi, A.S. Occupational health and safety in the industry 4.0 era: A cause for major concern? Saf. Sci. 2018, 109, 403–411. [CrossRef]
- Neumann, W.P.; Winkelhaus, S.; Grosse, E.H.; Glock, C.H. Industry 4.0 and the human factor—A systems framework and analysis methodology for successful development. Int. J. Prod. Econ. 2021, 233, 107992. [CrossRef]
- Stachová, K.; Papula, J.; Stacho, Z.; Kohnová, L. External Partnerships in Employee Education and Development as the Key to Facing Industry 4.0 Challenges. Sustainability 2019, 11, 345. [CrossRef]
- Hirman, M.; Benesova, A.; Steiner, F.; Tupa, J. Project Management during the Industry 4.0 Implementation with Risk Factor Analysis. Procedia Manuf. 2019, 38, 1181–1188. [CrossRef]
- López-Robles, J.R.; Otegi-Olaso, J.R.; Porto Gómez, I.; Cobo, M.J. 30 years of intelligence models in management and business: A bibliometric review. Int. J. Inf. Manag. 2019, 48, 22–38. [CrossRef]
- López-Robles, J.R.; Otegi-Olaso, J.R.; Cobo, M.J.; Bertolin-Furstenau, L.; Leonardo and Kremer-Sott, M.; López-Robles, L.D.; Gamboa-Rosales, N.K. The relationship between Project Management and Industry 4.0: Bibliometric analysis of main research areas through Scopus. In Proceedings of the 3rd International Conference on Research and Education in Project Management— REPM 2020, Bilbao, Spain, 20–21 February 2020.
- Marnewick, C.; Marnewick, A.L. The Demands of Industry 4.0 on Project Teams. *IEEE Trans. Eng. Manag.* 2019, 67, 941–949. [CrossRef]
- Weber, B.; Butschan, J.; Heidenreich, S. Tackling hurdles to digital transformation—The role of competencies for successful IIoT implementation. In Proceedings of the 2017 IEEE Technology & Engineering Management Conference (TEMSCON), Santa Clara, CA, USA, 8–10 June 2017; pp. 312–317.
- Win, T.Z.; Kham, S.M. Transformation of Project Management in Industry 4.0. In Proceedings of the 12th International Conference on Project Management (ProMAC2018), Bangkok, Thailand, 27 November–1 December 2018.
- Project Management Institute. The Project Manager of the Future: Developing Digital-Age Project Management Skills to Thrive In Disruptive Times; Project Management Institute: Newtown, PA, USA, 2018.
- Cakmakci, M. Interaction in Project Management Approach Within Industry 4.0. In Advances in Manufacturing II; Trojanowska, J., Ciszak, O., Machado, J.M., Pavlenko, I., Eds.; Lecture Notes in Mechanical Engineering; Springer: Berlin/Heidelberg, Germany, 2019; pp. 176–189. ISBN 978-3-030-18714-9.
- Simion, C.P.; Popa, S.C.; Albu, C. Project Management 4.0—Project Management in the Digital Era; Faculty of Management, Academy of Economic Studies: Bucharest, Romania, 2018; Volume 12, pp. 93–100.
- Amadi-Echendu, J.E. Thinking styles of technical knowledge workers in the systems of innovation paradigm. *Technol. Forecast.* Soc. Change 2007, 74, 1204–1214. [CrossRef]
- 27. Lee, M.; Lee, Y.; Chou, C. Essential Implication of the Digital Transformation in Industry 4.0. J. Sci. Ind. Res. 2017, 76, 465–467.
- 28. Schwab, K. The Fourth Industrial Revolution; World Economic Forum: Cologny, Switzerland, 2016.
- 29. Culot, G.; Nassimbeni, G.; Orzes, G.; Sartor, M. Behind the definition of Industry 4.0: Analysis and open questions. *Int. J. Prod. Econ.* 2020, 226, 107617. [CrossRef]
- Lezzi, M.; Lazoi, M.; Corallo, A. Cybersecurity for Industry 4.0 in the current literature: A reference framework. *Comput. Ind.* 2018, 103, 97–110. [CrossRef]
- Mariani, M.; Borghi, M. Industry 4.0: A bibliometric review of its managerial intellectual structure and potential evolution in the service industries. *Technol. Forecast. Soc. Change* 2019, 149, 119752. [CrossRef]

- Becker, T.; Stern, H. Future Trends in Human Work area Design for Cyber-Physical Production Systems. Procedia CIRP 2016, 57, 404–409. [CrossRef]
- Kadir, B.A.; Broberg, O. Human well-being and system performance in the transition to industry 4.0. Int. J. Ind. Ergon. 2020, 76, 102936. [CrossRef]
- Kiel, D.; Müller, J.M.; Arnold, C.; Voigt, K.-I. Sustainable Industrial Value Creation: Benefits and Challenges of Industry 4.0. Int. J. Innov. Manag. 2017, 21, 1740015. [CrossRef]
- Kane, G.C.; Palmer, D.; Phillips, A.N.; Kiron, D.; Buckley, N. Coming of Age Digitally. MIT Sloan Manag. Rev. Deloitte Insights 2018, 59, 1–10.
- Ustundag, A.; Cevikcan, E. Industry 4.0: Managing The Digital Transformation; Springer Series in Advanced Manufacturing; Springer: Berlin/Heidelberg, Germany, 2018; ISBN 978-3-319-57869-9.
- Kagermann, H.; Riemensperger, F.; Hoke, D.; Schuh, G.; Scheer, A.-W.; Spath, D.; Path, B.; Leukert, B.; Wahlster, W.; Rohleder, B.; et al. *Smart Service Welt*, Abschlussbericht, Langversion; Federal Ministry for Economic Affairs and Energy: Berlin, Germany, 2015.
- Fareri, S.; Fantoni, G.; Chiarello, F.; Coli, E.; Binda, A. Estimating Industry 4.0 impact on job profiles and skills using text mining. Comput. Ind. 2020, 118, 103222. [CrossRef]
- Chui, M.; Manyika, L.; Miremadi, M. Where Machines could replace humans and where they can't (yet). McKinsey Q. 2016, 3, 58–69.
- Chryssolouris, G.; Mavrikios, D.; Mourtzis, D. Manufacturing Systems: Skills & Competencies for the Future. Procedia CIRP 2013, 7, 17–24. [CrossRef]
- Frey, C.B.; Osborne, M.A. The future of employment: How susceptible are jobs to computerisation? *Technol. Forecast. Soc. Change* 2017, 114, 254–280. [CrossRef]
- Weber, E. Industry 4.0: Job-Producer or Employment-Destroyer; Institut f
 ür Arbeitsmarkt- und Berufsforschung (IAB): N
 ürnberg, Germany, 2016.
- Guzmán, V.E.; Muschard, B.; Gerolamo, M.; Kohl, H.; Rozenfeld, H. Characteristics and Skills of Leadership in the Context of Industry 4.0. Procedia Manuf. 2020, 43, 543–550. [CrossRef]
- Instituto Universitário de Lisboa and CIEO, Algarve University, Lisboa, Portugal; Sousa, M.J.; Santos, V.; Sacavém, A.; dos Reis, I.P.; Sampaio, M.C. 4.0 Leadership Skills in Hospitality Sector. J. Rev. Glob. Econ. 2019, 8, 105–117. [CrossRef]
- Whysall, Z.; Owtram, M.; Brittain, S. The new talent management challenges of Industry 4.0. J. Manag. Dev. 2019, 38, 118–129. [CrossRef]
- Bauer, W.; Schuler, S.; Hornung, T.; Decker, J. Development of a Procedure Model for Human-Centered Industry 4.0 Projects. Procedia Manuf. 2019, 39, 877–885. [CrossRef]
- Brandshaug, S.W.; Sjølie, E. In liminality: Interdisciplinary teams learning through challenges. *High. Educ. Skills Work-Based Learn.* 2021, 11, 406–419. [CrossRef]
- Bercovitz, J.; Feldman, M. The mechanisms of collaboration in inventive teams: Composition, social networks, and geography. *Res. Policy* 2011, 40, 81–93. [CrossRef]
- 49. Cummings, J.N.; Kiesler, S. Organization Theory and the Changing Nature of Science. J. Organ. Des. 2014, 3, 1–16. [CrossRef]
- Haeussler, C.; Sauermann, H. Division of labor in collaborative knowledge production: The role of team size and interdisciplinarity. *Res. Policy* 2020, 49, 103987. [CrossRef]
- Edmondson, A.C.; Nembhard, I.M. Product Development and Learning in Project Teams: The Challenges Are the Benefits. J. Prod. Innov. Manag. 2009, 26, 123–138. [CrossRef]
- Zdonek, I.; Podgórska, M.; Hysa, B. The Competence for Project Team Members in the Conditions of Remote Working. Found. Manag. 2017, 9, 213–224. [CrossRef]
- Kipper, L.M.; Iepsen, S.; Dal Forno, A.J.; Frozza, R.; Furstenau, L.; Agnes, J.; Cossul, D. Scientific mapping to identify competencies required by industry 4.0. Technol. Soc. 2021, 64, 101454. [CrossRef]
- 54. Wahl, M. Strategic Factor Analysis for Industry 4.0. J. Secur. Sustain. Issues 2015, 5, 241-247. [CrossRef]
- 55. Hirsch-Kreinsen, H. Digitization of industrial work: Development paths and prospects. J. Labour Mark. Res. 2016, 49, 1–14. [CrossRef]
- Abele, E.; Metternich, J.; Tisch, M.; Chryssolouris, G.; Sihn, W.; ElMaraghy, H.; Hummel, V.; Ranz, F. Learning Factories for Research, Education, and Training. *Procedia CIRP* 2015, 32, 1–6. [CrossRef]
- Butler, C.W.; Vijayasarathy, L.R.; Roberts, N. Managing Software Development Projects for Success: Aligning Plan- and Agility-Based Approaches to Project Complexity and Project Dynamism. *Proj. Manag. J.* 2020, *51*, 262–277. [CrossRef]
- Scholz, J.-A.; Sieckmann, F.; Kohl, H. Implementation with agile project management approaches: Case Study of an Industrie 4.0 Learning Factory in China. *Procedia Manuf.* 2020, 45, 234–239. [CrossRef]
- Dreesen, T.; Diegmann, P.; Rosenkranz, C. The Impact of Modes, Styles, and Congruence of Control on Agile Teams: Insights from a Multiple Case Study. In Proceedings of the HICSS 2020, Maui, HI, USA, 7–10 January 2020.
- Hennel, P.; Rosenkranz, C. Investigating the "Socio" in Socio-Technical Development: The Case for Psychological Safety in Agile Information Systems Development. Proj. Manag. J. 2021, 52, 11–30. [CrossRef]
- Recker, J.; Holten, R.; Hummel, M.; Rosenkranz, C. How Agile Practices Impact Customer Responsiveness and Development Success: A Field Study. Proj. Manag. J. 2017, 48, 99–121. [CrossRef]

- National Academy of Sciences; National Academy of Engineering; Institute of Medicine. National Academies Committee on Facilitating Interdisciplinary Research; The National Academies Press: Columbia, WA, USA, 2005.
- Chiocchio, F. Defining Project Teams. In *The Psychology and Management of Project Teams*; Chiocchio, F., Kelloway, Hobbs, B., Eds.; Oxford University Press: Oxford, UK, 2015; pp. 40–73. ISBN 978-0-19-986137-8.
- Cannella, A.A.; Park, J.-H.; Lee, H.-U. Top Management Team Functional Background Diversity and Firm Performance: Examining The Roles of Team Member Colocation and Environmental Uncertainty. Acad. Manag. J. 2008, 51, 768–784. [CrossRef]
- Joseph Garcia, A.; Mollaoglu, S. Individuals' Capacities to Apply Transferred Knowledge in AEC Project Teams. J. Constr. Eng. Manag. 2020, 146, 4020016. [CrossRef]
- Balakrishnan, A.D.; Kiesler, S.; Cummings, J.; Zadeh, R. Research Team Integration: What It Is and Why it Matters; ACM Press: New York, NY, USA, 2011; pp. 523–532.
- Kozlowski, S.W.J.; Chao, G.T. Macrocognition, team learning, and team knowledge: Origins, emergence, and measurement. In *Theories of Team Cognition: Cross-Disciplinary Perspectives*; Series in applied psychology; Routledge/Taylor & Francis Group: New York, NY, USA, 2012; pp. 19–48. ISBN 978-0-415-87413-7.
- Dougherty, D.; Tolboom, J.N. Creative organizing to enable organizational creativity. In Handbook of Organizational Creativity; Lawrence Erlbaum Associates: New York, NY, USA, 2008; ISBN 978-0-8058-4072-8.
- Bechky, B.A. Sharing meaning across occupational communities: The transformation of understanding on a production floor. Organ. Sci. 2003, 14, 312–330. [CrossRef]
- Hargadon, A.B.; Bechky, B.A. When Collections of Creatives Become Creative Collectives: A Field Study of Problem Solving at Work. Organ. Sci. 2006, 17, 484–500. [CrossRef]
- 71. Tsoukas, H. A Dialogical Approach to the Creation of New Knowledge in Organizations. Organ. Sci. 2009, 20, 941–957. [CrossRef]
- 72. Walsh, J.P.; Lee, Y.-N. The bureaucratization of science. Res. Policy 2015, 44, 1584–1600. [CrossRef]
- Leahey, E.; Beckman, C.M.; Stanko, T.L. Prominent but Less Productive: The Impact of Interdisciplinarity on Scientists' Research. Adm. Sci. Q. 2017, 62, 105–139. [CrossRef]
- Boix Mansilla, V.; Lamont, M.; Sato, K. Shared Cognitive–Emotional–Interactional Platforms: Markers and Conditions for Successful Interdisciplinary Collaborations. Sci. Technol. Hum. Values 2016, 41, 571–612. [CrossRef]
- Fiore, S.M. Interdisciplinarity as Teamwork: How the Science of Teams Can Inform Team Science. Small Group Res. 2008, 39, 251–277. [CrossRef]
- 76. Bennett, L.M.; Gadlin, H. Collaboration and Team Science: From Theory to Practice. J. Investig. Med. 2012, 60, 768–775. [CrossRef]
- Majchrzak, A.; More, P.H.B.; Faraj, S. Transcending Knowledge Differences in Cross-Functional Teams. Organ. Sci. 2012, 23, 951–970. [CrossRef]
- Cronin, M.A.; Weingart, L.R. Representational gaps, information processing, and conflict in functionally diverse teams. *Acad. Manag. Rev.* 2007, 32, 761–773. [CrossRef]
- Van Der Vegt, G.S.; Bunderson, J.S. Learning and Performance in Multidisciplinary Teams: The Importance of Collective Team Identification. Acad. Manag. J. 2005, 48, 532–547. [CrossRef]
- Dougherty, D. Reimagining the Differentiation and Integration of Work for Sustained Product Innovation. Organ. Sci. 2001, 12, 612–631. [CrossRef]
- Hackman, J.R. Leading Teams: Setting the Stage for Great Performances; Harvard Business School Press: Boston, MA, USA, 2002; ISBN 978-1-57851-333-8.
- Ilgen, D.R.; Hollenbeck, J.R.; Johnson, M.; Jundt, D. Teams in Organizations: From Input-Process-Output Models to IMOI Models. Annu. Rev. Psychol. 2005, 56, 517–543. [CrossRef]
- Burke, C.S.; Fiore, S.M.; Salas, E. The Role of Shared Cognition in Enabling Shared Leadership and Team Adaptability. In Shared Leadership: Reframing the Hows and Whys of Leadership; SAGE Publications, Inc.: Thousand Oaks, CA, USA, 2003; pp. 103–122. ISBN 978-0-7619-2624-5.
- Salazar, M.R.; Widmer, K.; Doiron, K.; Lant, T.K. Leader Integrative Capabilities: A Catalyst for Effective Interdisciplinary Teams. In *Strategies for Team Science Success*; Hall, K.L., Vogel, A.L., Croyle, R.T., Eds.; Springer: Berlin/Heidelberg, Germany, 2019; pp. 313–328. ISBN 978-3-030-20990-2.
- Kelly, R. Constructing Leadership 4.0: Swarm Leadership and the Fourth Industrial Revolution, 1st ed.; Springer: Berlin/Heidelberg, Germany, 2019; ISBN 978-3-319-98062-1.
- Oberer, B.; Erkollar, A. Leadership 4.0: Digital Leaders in the Age of Industry 4.0. Int. J. Organ. Leadersh. 2018, 7, 404–412. [CrossRef]
- 87. Bolte, S.; Dehmer, J.; Niemann, J. Digital Leadership 4.0. Acta Tech. Napoc.-Ser. Appl. Math. Mech. Eng. 2018, 61, 637–646.
- Sivathanu, B.; Pillai, R. Smart HR 4.0—How industry 4.0 is disrupting HR. Hum. Resour. Manag. Int. Dig. 2018, 26, 7–11. [CrossRef]
- Zaccaro, S.J.; Heinen, B.; Shuffler, M. Team leadership and team effectiveness. In *Team Effectiveness in Complex Organizations:* Cross-Disciplinary Perspectives and Approaches; Taylor & Francis Group—Routledge: London, UK, 2009.
- Harvey, S. Creative Synthesis: Exploring the Process of Extraordinary Group Creativity. Acad. Manag. Rev. 2014, 39, 324–343. [CrossRef]
- Taylor, A.; Greve, H.R. Superman or the Fantastic Four? knowledge combination and experience in Innovative Teams. Acad. Manag. J. 2006, 49, 723–740. [CrossRef]

- 92. Yin, R.K. Case Study Research: Design and Methods, 5th ed.; SAGE: Thousand Oaks, LA, USA, 2014; ISBN 978-1-4522-4256-9.
- Eisenhardt, K.M.; Graebner, M.E. Theory Building from Cases: Opportunities And Challenges. Acad. Manag. J. 2007, 50, 25–32. [CrossRef]
- 94. Selsor, W. Managerial Competencies Driving Successful Change Initiatives: A Multiple Case Study of Healthcare Administrators. Ph.D. Thesis, Walden University, Minneapolis, MN, USA, 2020. Walden Dissertations and Doctoral Studies.
- Müller, J.M. Assessing the barriers to Industry 4.0 implementation from a workers' perspective. *IFAC-PapersOnLine* 2019, 52, 2189–2194. [CrossRef]
- 96. Guion, L.A.; Diehl, D.C.; McDonald, D. Triangulation: Establishing the Validity of Qualitative Studies. *EDIS* 2011, 2011, 3. [CrossRef]
- Merriam, S.B.; Tisdell, E.J. Qualitative Research: A Guide to Design and Implementation, 4th ed.; The Jossey-Bass Higher and Adult Education Series; John Wiley & Sons: Hoboken, NJ, USA, 2015; ISBN 978-1-119-00365-6.
- 98. Gibbert, M.; Ruigrok, W.; Wicki, B. What passes as a rigorous case study? Strateg. Manag. J. 2008, 29, 1465–1474. [CrossRef]
- Huber, G.P.; Power, D.J. Retrospective reports of strategic-level managers: Guidelines for increasing their accuracy. *Strateg. Manag.* J. 1985, 6, 171–180. [CrossRef]
- Miles, M.B.; Huberman, A.M. Qualitative Data Analysis: An Expanded Sourcebook, 2nd ed.; Sage Publications: Thousand Oaks, CA, USA, 1994; ISBN 978-0-8039-4653-8.
- 101. Braun, V.; Clarke, V. Using thematic analysis in psychology. Qual. Res. Psychol. 2006, 3, 77–101. [CrossRef]
- Guest, G.; MacQueen, K.; Namey, E. Applied Thematic Analysis; SAGE Publications, Inc.: Thousand Oaks, CA, USA, 2012; ISBN 978-1-4129-7167-6.
- Edmondson, A.C.; Mcmanus, S.E. Methodological fit in management field research. Acad. Manag. Rev. 2007, 32, 1246–1264. [CrossRef]
- 104. Holsti, O.R. Content Anaylis for the Social Sciences and Humanities; Addison-Wesley: Boston, MA, USA, 1969; ISBN 978-0-201-02940-6.
- Romero, D.; Stahre, J.; Taisch, M. The Operator 4.0: Towards socially sustainable factories of the future. Comput. Ind. Eng. 2020, 139, 106128. [CrossRef]
- Sgarbossa, F.; Grosse, E.H.; Neumann, W.P.; Battini, D.; Glock, C.H. Human factors in production and logistics systems of the future. *Annu. Rev. Control* 2020, 49, 295–305. [CrossRef]
- Bokrantz, J.; Skoogh, A.; Berlin, C.; Stahre, J. Maintenance in digitalised manufacturing: Delphi-based scenarios for 2030. Int. J. Prod. Econ. 2017, 191, 154–169. [CrossRef]
- Enke, J.; Glass, R.; Kreß, A.; Hambach, J.; Tisch, M.; Metternich, J. Industrie 4.0—Competencies for a modern production system. Procedia Manuf. 2018, 23, 267–272. [CrossRef]
- Taner, Z.T.; Zübeyde, O.P.B. Industry 4.0 Impact on Project Management Factors. In Proceedings of the 6th international Project and Construction Management Conference (e-IPCMC2020), Istanbul, Turkey, 12–14 November 2020.
- Harrison, D.A.; Klein, K.J. What's the difference? diversity constructs as separation, variety, or disparity in organizations. *Acad. Manag. Rev.* 2007, 32, 1199–1228. [CrossRef]
- Frodeman, R. Sustainable Knowledge: A Theory of Interdisciplinarity; Palgrave Pivot; Palgrave Macmillan: London, UK, 2014; ISBN 978-1-137-30301-1.
- Brittain, S.; Sonntag, K. Interdisciplinary Collaboration How to Foster the Dialogue Across Disciplinary Borders. In Advances in Ergonomic Design of Systems, Products and Processes: Proceedings of the Annual Meeting of GFA 2015; Springer: Berlin/Heidelberg, Germany, 2016; ISBN 978-3-662-48659-7.
- Shen, J.; Sung, S.; Zhang, D. Toward an Analytic Framework of Interdisciplinary Reasoning and Communication (IRC) Processes in Science. Int. J. Sci. Educ. 2015, 37, 2809–2835. [CrossRef]
- 114. Pinto, J. Viewing team selection through a temporal lens. Organ. Psychol. Rev. 2017, 7, 171–194. [CrossRef]
- Sankaran, S.; Vaagaasar, A.L.; Bekker, M.C. Assignment of project team members to projects: Project managers' influence strategies in practice. Int. J. Manag. Proj. Bus. 2019, 13, 1381–1402. [CrossRef]
- Drouin, N.; Müller, R.; Sankaran, S.; Vaagaasar, A.L. Balancing vertical and horizontal leadership in projects: Empirical studies from Australia, Canada, Norway and Sweden. Int. J. Manag. Proj. Bus. 2018, 11, 986–1006. [CrossRef]





Driving Sustainable Innovation in New Ventures: A Study Based on the fsQCA Approach

Yu Liu¹ and Hao Zhang^{2,*}

Article

- ¹ School of Business and Management, Jilin University, Changchun 130012, China; liuyu901129@163.com
- ² Greater Bay Area International Institute for Innovation, Shenzhen University, Shenzhen 518061, China
 - * Correspondence: zhanghaord@163.com; Tel.: +86-1580-430-4210

Abstract: With the external environment becoming increasingly complex and changeable, how we can effectively enhance the innovation of companies in sustainability has become the focus of research. For startups, due to their lack of resources and poor independent innovation capabilities, they need to search for external knowledge from outside to meet their own needs. Therefore, obtaining external knowledge sources and adopting appropriate methods for knowledge search is the key to affecting innovation in sustainability. Moreover, enterprise capability is also an important factor restricting sustainable innovation. In this paper, we construct an integrated framework of resources and capabilities based on theoretical learning and practice between 2018 and 2021, containing technical knowledge, market knowledge, a formal search, an informal search, organizational learning, and strategic flexibility. Taking 450 new ventures in China as the research sample, we adopt the fsQCA method and derive the path driving the sustainable innovation of new ventures. The results show that resources and enterprise capabilities combine to influence sustainable innovation, and there are two configuration paths driving the sustainable innovation of new ventures. In the first pathway, a technical knowledge search, a market knowledge search, organizational learning and strategic flexibility are the core conditions; in the second pathway, a formal search, an informal search, organizational learning, and strategic flexibility are the core conditions.

Keywords: sustainable innovation; knowledge search; organizational learning; strategic flexibility; new ventures; QCA

1. Introduction

With the COVID-19 pandemic sweeping across the globe, it has become increasingly difficult for companies to maintain sustainable innovation, and startups are the first to bear the brunt. Enterprise innovation in sustainability refers to continuously obtaining technical and market information, launching innovative products or service projects, and constantly obtaining economic, environmental, and social benefits [1]. By exploring innovation in sustainability, they can create dual value of economic and social sustainability [2]. Moreover, sustainable innovation also provides a new source for the sustainable development of enterprises [3]. Therefore, it is of great significance to explore what the important factors are in driving sustainable innovation and how to maintain sustainable innovation in new ventures. Given the lack of research on the integration of resources and capabilities [4–6], this paper attempts to select appropriate variables in terms of resources and capabilities, trying to explore how they work together to influence sustainable innovation. With the increasing complexity of technology, rising R&D costs, accelerating knowledge growth and diffusion, and shortening product life cycles, the external knowledge search has become the most important factor affecting sustainable innovation of enterprises [7–9]. Dynamic capabilities theory tells us that enterprises can rationally integrate and allocate resources through "enterprise capabilities" and should be able to cope with the external uncertain environments [4,10]. Organizational learning is the course of absorbing and processing new

Citation: Liu, Y.; Zhang, H. Driving Sustainable Innovation in New Ventures: A Study Based on the fsQCA Approach. *Sustainability* 2022, 14, 5738. https://doi.org/10.3390/ su14095738

Academic Editors: Adam R. Szromek and Iwona Zdonek

Received: 14 April 2022 Accepted: 7 May 2022 Published: 9 May 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). knowledge [11], and it involves the recognition, screening, integration, and transformation of knowledge [12]. As one of the most important capabilities of an enterprise, it can provide a core competitive advantage [13]. However, for startups, due to their short establishment time and insufficient experience, they have not yet formed mature learning practices, and so it is not appropriate to apply traditional organizational learning theories. Therefore, it is necessary to explore the inner learning mode of new ventures and its effects on their sustainable innovation. Strategic flexibility is the special organizational ability of enterprises to react to changes in the external environment by reallocating organizational resources, processes, and management activities [14,15]. Enterprises with higher strategic flexibility can reduce their reaction time to environmental changes, and can redeploy resources more effectively, thereby increasing the value of innovative resources [15,16]. Therefore, strategic flexibility can effectively enhance the sustainable innovation of enterprises.

Following the study of Sofka and Grimpe [17] and Guo B and Guo JJ [18] concerning what to search for and how to search, this paper adds two capability variables and constructs an integrated framework of resources and capabilities. DZ Auto Equipment Manufacturing Company is a high-tech intelligent equipment manufacturing enterprise specializing in a robotic automatic welding production line, electrical automation, and robot application. During the survey, it was found that conducting external knowledge search activities and having better organizational learning and strategic flexibility are the keys to sustainable innovation. First, they obtained automation-related technologies and parts processing production lines from suppliers through search methods such as technology purchases and technology licenses, and obtained the core technologies of enterprise production; secondly, they cooperated with Jilin University, Chinese Academy of Sciences and other units to continuously learn new knowledge and upgrade core technologies to improve their own competitive advantages; finally, they were able to flexibly integrate and allocate resources within the enterprise, and respond quickly to changes in the external environment, which is also the key to maintaining competitiveness during the COVID-19 pandemic.

Indeed, sustainable innovation is a necessary condition for the application of new knowledge and can improve organizational learning capabilities while promoting economic growth of enterprises [19,20]. Therefore, sustainable innovation is the result of the combined effect of enterprise resources and capabilities.

Through a literature review, it is found that the following two aspects are mainly studied with regard to knowledge search. The first is the impact of knowledge search on innovation [21–23], while the second is the mechanism of knowledge search on innovation from different perspectives, such as the knowledge integration perspective [24,25], the absorptive capacity perspective [26–28], etc. Most studies used the dimensions of search width and depth, which cannot solve the problem of what knowledge enterprises search for and how to search for knowledge. To address this problem, we take a two-dimensional partitioning approach in our study for knowledge search and market knowledge search, and the search method is divided into a formal search and an informal search.

In addition, most of the existing literature used regression empirical research methods to explore the relationship between resources and capabilities and enterprise sustainable innovation. However, this study method can only solve the marginal effect between variables and cannot solve the complex multivariate interaction [29]. Therefore, we take configuration as the study perspective, adopting the fuzzy set qualitative comparative analysis approach to incorporate the content attributes and method attributes of an external knowledge search, organizational learning, and strategic flexibility into a research model. In this way, it not only examines the impact of a single variable on sustainable innovation, but also explores the interaction between variables, and derives the way to drive sustainable innovation in new ventures.

Based on the analysis of the above arguments, we construct an integrated model of resources and capability, taking 450 new ventures in China as a research sample, and

conduct empirical analysis through the fsQCA approach. The results show that there are two configuration paths driving the sustainable innovation of new ventures. In the first pathway, a technical knowledge search, market knowledge search, organizational learning, and strategic flexibility are the core conditions; in the second pathway, a formal search, an informal search, organizational learning, and strategic flexibility are the core conditions.

Therefore, this paper makes the following related theoretical contributions. Our study's first contribution is constructing an integrated framework of resources and capabilities and exploring whether they are sufficient and necessary to drive sustainable innovation.

The second contribution is that it employs emerging management research methods and presents the configuration paths that drive the sustainable innovation of startups.

The third contribution is that it divides knowledge search into technical knowledge and market knowledge according to the search content, and divides knowledge search into formal search and informal search according to the search method. It also enriches the open innovation theory and expands the research boundary of knowledge search.

The aim here, therefore, is to answer three research questions: (1) Are knowledge search, organizational learning, and strategic flexibility the main factors and necessary conditions driving sustainable innovation? (2) Under the interaction of multiple variables, do resources and capabilities jointly drive the sustainable innovation? (3) What is the configuration path to enhance sustainable innovation in new ventures?

Based on the resources–capabilities integration view, this paper explores the factors that affect innovation for sustainable business and the configuration path that drives sustainable innovation in new ventures. The following section presents relevant theories and literature reviews on knowledge search, organizational learning, and strategic flexibility, which provides theoretical and literature support for our research on sustainable innovation. Next, we propose a conceptual model that drives sustainable innovation and tests the model with the fsQCA method. The final section provides conclusions from the study, managerial implications, and suggestions for sustainable innovation research in the future.

2. Theoretical Retrospection

The open innovation theory holds that the innovation of enterprises does not only rely on internal research and development, but more and more on external knowledge. Enterprises achieve innovation through the purposeful inflow and outflow of knowledge, thereby improving their sustainable competitive advantage in the market [30]. Therefore, enterprises can acquire the new knowledge they need for themselves through external knowledge search activities [21,31].

Nevertheless, the knowledge-based view believes that knowledge is an important resource for enterprises and can provide sustainable competitive advantages [32]. Knowledge management is characterized by the implementation of knowledge strategies and processes in an organization to increase the effectiveness and efficiency [33].

In addition, the dynamic capability theory holds that, facing the complexity and uncertainty of the external environment, enterprises need to cultivate their own dynamic capabilities. Strategic flexibility is a dynamic ability to flexibly allocate resources and coordinate processes when an organization faces environmental threats. Enterprises with high strategic flexibility can use their existing resources more flexibly, thereby increasing the innovative value of resources [15,34].

2.1. Open Innovation Theory

One of the most important theoretical backgrounds of open innovation is cooperative research and development. Before the rise of cooperative R&D, most of the R&D models followed the Schumpeter style, which was called "closed innovation" by Chesbrough [30]. By the end of the 20th century, the closed innovation model was gradually replaced by open innovation, and the new innovation process became more open, decentralized, and democratic, highlighting the interactive characteristics of the innovation process. Based on the open innovation theory proposed by Chesbrough [30], modern companies can not

only integrate beneficial creative resources from within, but also carry out cooperative innovation by introducing, digesting, and absorbing external diversified resources. This makes external knowledge search activities increasingly common.

Dahlander and Gann [35] summarized four reasons why open innovation is generally accepted:

- It reflects socioeconomic shifts in work patterns, with skilled workers seeking a portfolio of jobs rather than a lifetime of work;
- Globalization expands the scope of the market, thereby allowing a finer division of labor;
- (3) An enhanced market system allows businesses to trade ideas;
- (4) New technologies provide cooperation and coordination across geographic distances.

The existing research on open innovation is mainly carried out from two perspectives: the process perspective and the organizational perspective. Open innovation based on the process perspective mainly takes the input and output processes of innovation as a classification dimension. Von Hippel [36] classified innovation processes and outcomes into two dimensions; Knudsen and Mortensen [37] distinguished four types of open innovation with progressive characteristics. The classification of open innovation based on organizational perspective is richer and more diverse. Keupp and Gassmann [38] redefined open innovation by benchmarking high–medium–low portfolio differences in open innovation breadth and depth across firms.

Therefore, open innovation brings more external resources to enterprises, and also brings the possibility for enterprises to search for external knowledge.

2.2. Knowledge-Based View

The resource-based theory tells us that resources should be valuable, scarce, and irreplaceable [39,40]. Therefore, knowledge is the most important strategic resource of an enterprise. Knowledge can be defined as the assessment of an individual's learning resources and knowledge that allow them to record a good performance or task and indicate their sufficiency or suitability for it [41]. The enterprise is a knowledge processing system under the condition of bounded rationality and the specialized division of labor. The types of knowledge mainly include tacit knowledge and explicit knowledge [42–44]. Explicit knowledge is that which can be directly recorded, encoded, or easily communicated and transmitted. Tacit knowledge refers to knowledge that cannot be directly or easily transferred. Tacit knowledge can only be transferred slowly through "learning by doing", so it can bring a sustainable competitive advantage to a business [45,46].

Based on the knowledge-based theory, knowledge is the most important strategic resource of an enterprise, which can be created, stored, and applied, and can contribute to the sustainable innovation. This is mainly reflected in the following four aspects:

- (1) A business is a knowledge system [47];
- (2) Knowledge is the most important strategic resource of an enterprise [48];
- (3) Companies can acquire and transform other resources by using knowledge [49];
- (4) Knowledge possessed by firms is a key factor in making differences between them [50].

It can be seen that the ability of enterprises to use knowledge is the main reason for enterprises to maintain and enhance their competitive advantages. Therefore, only through learning can an enterprise transform the new knowledge acquired from the outside into the competitiveness.

2.3. Dynamic Capability Theory

Enterprise capability is an important strategy, which comes from learning, the acquisition of external resources, and the reallocation of resources in addition to individual capabilities [51]. Makadok [52] defined capability as a special type of resource whose purpose is to increase the productivity of other resources owned by a firm; thus emphasizing the distinction between capability and resource. According to the dynamic capability view, the term "dynamic" refers to the ability to update existing capabilities to achieve flexibility in response to changing circumstances. The term "capability" emphasizes the need for strategic management to properly adjust, integrate, and reallocate internal and external organizational resources and capabilities to adapt to changing circumstances [4,53].

There are two main research perspectives on dynamic capabilities: the cognitive perspective and the process perspective. Scholars with a cognitive perspective argue that firms with highly dynamic capabilities can identify threats and opportunities, influence external changes that align an organization with its business environment, and prevent the emergence of organizational rigidity or inertia [10,51]. Scholars who hold a process perspective view dynamic capabilities as firm practices or processes, and they see dynamic capabilities as a tool that can exist in the form of specific and identifiable processes [4].

In addition, the current background of the digital economy, open innovation, "Internet+" and the COVID-19 pandemic have endowed enterprises with new situations of dynamic capabilities, and the study of dynamic capabilities in special situations has also become a hot spot at this stage.

2.4. External Knowledge Search

As technology becomes increasingly complex, R&D costs continue to rise, and product life cycles continue to shorten, traditional "closed innovation" can no longer meet the knowledge needs of these startups. As a result, they increasingly emphasize the importance of external knowledge, and generally use external knowledge sources. By combing through the existing literature, we have found that the related research of external knowledge search mainly focuses on the following aspects: the first is the emphasis of external sources of knowledge, such as technology purchases, R&D cooperation, alliances, etc. [54,55]. The second is the emphasis of influencing factors of knowledge search, such as the network environment, previous experience, innovation objectives, etc. The third is the study of relationship between the external knowledge search and performance. For example, Wang et al. [8] took data from 187 corporate entities as a research sample, and divided knowledge search into two dimensions of search width and search depth and built a relationship model between knowledge search and organizational innovation performance. Although the research on knowledge search is increasingly rich, more research needs to be done. One reason for this is that the research boundary of knowledge search is still very vague. Some scholars believe that knowledge search is a process of searching for and acquiring knowledge from external knowledge sources. However, some scholars argue that knowledge search includes not only the acquisition process, but also the knowledge integration and creation. The second is that, in existing research, the division of knowledge search dimensions mainly adopts the research results of Laursen and Salter [21], who divided knowledge search into search width and search depth. However, the knowledge search is a complicated process. Questions such as what knowledge to search for and how to search for it need to be solved urgently. The third is that the past research objects were mainly mature enterprises. These authors believed that the relationship between knowledge search and innovation was an "inverted U-shaped" one and concluded that the search was excessive. However, due to a lack of resources, startups have a more obvious dependence on external resources, and so it is more meaningful to take startups as research objects. Based on this, our research constructs a multi-dimensional knowledge search framework, divided into technical knowledge search and market knowledge search in accordance with the search content, and divided into a formal and informal search in accordance with the search method, as is shown in Figure 1.

In future research, first, it is necessary to further explore the research boundary of external knowledge search, and build a new research perspective in the context of the new era, such as the digital economy and the COVID-19 pandemic; secondly, it is necessary to further explore issues related to external knowledge search for small and medium-sized enterprises and start-ups, as well as their differences and connections with mature enterprises; thirdly, it is necessary to examine the influence and mechanism of the combination and balance of different dimensions of knowledge search on innovation.



Knowledge Content

Figure 1. Knowledge search framework. Note: Figure 1 is modified and proposed by the author according to the views of Sofka and Grimpe [17] and Guo B and Guo JJ [18].

2.5. Organizational Learning

The theory of organizational learning was first put forth by Argyris and Schon, who considered it as the result of members interacting within the internal environment of the enterprise [56,57]. In accordance with earlier studies, scholars believed that organizational changes driven by management issues are the key to learning [58]. By the mid-1990s, people generally believed that various organizations could adapt to the changing environment through learning, learn from past successes or failures, and foresee and respond to upcoming threats [59,60]. Through the research on organizational learning, it was found that organizational learning generally has two research perspectives: the ability view and the process view. Scholars who hold the concept of competence believe that organizational learning organization [61]. Scholars who hold a process view believe that organizational learning is a process of knowledge processing by an organization, including knowledge acquisition, knowledge integration, knowledge allocation and utilization, and knowledge creation [62].

However, the existing research still has some limitations. For example, in the early studies, only knowledge management processes were analyzed in intensive departments, and the acquisition, sharing, utilization, creation, and storage of knowledge were not studied as a whole framework. In addition, although the organizational learning mode and framework for large-scale enterprises are relatively mature, organizational contexts, learning approaches, and entrepreneurial roles in startups are different; therefore, the original mature theory cannot be fully applied. As a result, this article explores the internal mechanism of organizational learning by taking startups as the research object.

For startups, organizational learning is a multi-level dynamic process that integrates psychological and social processes [63]. It includes knowledge absorption and integration by members in the organization, the transfer and sharing of knowledge, the application of knowledge, the creation of knowledge, and the storage of new knowledge, as shown in Figure 2. Knowledge recognition involves screening the inflow of new knowledge and integrating it with the internal understanding of the enterprise. In this way, the enterprise can quickly identify the nature of knowledge and shorten the knowledge distance. The essence of knowledge sharing in startups is the flow of knowledge sharing, members of different departments can exchange knowledge, which can improve the overall knowledge level. Knowledge integration is the process of reallocating the internal knowledge of the enterprise. The integrated knowledge will have the characteristics of the actual new understanding and some differences, and this kind of knowledge bears the unique context of the organization. After identifying, sharing, and integrating knowledge, enterprises can

fully absorb the acquired new knowledge and create new knowledge. Then, the enterprises need to perform the explicit processing and storage of new knowledge to increase the firm's total amount of knowledge, and the spillover effect of knowledge can also bring new opportunities for the enterprises.



Figure 2. Internal framework of organizational learning. Note: Figure 2 is the author's point of view based on the above literature and related theories.

The first aim is to continue to explore the theoretical framework and process of organizational learning, and the main research objects are small and medium-sized enterprises and new ventures. As small and medium-sized enterprises and new ventures are more dependent on external knowledge, it is crucial for them to organize learning internally, especially with new and emerging ways of learning. The second is the division of organizational learning dimensions from various perspectives and their impact on innovation. Due to changes in the environment, learning strategies will be continuously adjusted, and the dimensions of organizational learning will be more diverse. The existing empirical research conclusions are inconsistent, and so it is necessary to continue to explore the relationship between organizational learning and innovation, especially with SMEs and new ventures as the research objects.

2.6. Strategic Flexibility

Strategic flexibility originated from the idea of "mutation management" and gradually coordinated with the resource-based view and contingency theory. In the 1980s, strategic flexibility helped in the first stage of development, emphasizing corporate strategic choices and options. At this stage, the traditional business assumptions were surpassed, and the concept and model of strategic learning were constructed. The second stage of the development of strategic flexibility was after the 1990s, reconsidered from the perspective of resources and capabilities. Compared with the previous stage, this stage originated from the fuzzy cognition of strategic flexibility. From the perspective of resources and capabilities, it explains why strategic flexibility can help enterprises deal with the threats of the external environment.

Strategic flexibility is a special organizational ability that an enterprise achieves to respond to changes in the external environment by re-allocating organizational resources, processes, and management activities [15,64]. Strategic flexibility can be proactive, for example, by sensing and adapting to changing customer needs in the market or by reacting to the emergence of new competitors [15]. Therefore, enterprises with strategic flexibility can reduce the response time to dynamic changes and can redeploy resources more effectively, thereby increasing the value of innovative resources [15,34].

However, although most studies have shown that strategic flexibility can positively impact the enterprise's competitiveness and innovation capabilities [65], the consensus on the relationship between strategic flexibility and corporate competitiveness has not been fully agreed upon, especially among small and medium-sized enterprises. The research on strategic flexibility still faces challenges and obstacles. Some scholars believe that the size of an enterprise affects the relationship between strategic flexibility and sustainable

innovation [66]. Compared with large enterprises, startups face many difficulties in fiercely competitive fields, and they are more likely to be exposed to unfamiliar environments. Some scholars believe that the impact of strategic flexibility on sustainable innovation was not significant [67], and some even came to the opposite view [68]. Therefore, the balance between the pros and cons of strategic flexibility in startups needs to be explored urgently.

3. Model

A technical knowledge search refers to the behavior of enterprises by crossing their organizational boundaries to search for and obtain knowledge related to product design, technology, and processes from outside [69,70]. Technical knowledge mainly comes from universities, scientific research institutions, suppliers, and other external enterprises. First of all, an enterprise can obtain the relevant technology needed by the technology purchasing, technology licensing, cooperative research and development, etc., which can be used to correct the technical defects of its products and improve the technical content of the enterprise. The new technical knowledge acquired by the enterprise from outside can be matched and integrated with the original knowledge to produce new knowledge to improve the innovation. Secondly, the scattered and fragmented technical knowledge acquired by the enterprise from the external network can be transformed into new knowledge to improve the level of technology [32,71]. In addition, through technical knowledge search activities, enterprises can greatly increase the knowledge reserves of their internal knowledge base, and at the same time improve their ability to allocate technical resources, which can help to better enhance their sustainable innovation.

A market knowledge search refers to the behavior of enterprises in searching for and obtaining knowledge related to product design solutions, marketing channels and business models by crossing the organizational boundaries, etc. [70]. Market knowledge mainly comes from competitors and customers in the same industry [21]. Market knowledge can also provide benefits for the sustainable innovation of startups. First of all, by establishing contact with customers, it is possible to get feedback on the performance, quality, sales status, and other factors of the enterprise's products, which is conducive to the improvement in knowledge and the increase in its market share. Secondly, the market knowledge acquired from competitors can enable enterprises to understand their own advantages and disadvantages, so as to develop choices and strategies for future knowledge acquisition. Finally, obtaining a large amount of market information through market research, information feedback, etc., can enable enterprises to fully realize market the situation of market complementary products and substitutes, and promote knowledge overflow.

A formal knowledge search refers to the behavior of establishing formal contact with external organizations and acquiring knowledge through contracts, agreements, etc., such as technology purchasing, technology licensing, R&D cooperation, technology alliances, etc. [18]. A formal knowledge search will affect the enterprise's sustainable innovation according to the following aspects: first, through a formal knowledge search, the enterprise's explicit knowledge will be increased, which will help improve the enterprise's knowledge stock and absorptive capacity. Second, establishing an excellent formal relationship with external organizations can increase mutual trust between organizations, establish a mutual learning mechanism, and reduce costs in subsequent searches [72].

Informal knowledge search refers to the behavior of acquiring knowledge through informal communication, exchanges, etc., such as private meetings, conversations, informal employment, reverse engineering, etc. [70]. Informal knowledge search will affect the sustainable innovation of enterprises according to the following aspects: First, employees of an enterprise can establish "weak relationships" by interacting with customers, suppliers, or competitors through private communication and other methods. By establishing this "weak relationship", the breadth of the knowledge search can be increased, thereby obtaining more required knowledge and saving costs. Second, through an informal search, it is more conducive for employees to understand tacit knowledge. Since tacit knowledge is less likely

to be imitated by other competitors, it makes enterprises more conducive to maintaining a sustainable competitive advantage.

It is obvious that knowledge resources constitute the most important part of organizational learning [73], which combines personal judgment, values, abilities, know-how, and technology [74]. Knowledge management is characterized by the implementation of knowledge strategies and processes in the organization to improve the effectiveness and efficiency of business processes and to maintain organizational innovation [60]. Therefore, an organization can provide itself with a core competitive advantage through various knowledge activities. First, personal learning can improve professional skills and practical experience. Entrepreneurs are the core leadership of startups. They can constantly imitate and reflect through learning methods such as experiential learning and cognitive learning, which can reduce the risk of the enterprise in its growth process, they can promote the formation of the enterprise's future innovation strategy, and they can continue to carrying out opportunity identification and utilization. Second, organizational learning can promote the transfer and sharing of knowledge. Organizational decision making, leadership, problem-solving speed, and innovation ability are all improved by spreading knowledge across the parties [75]. Finally, new knowledge can be created through organizational learning. Knowledge creation is characterized by cultivating new abilities and expertise within the organization [76]. Startups can effectively maintain their competitive advantage by producing new knowledge and improving innovation capabilities.

Strategic flexibility reflects the ability of enterprises to flexibly use and allocate resources in response to environmental changes [15]. In today's rapidly changing environment, enterprises need to be able to quickly invest resources and use them freely when responding to changes—that is, to have strategic flexibility [77]. Strategic flexibility allows firms to easily devote resources to different products, regroup resources for production, redefine products and markets, reallocate resources to support new product strategies, etc. [15]. Therefore, the improvement in strategic flexibility can provide sustainable competitive advantages and innovation [78,79]. This is mainly reflected in the following two aspects: First, through the flexible allocation and use of resources, strategic flexibility can meet the needs of innovation for resources, so that enterprises can change key elements or the structure connecting these elements, and then carry out innovation activities [80,81]. Second, strategic flexibility enables enterprises to flexibly allocate and utilize resources and can motivate enterprises to innovate in order to realize the value of internal resource endowments [15].

In summary, this research adopts the open innovation theory and the dynamic capability theory. It takes configuration as the research perspective to construct a framework for the effects of external knowledge search, organizational learning, and strategic flexibility on sustainable innovation in new ventures. The specific model is shown in Figure 3.



Figure 3. Causal configuration analysis model. Note: The author proposes this model based on the above literature and theory; "*High" represents the promotion of Sustainable Innovation; "*Low" represents a reduction of Sustainable Innovation.

4. Data and Methods

4.1. Research Methods

Charles Ragin first proposed qualitative comparative analysis (QCA). It is a research method between qualitative case analysis and quantitative statistical analysis and has the advantages of both [82]. Its main idea is to apply set theory and Boolean algebra to discuss the influence of conditional variables on result variables under configuration matching. Since this method is not limited by sample size, it was first applied to the field of sociology and political science [83]. When Fiss published an article on the feasibility of QCA in organizational strategy in the "Academy of Management Review" in 2007, the QCA method was brought into the field of management. Now, with the continuous optimization of this method, it has become an important tool for research in various management fields [84]. FsQCA is a more recent and extended version of the QCA that may be used for contextual analysis, which investigates how causal relationships are dependent on contextual conditions and is furthermore much closer to statistical approaches [85,86]. This article chose the fsQCA method mainly for the following reasons: One is that the traditional statistical analysis method solves the causal relationship between variables, ignoring the mutual influence between the independent variables. The second is that QCA advocates causal asymmetry and can simultaneously propose two paths of high-level configuration and low-level configuration. Third, compared to the two research methods of CsQCA and mvQCA, the fuzzy set has both qualitative and quantitative attributes, and it can distinguish them accurately [87].

4.2. Samples and Data

(1) Sample selection and collection. Based on the research purposes, this article focuses on startups in China. First of all, we conducted a two month follow-up survey at the DZ Automotive Equipment Manufacturing Company in the summer of 2018 (once a week on average) and conducted interviews and exchanges with the company's general manager and other department heads to understand the company's knowledge needs, knowledge search methods, market operation models, and a series of problems at this stage, laying a practical foundation for this research. Then, from September 2019 to March 2020, we performed an in-depth examination of the KD technology business incubator to conduct research and interviews with the executives. We came to understand the current situation of incubated enterprises, and understand their knowledge sources and methods, as well as the learning process, which provides important practical support for this study. Finally, our questionnaire was distributed and collected from September 2020 to March 2021. According to the relevant research of "Global Entrepreneurship Monitor- China Report", it was divided into areas with developed entrepreneurial activities and areas with less developed entrepreneurial activities. Therefore, this study selected enterprises in Shenzhen, Guangzhou, and Beijing as representatives in the entrepreneurship-developed areas, and selected enterprises in Changchun and Shenyang as representatives in the less-developed entrepreneurship areas, which is basically applicable to new ventures in China. Offline, we collected questionnaires through technology parks, incubated enterprises, talent markets, and university seminars, while online, we delivered questionnaires by email and contacted the local alumni association for them to fill in. The reasons for choosing new ventures as the research object were as follows. First, the internal resources of new ventures are more scarce, and they are more dependent on external resources, and the knowledge search activities are more obvious; second, as an emerging market in China, startups occupy a higher proportion; and third, with the intensification of the COVID-19 pandemic, they are facing greater challenges, and so it is more meaningful to study new ventures.

A total of 1036 questionnaires were issued this time, and 545 questionnaires were collected. After excluding those invalid questionnaires with missing data or inadequate answers, a total of 450 valid questionnaires were obtained, accounting for 43.4% of the total number of questionnaires issued. For details, see Table 1.

Feature Items		Quantity	Percentage
Pagional course	Entrepreneurial developed areas	339	75.3%
Regional source	Entrepreneurial underdeveloped areas	111	24.7%
	Questionnaires issued	1036	E2 (9/
Questionnaire Features	Questionnaire recovery	545	52.6%
	Questionnaire valid	450	43.4%

Table 1. Basic features of the questionnaire.

(2) The statistical characteristics of the sample. The number of valid samples from Shenzhen, Guangzhou, and Beijing was 339, accounting for 75.3% of the effective sample number, and the number of valid samples from Changchun and Shenyang was 111, accounting for 24.7%. In terms of corporate age, 211 enterprises were less than 5 years old, accounting for 46.9% of the total, and 239 enterprises were 6–8 years old, accounting for 53.1% of the total. In terms of enterprise scale, there were 344 firms with less than 500 employees, accounting for 76.5% of the total, and 106 enterprises with more than 500 employees, accounting for 23.5%. In terms of industry, there were 148 enterprises in the manufacturing industry, accounting for 32.9% of the sample; there were 151 enterprises in the information/computer/software industry, accounting for 9.1% of the total sample; there were 70 enterprises in the service industry, accounting for 15.6% of the total sample, and there were 40 enterprises in other industries, accounting for 8.9% of the total sample.

Specific sample characteristics are shown in Table 2.

Table 2. Sample characteristics.

Characteristics	Percentage	Characteristics	Percentage
Age		Scale	
1–2 years	18.9%	1-20	3.1%
3–5 years	28.0%	21-50	10.2%
6–8 years	53.1%	51-100	25.6%
Industry		101-200	16.0%
Manufacturing	32.9%	201-500	21.6%
Information industry	33.6%	501-1000	11.3%
Retail industry	9.1%	More than 1000	12.2%
Service industry	15.6%		
Other	8.9%		

4.3. Variables

The condition variables used in this study are external knowledge search, organizational learning, and strategic flexibility, and the result variable is the sustainable innovation. In the questionnaire design process, the following principles were mainly followed: one was to choose a scale that has been cited many times; the second was to adopt a scale with higher reliability and validity; and the third was to choose a widely used scale. First of all, relying on the focal discussion of the original scale of our study by the teachers and doctoral students of our team at the regular academic meeting, we revised the expression content of the items in the scale combined with the topics to be studied, and added and deleted some items to form the scale used in the study, and constructed a research questionnaire. Then, we communicated with entrepreneurs and executives of some new enterprises on the content of the scale and asked them to fill out the questionnaire to see whether each item is clearly expressed with regard to semantics and whether there are any comprehension issues. At the same time, we tested whether each dimension was significant. This research used the 7-point Likert scoring method, where "1" represents completely inconsistent with the item and "7" represents completely consistent with the item. The following is the measurement of each variable in this study.

 Sustainable innovation (SI). This article mainly drew on the research of Story, Boso, and Cadogan [88] and Hansen and Birkinshaw [89], involving a total of 6 items such as the innovation ability of the enterprise and the satisfaction of the product or service. Examples include "We have a strong ability to innovate", "Compared with competitors, our products or services have more advantages", and so on. The Cronbach's alpha of this variable was 0.803, and the AVE was 0.54, indicating that it reached acceptable standards in terms of reliability and validity.

- (2) External knowledge search. In terms of a technical knowledge search (TKS) and market knowledge search (MKS), this research referred to the research of Sofka and Grimpe [17] and Ki H and Jina [55]. The formal search (FS) and informal search (IS) are based on the research of Guo B and Guo JJ [18]. Examples include "We are good at seeking technical knowledge through purchasing technology, licensing, etc.", "We are good at seeking market information such as new sales channels and marketing strategies from enterprises in the same industry through formal cooperation", "We are good at seeking product or service-related market information from customers through informal communication". The Cronbach's alpha of each variable was 0.830, 0.827, 0.816, and 0.803, and the average was 0.40, 0.49, 0.45, and 0.54, indicating that the reliability and validity reached acceptable standards.
- (3) Organizational learning (OL). This article drew on the research of Morales et al. [90], involving five items including knowledge acquisition ability and knowledge sharing. Examples include "The new knowledge or skills we acquire can bring a competitive advantage to the company" and "We are good at sharing and communicating knowledge to improve our skills." The Cronbach's alpha of this variable was 0.772, and the AVE was 0.53, indicating that it reached acceptable standards in terms of reliability and validity.
- (4) Strategic flexibility (SF). This paper drew on the research of Zhou and Wu [15] and divided strategic flexibility into two dimensions—resource flexibility and coordination flexibility. Examples include "We flexibly allocate production resources to manufacture various products" and "We will effectively redeploy organizational resources to support the firm". The Cronbach's alpha of this variable was 0.791 and the AVE was 0.54, indicating that it reached acceptable standards in terms of reliability and validity.

5. Results

5.1. Data Calibration

It is necessary to calibrate the sample data of the result variable and the condition variable before the QCA test. The so-called data calibration involves transforming the existing raw data into fuzzy-set data required for QCA analysis. In addition to constructing total membership points and non-membership points in this study, it is also necessary to build crossover points [83]. Because the mean value represents the degree of aggregation of the data, and the standard deviation represents the degree of dispersion of the data, according to the research method of Kraus et al. [91], the full membership point is set as "the sum of the mean value and a standard deviation", the non-membership point is set as set to the mean value. The calibration results are shown in Table 3. It can be seen from the results that the consistency level of all the condition variables is less than 0.9, and so there is no necessary condition.

Table 3. Variable calibration results.

Variable	Full Membership	Crossover Point	Non-Membership
SI	6.1296	5.2937	4.4578
TKS	6.0416	5.3873	4.7330
MKS	6.1008	5.4087	4.7166
FS	6.0969	5.4894	4.8819
IS	6.0752	5.3030	4.5308
OL	6.2748	5.4818	4.6888
SF	6.1417	5.3515	4.5614

5.2. Necessity Analysis

Necessity analysis mainly examines the extent to which it constitutes a subset of the condition. If a particular condition variable always appears in the configuration path, it is of necessity, and the measure of the necessity is consistency, the minimum standard of which constitutes a necessary condition, generally considered to be 0.9 [92]. Table 4 shows the conditional necessity test results of conditional variables and outcome variables.

Table 4. Conditional necessity test results.

17 11.	High	n-SI	Low	-SI
variable	Consistency	Coverage	Consistency	Coverage
TKS	0.7744	0.7847	0.4509	0.4093
~TKS	0.4169	0.4588	0.7627	0.7518
MKS	0.7635	0.7817	0.4546	0.4170
~MKS	0.4307	0.4686	0.7620	0.7426
FS	0.7699	0.7818	0.4586	0.4172
~FS	0.4260	0.4677	0.7601	0.7579
IS	0.7824	0.7849	0.4613	0.4145
~IS	0.4164	0.4631	0.7605	0.7579
OL	0.7898	0.7741	0.4888	0.4291
~OL	0.4175	0.4769	0.7426	0.7599
SF	0.8057	0.7877	0.4729	0.4141
~SF	0.4007	0.4591	0.7577	0.7775

Notes: "~" represents the logical operation "not". "High-SI" represents the path to enhance sustainable innovation; "Low-SI" represents the path to reduce sustainable innovation.

5.3. Causal Configuration Analysis

Configuration analysis attempts to expose the causal relationship of multiple conditional variables. In constructing the truth table, the adequacy analysis of the configuration is required. According to the research ideas of Ragin [82], this paper sets the case selection frequency to "1", with the consistency level being greater than 0.8, and the PRI consistency level being greater than 0.75. In the end, we obtain parsimonious, intermediate, and complex solutions. In this paper, the intermediate solution is used as the main reference, and the parsimonious solution is used as an auxiliary reference. As a consistent conclusion of the relationship between the conditional variables and the outcome variables in this paper has not been reached in the academic world; in the analysis process, this article will include all six variables in the "may exist/not exist" option, and analysis by using the software fsQCA3.0 yields the results of the configuration path analysis, as shown in Table 5.

Through the above configuration analysis results, it is found that the solution consistency of high sustainable innovation is 0.8978, which is greater than the acceptable standard of 0.85, and the solution coverage is about 0.64, indicating that a total of 64% of the sample companies are explained. Therefore, these two configuration paths can be used as the driving conditions for the sustainable innovation of the startups. The solution consistency of low sustainable innovation is 0.8768, which is greater than the acceptable standard of 0.85, and the solution coverage is about 0.52, indicating that a total of 52% of the sample enterprises are explained. Therefore, these two paths can be used as a combination of conditions that hinder the sustainable innovation of startups.

X7 . 11	Hig	h-SI	Low-SI			
variable —	Path1	Path2	Path1	Path2		
TKS		•		\otimes		
MKS	•		\otimes	\otimes		
FS	•	•	\otimes	\otimes		
IS		•	\otimes			
OL	•	•	\otimes	\otimes		
SF		•	\otimes	\otimes		
Consistency	0.9167 0.5880	0.9128	0.8767 0.5049	0.8814		
Solution consistency	0.8	978	0.8	768		
Solution coverage	0.6	434	0.5	188		

Table 5. Configuration path analysis results.

Notes: \P = core causal condition (present), implying that the presence of the condition is crucial to the outcome; Θ = core causal condition (absent), suggesting that the absence of the condition is crucial to the outcome; Θ = contributing causal condition (present), implying that the presence of the condition is not essential to the outcome; Θ = contributing causal condition (absent), suggesting that the absence of the condition is not essential to the outcome; Θ = contributing causal condition (absent), suggesting that the absence of the condition is not essential to the outcome; Θ = contributing causal condition (absent), suggesting that the absence of the condition is not essential to the outcome; "High-SI" represents the path to enhance sustainable innovation; "Low-SI" represents the path to reduce sustainable innovation.

It can be seen from the analysis results that there are two configuration paths for driving high levels of sustainable innovation. In configuration path H1, the technical knowledge search, market knowledge search, organizational learning, and strategic flexibility are the core conditions, and the informal search is the edge condition. Therefore, this paper summarizes this path as a knowledge content-driven path. In configuration path H2, the formal search, informal search, organizational learning, and strategic flexibility are the core conditions, and the market knowledge search is the edge condition. This paper summarizes the path as a search method-driven path.

(1)Content-driven path (TKS \times MKS \times OL \times SF). This configuration path emphasizes the importance of external knowledge sources. Moreover, the consistency between the condition variables and sustainable innovation is between 0.7 and 0.9, see Table 4. Therefore, we believe these variables are sufficient conditions to affect sustainable innovation. First, new ventures applying this driving path emphasize the importance of technical knowledge and market knowledge in terms of resources. This is consistent with studies by Voss [93] and Sofka and Grimpe [17]. They believed that technical knowledge is a key factor driving product development and innovation [17,46]. The emergence of new technologies will promote the upgrading of products, so that enterprises can gain greater competitive advantages and occupy a higher market share [94]. However, focusing only on technical knowledge and ignoring market knowledge is not advisable. After the enterprise has mastered the relevant market development trend and demanded change information, it can obtain various technical knowledge required for business development in time to protect the existing market position [95]. Thus, new ventures promote sustainable innovation with the combined effect of technical knowledge and market knowledge. Second, new ventures that apply this driving path emphasize the importance of organizational learning and strategic flexibility in terms of capabilities. This is consistent with the results of Liao et al. [96] and Sanchez [97]. They believe that the level of learning ability determines the effect of knowledge utilization, and the application of new knowledge can enhance sustainable innovation among enterprises [96]. In addition, the enterprise has high strategic flexibility, can better integrate and allocate the internal resources, and can cope with the changing external environment [14,97]. To sum up, this configuration path can promote sustainable innovation in new ventures through the resources-capabilities integration.

(2) Method-driven path (FS × IS × OL × SF). This configuration path emphasizes the importance of searching methods. Moreover, the consistency between the condition variables and sustainable innovation is between 0.7 and 0.9, see Table 4. Therefore, we believe these variables are sufficient conditions to affect sustainable innovation.

First, new ventures applying this driving path emphasize the importance of formal and informal searches in terms of resources. They focus more on knowledge search methods than knowledge sources. This is consistent with the research of Guo B and Guo JJ [18]. They argued that alliances, R&D cooperation, informal exchanges, hiring employees, reverse engineering, professional knowledge training, technology licensing, and patent purchases are all important ways for companies to acquire knowledge resources [18]. Through formal search activities, companies can quickly acquire the knowledge they need and build inter-organizational trust to foster sustainable innovation [98]. An informal search is more conducive to the transfer of tacit knowledge, and the cost is lower [69,99]. In addition, new ventures applying this driving path also emphasize the importance of organizational learning and strategic flexibility in terms of capabilities. In short, through the resources–capabilities integration, this configuration path can also promote sustainable innovation in new ventures.

In addition, two configuration paths have also been derived from this research that lead to reducing sustainable innovation in new ventures, as shown in Table 4. In the configuration path L1, ~FS, ~IS, ~OL, and ~SF are the core conditions, and ~MKS is the edge condition. It can be seen that the lack of knowledge search methods and the lack of organizational learning and strategic flexibility are the reasons for the low sustainable innovation levels of startups. In the configuration path L2, ~TKS, ~MKS, ~OL, and ~SF are the core conditions, and ~FS is the edge condition. It can be seen that insensitivity to the knowledge they need and the lack of organizational learning and strategic flexibility are further reasons for the low sustainable innovation of startups.

5.4. Robustness Test

This study conducted robustness testing by changing the consistency threshold and the number of samples [100,101] to ensure the robustness that drives high sustainable innovation. First, the consistency threshold was increased from 0.8 to 0.85, the sample frequency remained unchanged, and the result did not change compared to before the adjustment.

Second, this study used random sampling to select general samples for re-testing, and the results are shown in Table 6. Comparing Tables 5 and 6, we can see that there are still two paths to drive high levels of sustainable innovation, and the core conditions that constitute this path are roughly the same. Furthermore, their level of consistency and coverage is not very different than before. Therefore, this study can be considered to have good robustness.

Variable	High-SI						
variable	Path1	Path2					
TKS		•					
MKS	•						
FKS	•	•					
IKS		•					
OL	•						
SF	•	•					
Consistency	0.9266	0.9370					
Coverage	0.6398	0.6171					
Solution consistency	0.9	268					
Solution coverage	0.6	657					

Table 6. Robustness test results.

Notes: \bullet = core causal condition (present), implying that the presence of the condition is crucial to the outcome; \bullet = contributing causal condition (present), implying that the presence of the condition is not essential to the outcome; blank spaces indicate that the presence or absence of the condition does not matter with regard to the outcome.

6. Conclusions, Implications, and Limitations

6.1. Conclusions

Based on the open innovation theory and dynamic capability theory, this paper took 450 startups in Shenzhen, Guangzhou, Beijing, Changchun, Shenyang, and other regions of China as the research sample. It took the configuration perspective and used the fsQCA method to explore configuration paths for the sustainable innovation of startups.

First, technical knowledge, market knowledge, a formal search, an informal search, organizational learning, and strategic flexibility are the main factors that affect the sustainable innovation in new ventures. However, they are not sufficient and necessary conditions to affect sustainable innovation—that is, the improvement in sustainable innovation of new ventures is not determined by one variable.

Second, the multiple interactions of resources and capabilities jointly affect the sustainable innovation in new ventures. Only relying on resources or enterprise capabilities cannot drive sustainable innovation. When startups have both resources and enterprise capabilities, sustainable innovation can be achieved.

Third, this study found that there are two pathways to drive the high sustainable innovation levels of startups in China. In the first path, a technical knowledge search, market knowledge search, organizational learning, and strategic flexibility are the core conditions, which is called the knowledge content-driven path. In the second path, a formal search, an informal search, organizational learning, and strategic flexibility are the core conditions, which is called the search method-driven path.

6.2. Theoretical Implications

This research has the following theoretical implications: First, it enriches the open innovation theory and expands the research boundary of the knowledge search. The previous research mainly focused on mature enterprises and reached the research conclusion of over-searching [31,102]. For startups, because they lack resources and rely more heavily on external knowledge, whether they have excessive search behaviors needs to be verified urgently [27]. In addition, in the existing research dimension construction, the duality of the knowledge search is relatively common—for example, search width and search depth [21,31], exploratory search and exploit search [103]. However, such a division approach does not solve the problem of what knowledge startups should search for, who should search for knowledge and how they should search for this knowledge. Therefore, this paper adopts a multi-angle division approach, combining the enterprise knowledge source (knowledge content) and the search method.

Second, it integrates the firm's "resources" and "capacity" frameworks and incorporates them into a research model. Most of the previous literature has addressed the impact of resource acquisition behavior or organizational capabilities on corporate competitiveness and performance or tried to open the black box. However, there is little research on the multivariate interaction of resources and capabilities. Therefore, based on the configuration perspective, this paper incorporates the knowledge search content, knowledge search methods, organizational learning capabilities, and strategic flexibility into a research framework.

Third, this paper presents the configuration paths that drive the sustainable innovation of startups. Most of the previous studies used regression analysis that obtained the marginal effects between variables and failed to draw a path to improve and reduce sustainable innovation. Therefore, this paper explores the integration of knowledge content, search methods, and organizational capabilities, uses the fsQCA method to analyze sample data, and obtains configuration paths that obtain and reduce the sustainable innovation of startups.

6.3. Management Enlightenment

The Fourteenth Five-Year Plan of China states that it is necessary to drive development with innovation and build the capacity for sustainable development in emerging industries. Acquiring the knowledge needed and improving their capabilities have enabled some Chinese enterprises to develop from technological catch-up to technological leadership. Startups are more dependent on external resources than mature enterprises due to their lack of resources, poor R&D capabilities, and insufficient social relations. Therefore, external knowledge search activities have become essential for startups to acquire new knowledge and maintain competitive advantages. Meanwhile, due to the increased uncertainty in the external environment, startups need to improve their dynamic capabilities through learning and flexible thinking. Therefore, acquiring knowledge through knowledge search activities and improving organizational learning capabilities and strategic flexibility can improve sustainable innovation among startups. The management enlightenment of this paper is mainly reflected in the following aspects:

- (1) New ventures should pay attention to external knowledge sources. It can be seen from this study that technical knowledge and market knowledge are the core to maintaining the sustainable innovation of startups. For them, there two main ways of acquiring knowledge, which are technological knowledge-dependent and market knowledgedependent. In the technical knowledge-dependent type, technical knowledge is the core element, and enterprises need to cooperate with suppliers, universities, and research institutions to obtain the technical knowledge. In the market knowledgedependent type, market knowledge is the core element, and enterprises need to obtain relevant market knowledge from organizations such as customers or competitors in order to change the existing business model or increase market share.
- (2) New ventures should pay equal attention to formal and informal searches. It can be seen from the above research that the use of formal and informal search methods by enterprises is also an important factor in maintaining competitiveness. Startups can quickly acquire a large amount of new knowledge through formal search methods, and the introduction of new production equipment or production lines and the employment of relevant technical personnel have made the inflow of knowledge more obvious. However, using informal search methods is equally important, because only through informal search methods can we obtain tacit knowledge or make tacit knowledge explicit, which is more conducive to sustainable innovation.
- (3) New ventures need to improve their organizational learning capabilities. To cope with the complex external environment, startups need to innovate continuously and develop into a learning organization. Looking at some leading enterprises of China, such as Ali, Tencent, and Huawei, we can see that their rapid development is inseparable from the learning and the formation of sustained competitive advantages. One step is to transform the acquired knowledge into the enterprise's innovation. The second step is to strengthen mutual benefit and trust among organizations, promote knowledge exchange and communication, and increase their knowledge transfer and sharing. The third step is to identify and discover new opportunities through the generation of new knowledge, expand the business scale, and enhance the sustainable innovation of the enterprise.
- (4) New ventures need to improve their strategic flexibility. At present, the survival environment of startups is becoming increasingly turbulent. Measures to survive in this complex and changeable environment have become a primary factor in research. However, it is undoubtedly a good method to allocate resources flexibly and reorganize resources. First of all, startups should pay close attention to changes in the business environment and reduce market risks. Secondly, startups need to improve their management capabilities to increase the integration efficiency of internal and external resources. Finally, business managers should cultivate flexible strategic thinking and form organizational routines. In this way, startups can improve their strategic flexibility and strengthen their current operating conditions to maintain sustainable innovation.

6.4. Limitations and Future Research

Based on the open innovation and dynamic capabilities theory, this paper constructed an integrated framework of resource acquisition behavior and capabilities from the perspective of configuration, explored the configuration path that drives the sustainable innovation of startups, and obtained some theoretical and practical implications. However, this research still has the following limitations. First, there are limitations in the research area and sample selection. In the future, sample sources and sample sizes can be expanded, and more regional data can be collected as samples to improve the quality and applicability of the research. Second, this paper did not subdivide enterprises according to the industry to which they belong. Since the ability and knowledge search activities of enterprises are affected by industry factors, the manufacturing, information, and service industries can be subdivided to study the drivers of sustainable innovation in a certain industry. Third, there are other integration frameworks in terms of resource acquisition methods and corporate capabilities. In future research, other different variables can be selected for study.

Author Contributions: Y.L. designed the research and wrote this paper; H.Z. helped with data acquisition and analyzed the data. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Not applicable.

Conflicts of Interest: The authors declare no conflict of interest.

References

- Foxon, T.; Pearson, P. Overcoming barriers to innovation and diffusion of cleaner technologies: Some features of a sustainable innovation policy regime. J. Clean. Prod. 2008, 16, S148–S161. [CrossRef]
- Zhao, D.; Tian, F.; Sun, X.; Zhang, D. The Effects of Entrepreneurship on the Enterprises' Sustainable Innovation Capability in the Digital Era: The Role of Organizational Commitment, Person–Organization Value Fit, and Perceived Organizational Support. Sustainability 2021, 13, 6156. [CrossRef]
- Lopes, J.; Gomes, S.; Pacheco, R.; Monteiro, E.; Santos, C. Drivers of Sustainable Innovation Strategies for Increased Competition among Companies. Sustainability 2022, 14, 5471. [CrossRef]
- 4. Eisenhardt, K.M.; Martin, J.A. Dynamic capabilities: What are they? Strateg. Manag. J. 2000, 21, 1105–1121. [CrossRef]
- Newbert, S.L. Value, rareness, competitive advantage, and performance: A conceptual-level empirical investigation of the resource-based view of the firm. *Strateg. Manag. J.* 2010, 29, 745–768. [CrossRef]
- Desa, G.; Basu, S. Optimization or Bricolage? Overcoming Resource Constraints in Global Social Entrepreneurship. Strateg. Entrep. J. 2013, 7, 26–49. [CrossRef]
- Feng, X.B.; Ma, X.; Shi, Z.; Peng, S.B. How knowledge search affects the performance of reverse internationalization enterprises: The co-moderating role of causation and effectuation. J. Knowl. Manag. 2021, 25, 1105–1127. [CrossRef]
- Wang, C.; Brunswicker, S.; Majchrzak, A. Knowledge search breadth and depth and OI projects performance: A moderated mediation model of control mechanism. J. Knowl. Manag. 2020, 12, 159–176. [CrossRef]
- Kim, C.Y.; Lim, M.S.; Yoo, J.W. Ambidexterity in External Knowledge Search Strategies and Innovation Performance: Mediating Role of Balanced Innovation and Moderating Role of Absorptive CapacityJ. Sustainability 2019, 11, 5111. [CrossRef]
- Teece, D.J. Explicating dynamic capabilities: The nature and microfoundations of (sustainable) enterprise performance. *Strateg. Manag. J.* 2010, 28, 1319–1350. [CrossRef]
- Real, J.C.; Roldán, J.L.; Leal, A. From Entrepreneurial Orientation and Learning Orientation to Business Performance: Analysing the Mediating Role of Organizational Learning and the Moderating Effects of Organizational Size. Br. J. Manag. 2014, 25, 186–208. [CrossRef]
- 12. Argote, L.; Miron-Spektor, E. Organizational Learning: From Experience to Knowledge. Organ. Sci. 2011, 22, 1123–1137. [CrossRef]
- Dawes, P.L.; Lee, D.Y.; Midgley, D. Organizational learning in high-technology purchase situations: The antecedents and consequences of the participation of external IT consultants. *Ind. Mark. Manag.* 2007, *36*, 285–299. [CrossRef]
- 14. Sanchez, R. Preparing for an uncertain future: Managing organizations for strategic flexibility. *Int. Stud. Manag. Organ.* 1997, 27, 71–94. [CrossRef]
- Zhou, K.Z.; Wu, F. Technology Capability, Strategic Flexibility, and Product Innovation. Strateg. Manag. J. 2010, 31, 547–561. [CrossRef]

- Liao, S.; Liu, Z.; Fu, L.; Ye, P. Investigate the role of distributed leadership and strategic flexibility in fostering business model innovation. *Chin. Manag. Stud.* 2019, 13, 93–112. [CrossRef]
- Sofka, W.; Grimpe, C. Specialized search and innovation performance: Evidence across Europe. R&D Manag. 2010, 40, 310–323. [CrossRef]
- Guo, B.; Guo, J.J. Patterns of technological learning within the knowledge systems of industrial clusters in emerging economies: Evidence from China. *Technovation* 2011, 31, 87–104. [CrossRef]
- Brown, J.S.; Duguid, P. Organizational Learning and Communities-of-Practice: Toward a Unified View of Working, Learning, and Innovation. Organ. Sci. 1991, 2, 40–57. [CrossRef]
- Jiménez-Jiménez, D.; Sanz-Valle, R. Innovation, organizational learning, and performance. J. Bus. Res. 2011, 64, 408–417. [CrossRef]
- Laursen, K.; Salter, A. Open for innovation: The role of openness in explaining innovation performance among U.K. manufacturing firms. Strateg. Manag. J. 2010, 27, 131–150. [CrossRef]
- Laursen, K.; Salter, A. Searching high and low: What types of firms use universities as a source of innovation? *Res. Policy* 2004, 33, 1201–1215. [CrossRef]
- Yu, Y.; Yuan, L.; Li, J. Knowledge search modes and innovation performance: The moderating role of strategic R&D orientation. *Technol. Anal. Strateg.* 2019, 31, 625–640. [CrossRef]
- Guo, R.P. Effectuation, opportunity shaping and innovation strategy in high-tech new ventures. *Manag. Decis.* 2019, 57, 115–130. [CrossRef]
- Amankwah-Amoah, J.; Adomako, S. The Effects of Knowledge Integration and Contextual Ambidexterity on Innovation in Entrepreneurial Ventures. J. Bus. Res. 2021, 127, 312–321. [CrossRef]
- Luis, J.; Newell, S.; Fernandez-Mesa, A.; Alegre, J. Depth and breadth of external knowledge search and performance: The mediating role of absorptive capacity. *Ind. Mark. Manag.* 2015, 47, 86–97. [CrossRef]
- Flor, L.M.; Cooper, S.Y. External knowledge search, absorptive capacity and radical innovation in high-technology firms. *Eur. Manag. J.* 2018, 34, 1453–1471. [CrossRef]
- Aliasghar, O.; Rose, E.L.; Chetty, S. Where to search for process innovations? The mediating role of absorptive capacity and its impact on process innovation. *Ind. Mark. Manag.* 2019, 82, 199–212. [CrossRef]
- Stefan, V.; Chaim, N. Set-theoretic methods for the social sciences: A guide to qualitative comparative analysis. Int. J. Soc. Res. Method 2013, 16, 165–166. [CrossRef]
- 30. Chesbrough, H.; Crowther, A. Beyond high tech: Early adopters of open innovation in other industries. *R&D Manag.* 2006, *36*, 229–236.
- Katila, R.; Ahuja, G. Something Old, Something New: A Longitudinal Study of Search Behavior and New Product Introduction. Acad. Manag. J. 2002, 45, 1183–1194. [CrossRef]
- 32. Grant, R.M. Toward a knowledge-based theory of the firm. Strateg. Manag. J. 1996, 10, 109–122. [CrossRef]
- Muthuveloo, R.; Shanmugam, N.; Ai, P.T. The impact of tacit knowledge management on organizational performance: Evidence from Malaysia. Asia Pac. Manag. Rev. 2017, 22, 192–201. [CrossRef]
- 34. Oviatt, B.M.; Mcdougall, P.P. Defining International Entrepreneurship and Modeling the Speed of Internationalization. *Entrep. Theory Pract.* 2010, 29, 537–554. [CrossRef]
- 35. Dahlander, L.; Gann, D. How open is innovation? Res. Policy 2010, 39, 699–709. [CrossRef]
- Von Hippel, E. Comment on "Is open innovation a field of study or a communication barrier to theory development?". *Technovation* 2010, 30, 555. [CrossRef]
- Knudsen, P.; Mortensen, B. Some immediate-but negative-effects of openness on product development performance. *Technovation* 2011, 31, 54–64. [CrossRef]
- 38. Keupp, M.; Gassmann, O. Determinants and archetype users of open innovation. R&D Manag. 2009, 39, 331–341. [CrossRef]
- Barney, J.B. Types of Competition and the Theory of Strategy: Toward an Integrative Framework. Acad. Manag. Rev. 1986, 11, 791–800. [CrossRef]
- 40. Barney, J.B. Firm Resource and Sustained Competitive Advantage. J. Manag. 1991, 17, 99–120. [CrossRef]
- Muoz-Pascual, L.; Galende, J. Ambidextrous Knowledge and Learning Capability: The Magic Potion for Employee Creativity and Sustainable Innovation PerformanceJ. Sustainability 2020, 12, 3966. [CrossRef]
- 42. Nonaka, I. The dynamic theory of knowledge creation. Organ Sci. 1994, 5, 14–37. [CrossRef]
- 43. Belderbos, R.; Faems, D.; Leten, B.; Looy, B.V. Technological activities and their impact on the financial performance of the firm: Exploitation and exploration within and between firms. *J. Prod. Innovat. Manag.* **2010**, *27*, 869–882. [CrossRef]
- Nonaka, I.; Takeuchi, N.H. The knowledge-creating company: How Japanese companies create the dynamics of innovation. Long Range Plan. 1996, 29, 592. [CrossRef]
- 45. Hakanson, L. Creating knowledge: The power and logic of articulation. Ind. Corp. Chang. 2007, 16, 51-88. [CrossRef]
- Kogut, B.; Zander, U. Knowledge of the Firm, Combinative Capabilities, and the Replication of Technology. Organ. Sci. 1992, 3, 383–397. [CrossRef]
- 47. Markus, R.; Torsten, R. Uncertainty, pluralism, and the knowledge-based theory of the firm: From J.-C. Spender's contribution to a socio-cognitive approach. *Eur. Manag. J.* 2013, *31*, 706–716. [CrossRef]
- Salunke, S.; Weerawardena, J.; McColl-Kennedy, J.R. The central role of knowledge integration capability in service innovationbased competitive strategy. Ind. Mark. Manag. 2019, 76, 144–156. [CrossRef]
- Yong, J.K.; Song, S.; Sambamurthy, V.; Lee, Y.L. Entrepreneurship, knowledge integration capability, and firm performance: An empirical study. *Inform. Syst. Front.* 2012, 14, 1047–1060.
- 50. Spender, J.D. Making knowledge the basis of a dynamic theory of the firm. Strateg. Manag. J. 1996, 17, 45–62. [CrossRef]
- 51. Teece, D.J. Dynamic capabilities and entrepreneurial management in large organizations: Toward a theory of the (entrepreneurial) firm. *Eur. Econ. Rev.* 2016, *86*, 202–216. [CrossRef]
- 52. Makadok, R.A. Rational-Expectations Revision of Makadok's Resource/Capability Synthesis. *Strateg. Manag. J.* 2002, 23, 1051–1057. [CrossRef]
- Bock, A.J.; Opsahl, T.; George, G.; Gann, D.M. The Effects of Culture and Structure on Strategic Flexibility During Business Model Innovation. J. Manag. Stud. 2012, 49, 279–305. [CrossRef]
- Kang, K.H.; Kang, J. How Do Firms Source External Knowledge for Innovation? Analysing Effects of Different Knowledge Sourcing Methods. Int. J. Innov. Manag. 2009, 13, 1–17. [CrossRef]
- Kang, K.H.; Kang, J. Do External Knowledge Sourcing Modes Matter for Service Innovation? Empirical Evidence from South Korean Service Firms. J. Prod. Innovat. Manag. 2013, 31, 176–191. [CrossRef]
- Wu, I.L.; Chen, J.L. Knowledge management driven firm performance: The roles of business process capabilities and organizational learning. J. Knowl. Manag. 2014, 18, 1141–1164. [CrossRef]
- Ashtari, A.; Salehi, J. A study on the effect of organizational learning on organizational performance with an emphasis on dynamic capacity. Int. J. Ind. Eng. Comp. 2014, 4, 1421–1424. [CrossRef]
- 58. Levitt, B.; March, J.G. Organizational learning. Annu. Rev. Sociol. 1988, 14, 319–338. [CrossRef]
- Crossan, M.M.; Lane, H.W.; White, R.E. An organizational learning framework: From intuition to institution. Acad. Manag. Rev. 1999, 24, 522–537. [CrossRef]
- 60. Levinthal, D.A.; March, J.G. The myopia of learning. Strateg. Manag. J. 1993, 14, 95–112. [CrossRef]
- Chien, S.Y.; Tscai, H. Dynamic capability, knowledge, learning, and firm performance. J. Organ. Chang. Manag. 2012, 25, 434–444. [CrossRef]
- Chiva, R.; Alegre, J. Organizational Learning Capability and Job Satisfaction: An Empirical Assessment in the Ceramic Tile Industry. Soc. Sci. Electron. Publ. 2010, 20, 323–340. [CrossRef]
- Souza, C.; Takahashi, A. Dynamic capabilities, organizational learning and ambidexterity in a higher education institution. *Learn.* Organ. 2019, 26, 397–411. [CrossRef]
- 64. Li, Y.; Su, Z.; Liu, Y. Can strategic flexib ility help firms profit from product innovation? Technovation 2010, 30, 300–309. [CrossRef]
- Xiu, L.; Liang, X.; Chen, Z.; Xu, W. Strategic flexibility, innovative HR practices, and firm performance A moderated mediation model. Pers. Rev. 2017, 46, 1335–1357. [CrossRef]
- Gorondutse, A.H.; Arshad, D.; Alshuaibi, A.S. Driving sustainability in SMEs' performance: The effect of strategic flexibility. J. Strategy Manag. 2021, 14, 64–81. [CrossRef]
- 67. Pagell, M.; Krause, D.R. Re-exploring the relationship between flexibility and the external environment. *Oper. Res.* 2004, 21, 629–649. [CrossRef]
- Bahrami, H.; Evans, S. Emerging organizational regimes in high technology firms: The bi-modal form. Hum. Resour. Manag. 2010, 28, 25–50. [CrossRef]
- Rosenkopf, L.; Nerkar, A. Beyond local search: Boundary: Panning, exploration, and impact in the optical disk industry. *Strateg. Manag. J.* 2001, 22, 287–306. [CrossRef]
- 70. Rosenkopf, L.; Almeida, P. Overcoming Local Search Through Alliances and Mobility. Manag. Sci. 2003, 49, 751–766. [CrossRef]
- Li, H.Y.; Anthea, Z.Y.; Marjorie, L. Special Issue: Knowledge Search, Spillovers, and Creation in Emerging Markets. *Manag. Organ. Rev.* 2010, 6, 309–311. [CrossRef]
- Li, J.J.; Poppo, L.; Zhou, K.Z. Relational mechanisms, formal contracts, and local knowledge acquisition by international subsidiaries. *Strateg. Manag. J.* 2010, 31, 349–370. [CrossRef]
- Acar, M.F.; Tarim, M.; Zaim, H.; Zaim, S.; Delen, D. Knowledge management and ERP: Complementary or contradictory? Int. J. Inf. Manag. 2017, 37, 703–712. [CrossRef]
- Yam, R.; Lo, W.; Tang, E.; Lau, A.K. Analysis of sources of innovation, technological innovation capabilities, and performance: An empirical study of Hong Kong manufacturing industries. *Res. Policy* 2011, 40, 391–402. [CrossRef]
- 75. Marques, J.M.R.; La Falce, J.L.; Marques, F.M.F.R.; De Muylder, C.F.; Silva, J.T.M. The relationship between organizational commitment, knowledge transfer and knowledge management maturity. *J. Knowl. Manag.* **2019**, *23*, 489–507. [CrossRef]
- Känsäkoski, H. Information and knowledge processes as a knowledge management framework in health care: Towards shared decision making? J. Doc. 2017, 35, 41–44. [CrossRef]
- 77. Brozovic, D. Strategic flexibility: A review of the literature. Int. J. Manag. Rev. 2018, 20, 2–31. [CrossRef]
- Chan, A.; Ngai, E.; Moon, K. The effects of strategic and manufacturing flexibilities and supply chain agility on firm performance in the fashion industry. *Eur. J. Oper. Res.* 2017, 259, 486–499. [CrossRef]
- Rajala, R.; Westerlund, M.; Moeller, K. Strategic flexibility in open innovation—Designing business models for open source software. *Eur. J. Market.* 2012, 46, 1368–1388. [CrossRef]

- Mai, Y.; Yang, H.; Zhang, G. Does Business Model Innovation Enhance the Sustainable Development of New Ventures? Understanding an Inverted-U Relationship. Sustainability 2021, 13, 262. [CrossRef]
- 81. Guo, H.; Su, Z.; Ahlstrom, D. Business model innovation: The effects of exploratory orientation, opportunity recognition, and entrepreneurial bricolage in an emerging economy. *Asia Pac. J. Manag.* **2016**, *33*, 533–549. [CrossRef]
- 82. Ragin, C.C. Set relations in social research: Evaluating their consistency and coverage. Polit. Anal. 2006, 14, 291–310. [CrossRef]
- Rihoux, B.; Álamos-Concha, P.; Bol, D.; Marx, A.; Rezsöhazy, I. From niche to mainstream method? A comprehensive mapping of QCA applications in journal articles from 1984 to 2011. Polit. Res. Q. 2013, 66, 175–184. [CrossRef]
- 84. Wu, C.W. Antecedents of franchise strategy and performance. J. Bus. Res. 2015, 68, 1581–1588. [CrossRef]
- Shipley, M.F.; Johnson, M.; Pointer, L.; Yankov, N. A fuzzy attractiveness of market entry (FAME) model for market selection decisions. J. Oper. Res. Soc. 2013, 64, 597–610. [CrossRef]
- 86. Denk, T.; Lehtinen, S. Contextual analyses with QCA-methods. Qual. Quant. 2014, 48, 3475–3487. [CrossRef]
- Cooper, B.; Glaesser, J. Using case-based approaches to analyse large datasets: A comparison of Ragin's fsQCA and fuzzy cluster analysis. Int. J. Soc. Res. Methodol. 2010, 14, 31–48. [CrossRef]
- Story, V.M.; Boso, N.; Cadogan, J.W. The Form of Relationship between Firm-Level Product Innovativeness and New Product Performance in Developed and Emerging Markets. J. Prod. Innovat. Manag. 2015, 32, 45–64. [CrossRef]
- 89. Hansen, M.T.; Birkinshaw, J. The Innovation Value Chain. Harv. Bus. Rev. 2007, 85, 121–130. [CrossRef]
- Morales, F.; Antonio, J.; Verdú, J. Influence of personal mastery on organizational performance through organizational learning and innovation in large firms and SMEs. *Technovation* 2007, 27, 547–568. [CrossRef]
- 91. Kraus, S.; Ribeiro-Soriano, D.; Schüssler, M. Fuzzy-set qualitative comparative analysis (fsQCA) in entrepreneurship and innovation research—The rise of a method. *Int. Entrep. Manag. J.* 2018, *14*, 15–33. [CrossRef]
- Beynon, M.J.; Jones, P.; Pickernell, D. Country-based comparison analysis using fsQCA investigating entrepreneurial attitudes and activity. J. Bus. Res. 2016, 69, 1271–1276. [CrossRef]
- Voss, G.B.; Voss, Z.G. Strategic Ambidexterity in Small and Medium-Sized Enterprises: Implementing Exploration and Exploitation in Product and Market Domains. Organ. Sci. 2013, 24, 1459–1477. [CrossRef]
- 94. Wiklund, J.; Shepherd, D. Knowledge-based resources, entrepreneurial orientation, and the performance of small and mediumsized businesses. *Strateg. Manag. J.* 2003, 24, 1307–1314. [CrossRef]
- 95. Kohli, A.K.; Jaworski, B. Market orientation: Antecedents and consequences. J. Mark. 1993, 57, 53–70. [CrossRef]
- Liao, S.H.; Fei, W.C.; Liu, C.T. Relationships between knowledge inertia, organizational learning and organization innovation. *Technovation* 2008, 28, 183–195. [CrossRef]
- 97. Sanchez, R. Strategic flexibility in product competition. Strateg. Manag. J. 2010, 16, 135–159. [CrossRef]
- Lefebvre, V.M.; Sorenson, D.; Henchion, M.; Gellynck, X. Social capital and knowledge sharing performance of learning networks. Int. J. Inf. Manag. 2016, 36, 570–579. [CrossRef]
- 99. Pyka, A. Informal networking and industrial life cycles. Technovation 2000, 20, S0166–S4972. [CrossRef]
- Schneider, C.Q.; Wagemann, C. Reducing complexity in Qualitative Comparative Analysis (QCA): Remote and proximate factors and the consolidation of democracy. *Eur. J. Polit. Res.* 2010, 45, 751–786. [CrossRef]
- Pappas, I.O.; Kourouthanassis, P.E.; Giannakos, M.N.; Chrissikopoulos, V. Explaining online shopping behavior with fsQCA: The role of cognitive and affective perceptions. J. Bus. Res. 2016, 69, 794–803. [CrossRef]
- Roper, S.; Love, J.H.; Bonner, K. Firms' knowledge search and local knowledge externalities in innovation performance. *Res. Policy* 2017, 46, 43–56. [CrossRef]
- 103. March, J.G. Exploration and Exploitation in Organizational Learning. Organ. Sci. 1991, 2, 71-87. [CrossRef]

MDPI St. Alban-Anlage 66 4052 Basel Switzerland Tel. +41 61 683 77 34 Fax +41 61 302 89 18 www.mdpi.com

Sustainability Editorial Office E-mail: sustainability@mdpi.com www.mdpi.com/journal/sustainability



MDPI St. Alban-Anlage 66 4052 Basel Switzerland Tel: +41 61 683 77 34

www.mdpi.com



ISBN 978-3-0365-6482-1